UNITED STATES SECURITIES AND EXCHANGE COMMISSION

WASHINGTON, D.C. 20549

FORM 8-K

CURRENT REPORT

Pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934

Date of Report (Date of earliest event reported): August 10, 2021

Recro Pharma, Inc.

(Exact name of Registrant as Specified in Its Charter)

Pennsylvania (State or Other Jurisdiction of Incorporation) 001-36329 (Commission File Number) 26-1523233 (IRS Employer Identification No.)

1 E. Uwchlan Ave, Suite 112 Exton, Pennsylvania (Address of Principal Executive Offices)

19341

(Zip Code)

Registrant's Telephone Number, Including Area Code: 770 534-8239

(Former Name or Former Address, if Changed Since Last Report)

Che	ck the appropriate box below if the Form 8-K filing is intend	led to simultaneously satisfy the filin	ng obligation of the registrant under any of the following provisions:			
	Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)					
	Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)					
	Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))					
	Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))					
	Securities registered pursuant to Section 12(b) of the Act:					
		Trading				
	Title of each class	Symbol(s)	Name of each exchange on which registered			
Common stock, par value \$0.01		REPH	The NASDAQ Stock Market LLC			
Indicate by check mark whether the registrant is an emerging growth company as defined in Rule 405 of the Securities Act of 1933 (§ 230.405 of this chapter) or Rule 12b-2 of the Securities Exchange Act of 1934 (§ 240.12b-2 of this chapter).						
Em	erging growth company \square					
If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised financial accounting standards provided pursuant to Section 13(a) of the Exchange Act.						

Item 1.01 Entry into a Material Definitive Agreement.

Acquisition of IriSys, LLC

Unit Purchase Agreement and Seller Note

On August 13, 2021, Recro Pharma, Inc. (the "Company") entered into a Unit Purchase Agreement (the "Purchase Agreement") by and among the Company, IriSys, LLC, a California limited liability company ("IriSys"), IriSys, Inc., a California corporation ("IriSys Parent"), Continent Pharmaceuticals U.S., Inc., a Delaware corporation, and EPS Americas Corp., a Delaware corporation (each, a "Seller" and collectively, the "Sellers"), and IriSys Parent, in its capacity as the representative of the Sellers, pursuant to which the Company agreed to purchase 100% of the outstanding units of IrySis from the Sellers (the "Acquisition"). The closing of the Acquisition (the "Closing") occurred simultaneously with execution of the Purchase Agreement.

Located in San Diego, California, IriSys provides contract pharmaceutical product development and manufacturing services, specializing in formulation research and development and good manufacturing practices of clinical trial materials and specialty pharmaceutical products.

The total purchase price (the "Purchase Price") payable by the Company for the Acquisition was approximately \$49.85 million, subject to certain adjustments as provided in the Purchase Agreement based on transaction expenses, cash and cash equivalents, indebtedness and net working capital of IriSys at the Closing. The Purchase Price consisted of: (i) \$25.5 million in cash paid to the Sellers at the Closing; (ii) 9,302,718 shares of the Company's common stock, par value \$0.01 per share, to be issued to the Sellers six (6) months after the Closing; and (iii) a subordinated promissory note issued to certain Sellers by the Company in the aggregate principal amount of approximately \$6.1 million (the "Note"). The Note is unsecured, has a three-year term, and bears interest at a rate of six percent (6%) per annum. The Note may be prepaid in whole or in part at any time prior to the maturity date. The Note is expressly subordinated in right of payment and priority to the Company's existing loans with Athyrium Capital Management issued under that certain Credit Agreement, dated as of November 17, 2017 (as amended from time to time, the "Credit Agreement"), by and among the Company, the Company's subsidiaries named as guarantors therein (the "Guarantors"), the lenders party thereto (the "Lenders"), and Athyrium Opportunities III Acquisition LP ("Athyrium Opportunities III") as administrative agent.

The Purchase Agreement contains customary representations and warranties regarding the Company and the Sellers, and customary covenants of the Company and the Sellers.

Sixth Amendment to Credit Agreement

On August 13, 2021, the Company entered into a Sixth Amendment to Credit Agreement (the "Amendment") by and among the Company, the Guarantors, the Lenders, and Athyrium Opportunities III, in its capacity as the administrative agent.

The Amendment provides (i) the Lenders' consent regarding the Acquisition, (ii) Athyrium Opportunities III's and the Lenders' consent to changes made to the Credit Agreement, including permitting the Note, subject to a subordination agreement which subordinates the obligations of the Company under the Note to the obligations under the Credit Agreement, (iii) for the inclusion of an updated fee letter in connection with the Credit Agreement, specifying the fees the Company will pay to Athyrium Opportunities III and the Lenders in connection with the Credit Agreement, as amended by the Amendment, and (iii) an extension of the maturity date of the loan issued under the Credit Agreement from March 31, 2023 to December 31, 2023.

The foregoing description of the Purchase Agreement, the Note, the Amendment, and the transactions contemplated thereby does not purport to be complete and is qualified in its entirety by reference to the Purchase Agreement, the Form of Note, and the Amendment, copies of which are filed as Exhibits 10.1, 10.2, and 10.3 hereto and are hereby incorporated into this Current Report on Form 8-K by reference. The Purchase Agreement has been included to provide shareholders and investors with information regarding its terms. It is not intended to provide any other factual information about the Company, the Sellers or IriSys. In particular, the assertions embodied in the representations and warranties contained in the Purchase Agreement are subject to qualifications and limitations agreed to by the respective parties in connection with negotiating the terms of the Purchase Agreement, including information contained in a confidential disclosure schedule provided by the Sellers and IriSys to the Company in connection with the signing of the Purchase Agreement. The confidential disclosure schedule attached to the Purchase Agreement contains information that modifies, qualifies and creates exceptions to the representations and warranties set forth in the Purchase Agreement. Moreover, certain representations and warranties in the Purchase Agreement were used for the purpose of allocating risk between the Company and the Sellers rather than establishing matters as facts. Accordingly, shareholders and investors should not rely on the representations and warranties in the Purchase Agreement as characterizations of the actual state of facts about the Company, the Sellers, or IriSys.

Item 2.01 Completion of Acquisition or Disposition of Assets.

Reference is made to the disclosure under Item 1.01 above which is hereby incorporated in this Item 2.01 by reference.

Item 2.03 Creation of a Direct Financial Obligation or an Obligation under an Off-Balance Sheet Arrangement of a Registrant.

Reference is made to the disclosure under Item 1.01 above which is hereby incorporated in this Item 2.03 by reference.

Item 3.02 Unregistered Sales of Equity Securities.

Reference is made to the disclosure under Item 1.01 above which is hereby incorporated in this Item 3.02 by reference.

Item 7.01 Regulation FD Disclosure.

On August 10, 2021, the Company issued a press release announcing its entrance into a letter of intent in connection with the Acquisition. A copy of the press release is attached as Exhibit 99.1 hereto and incorporated by reference herein.

On August 13, 2021, the Company issued a press release announcing the closing of the Acquisition. A copy of the press release is attached as Exhibit 99.2 hereto and incorporated by reference herein.

The information in this Item 7.01, including Exhibits 99.1. and 99.2, is furnished and shall not be deemed "filed" for purposes of Section 18 of the Securities Exchange Act of 1934, as amended (the "Exchange Act"), or otherwise subject to liabilities under that section, and shall not be deemed to be incorporated by reference into the filings of the Company under the Securities Act of 1933, as amended, or the Exchange Act, regardless of any general incorporation language in such filings. This Current Report on Form 8-K will not be deemed an admission as to the materiality of any information of the information in this Item 7.01, including Exhibits 99.1, and 99.2.

Item 9.01 Financial Statements and Exhibits.

(a) Financial Statements of Business Acquired.

As permitted by Item 9.01(a)(3) of Form 8-K, the financial statements required by Item 9.01(a) of Form 8-K will be filed by the Company by an amendment to this Current Report on Form 8-K not later than 71 days after the date upon which this Current Report on Form 8-K was required to be filed.

(b) Pro Forma Financial Information.

As permitted by Item 9.01(b)(2) of Form 8-K, the pro forma financial information required by Item 9.01(b) of Form 8-K will be filed by the Company by an amendment to this Current Report on Form 8-K not later than 71 days after the date upon which this Current Report on Form 8-K was required to be filed.

(d) Exhibits

Below is a list of exhibits included with this Current Report on Form 8-K.

Exhibit No.	Document
110.	
10.1	Unit Purchase Agreement, dated August 13, 2021, by and among Recro Pharma, Inc., IriSys, LLC, the Sellers (as defined therein), and IriSys, Inc. as the
	Seller's Representative
10.2	Form of Subordinated Promissory Note
10.3	Sixth Amendment to Credit Agreement, dated as of August 13, 2021, by and among Recro Pharma, Inc., certain subsidiaries of Recro Pharma, Inc., named as
	guarantors therein, the lenders named therein and Athyrium Opportunities III Acquisition LP, as administrative agent
99.1	Press release of Recro Pharma, Inc., dated August 10, 2021
99.2	Press release of Recro Pharma, Inc., dated August 13, 2021
104	Cover Page Interactive Data File (embedded within the Inline XBRL document)

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

RECRO PHARMA, INC.

Date: August 13, 2021 /s/ J. David Enloe, Jir.

J. David Enloe, Jr.
President and Chief Executive Officer

UNIT PURCHASE AGREEMENT

by and among

RECRO PHARMA, INC.,

IRISYS, LLC,

THE SELLERS (AS DEFINED THEREIN)

and

IRISYS, INC., AS THE SELLERS' REPRESENTATIVE

Dated as of August 13, 2021

TABLE OF CONTENTS

		<u>Page</u>
ARTICLE I DE	FINITIONS	1
ARTICLE II PU	RCHASE AND SALE OF UNIT; CLOSING; CONSIDERATION	11
2.1	Purchase and Sale of Units	11
2.2	Closing	11
2.3	Closing Consideration	11
2.4	Purchaser Shares	11
2.5	Escrow Amount	12
2.6	Closing Payment Certificate	13
2.7	Final Calculations; Disputes	13
2.8	Adjustments	15
2.9	Closing Transaction Expenses; Closing Indebtedness	16
2.10	Reliance on Consideration Spreadsheet	16
2.11	Withholding	16
2.12	Registration of Purchaser Shares.	16
ARTICLE III R	RTICLE III REPRESENTATIONS AND WARRANTIES OF THE PURCHASER	
3.1	Organization; Corporate Power and Authorization	18
3.2	Binding Effect; Noncontravention; No Approvals	18
3.3	Broker Fees	
3.4	Investment	18
3.4	Investment No Litigation	18 19
3.6	No Litigation Purchaser SEC Filings.	19
3.7	Purchaser Shares.	19
3.8	Exchange Act Compliance.	19
3.9	Form S-3 Eligibility.	19
3.10	Number of Purchaser Shares.	20
		20
ARTICLE IV R	EPRESENTATIONS AND WARRANTIES OF THE SELLERS	20
4.1	Power and Authorization	20
4.2	Binding Effect and Noncontravention; Approvals	20
4.3	Units	20
4.4	No Litigation	21
4.5	Broker Fees	21
4.6	Purchaser Shares.	21
ARTICLE V RE	EPRESENTATIONS AND WARRANTIES OF THE COMPANY	22
5.1	Organization; Qualification; Limited Liability Company Power and Authorization	22
5.2	Binding Effect and Noncontravention; Approvals	22
5.3	Capitalization	23
5.4	Subsidiaries	23
5.5	Financial Statements	24
5.6	Absence of Certain Developments	24
5.7	Assets	26
5.8	Tax Matters	26
5.9	Environmental Matters	27

5.10	Intellectual Property	28
5.11	Real Estate	29
5.12	Litigation	30
5.13	Employee and Labor Relations	30
5.14	Employee Plans	
		31
5.15	Affiliate Transactions	33
5.16	Insurance	33
5.17	Material Contracts	34
5.18	FDA and Regulatory Matters	35
5.19	Accounts Receivable	37
5.20	Customers and Suppliers	37
5.21	Broker Fees	38
5.22	Bank Accounts	38
5.23	Corrupt Practices Act; Ex-Im Laws	38
5.24	Government Contracts and Regulatory Matters.	38
5.25	Disclaimer of the Company	42
ARTICLE VI C	OVENANTS AND OTHER AGREEMENTS	42
6.1	Public Announcements	42
6.2		43
6.3	Employee Matters Record Retention	44
6.4		44
6.5	Indemnification of Directors and Officers; Insurance Tax Matters	45
6.6 6.7	Sellers' Representative	48 49
6.8	R&W Policy Reserved.	49
6.9	Release	49
6.10	Settlement Amounts.	50
6.11	Employee Equity Grants.	50
6.12	Patent Matters	50
		30
ARTICLE VII I	NDEMNIFICATIONS; SURVIVAL	50
7.1	Survival of Representations, Warranties and Covenants	50
7.2	Indemnification by the Sellers	51
7.3	Indemnification by the Purchaser	52
7.4	Indemnity Limitations	52
7.5	Procedures Relating to Indemnification	54
7.6	Further Acknowledgement by the Purchaser	56
7.7	Tax Treatment of Indemnification Payments	57
7.8	Exclusive Remedy	57
ARTICLE VIII	CLOSING DELIVERABLES	57
8.1	Closing Deliverables of the Company and the Sellers	57
8.2	Closing Deliverables of the Purchaser	58
ARTICLE IX M	IISCELLANEOUS	59
9.1	Governing Law	59
9.2	Jurisdiction	59
9.3	Waiver of Jury Trial	59
9.4	Expenses	59

9.5	Attorneys' Fees	59
9.6	Waiver; Remedies Cumulative	59
9.7	Notices	60
9.8	Assignment	61
9.9	No Third-Party Beneficiaries	61
9.10	Amendments	61
9.11	Company Disclosure Schedule	61
9.12	Specific Performance	61
9.13	Construction	61
9.14	Further Assurances	62
9.15	Entire Agreement	62
9.16	Severability	62
9.17	Mutual Drafting	62
9.18	Counterparts; Facsimile	62
9.19	Waiver of Conflicts Regarding Representation	62

EXHIBITS

Exhibit A Form of Escrow Agreement

Exhibit B Illustrative Calculation of Net Working Capital Amount

Exhibit C Purchase Price Allocation

Exhibit D Form of Note

Exhibit E R&W Policy

SCHEDULES

Schedule I Indebtedness

UNIT PURCHASE AGREEMENT

This UNIT PURCHASE AGREEMENT (the "Agreement") is made as of August 13, 2021, by and among (i) Recro Pharma, Inc., a Pennsylvania corporation (the "Purchaser"), (ii) IriSys, LLC, a California limited liability company (the "Company"), (iii) IriSys, Inc., a California corporation ("IriSys"), Continent Pharmaceuticals U.S., Inc., a Delaware corporation, and EPS Americas Corp., a Delaware corporation (each, a "Seller" and collectively, the "Sellers"), and (iv) IriSys, in its capacity as the representative of the Sellers (the "Sellers' Representative"). The Purchaser, the Company, the Sellers and the Sellers' Representative are sometimes referred to individually as a "Party" and collectively as the "Parties." Certain capitalized terms that are used herein are defined in Article I herein.

WHEREAS, the Sellers collectively own 100% of the outstanding units (the "<u>Units</u>") of the Company, with the Units owned by each Seller set forth on <u>Section 5.3(a)</u> of the Company Disclosure Schedule.

WHEREAS, the Purchaser desires to acquire from the Sellers all of the Units, and the Sellers desire to sell to the Purchaser all of the Units, in accordance with the terms and subject to the conditions set forth in this Agreement.

NOW, THEREFORE, in consideration of the premises and the mutual promises made herein, and in consideration of the representations, warranties, covenants and agreements herein contained, intending to be legally bound, the Parties hereby agree as follows:

ARTICLE I

DEFINITIONS

For the purposes of this Agreement, the following terms have the meanings set forth below:

- "Action" means any action, suit, claim, order, charge, mediation, litigation, hearing, proceeding, investigation, audit, inquiry, complaint, grievance or arbitration commenced, brought, conducted or heard by or before any Governmental Entity.
 - "Adjustment Escrow Amount" means Three Hundred Thousand Dollars (\$300,000).
- "Affiliate" means, with respect to any Person, any other Person directly or indirectly controlling, controlled by or under common control with such first Person within the meaning of the Exchange Act.
- "Affiliated Group" means any affiliated group within the meaning of Section 1504(a) of the Code, or any consolidated, combined, unitary or similar group for Tax purposes.
- "Aggregate Consideration" means the sum of (a) the Closing Cash Consideration, (b) the Purchaser Shares, (c) the principal amount of the Note, (d) any Closing Adjustment Surplus Amount received by the Sellers, and (e) any Escrow Amount received by the Sellers.
- "Anti-Corruption Laws" means all applicable U.S. and non-U.S. Laws relating to the prevention of corruption and bribery, including the U.S. Foreign Corrupt Practices Act of 1977.
- "Approval" means any authorization, consent or approval required to be obtained from any Governmental Entity or other Person in connection with this Agreement and the consummation of the Transaction.

"Business Day" means any day excluding Saturday, Sunday and any day that is a legal holiday under the Laws of the State of California or is a day on which banking institutions located in California or New York, New York are authorized or required by Law to close.

"CARES Act" means The Coronavirus Aid, Relief, and Economic Security Act, Pub.L. 116-136 (03/27/2020).

"cGMPs" shall mean applicable current good manufacturing practices as defined in the U.S. Code of Federal Regulations, 21 CFR Part 210 et seq., the European Union Guidelines to Good Manufacturing Practices for Medicinal Products for Human and Veterinary Use (Vol. IV Rules Governing Medicinal Products in the European Union 2004), and any successor regulatory schemes, as well as any corresponding requirements in other regulatory jurisdictions as applicable.

"Closing Cash" means all cash and cash equivalents (including marketable securities, bank deposits and short-term investments convertible to cash within ninety (90) days, net of any breakage costs required to liquidate such securities, deposits and short term investments), in each case, of the Company, as of immediately prior to the Closing. Notwithstanding anything to the contrary contained herein, "Closing Cash" shall (i) include the amount of any uncleared checks and drafts and wires received by the Company prior to the Closing and (ii) exclude (x) the amount of any outstanding uncleared checks and drafts and wires issued by the Company prior to the Closing, (y) the amount of any cash or cash equivalents that is not freely usable by the Company because it is subject to restrictions, limitations or Taxes on use or distribution by any Law, contract or otherwise, including restrictions on dividends and repatriations or any other form of restriction (including if subject to incremental costs, expenses, penalties, Taxes or other amounts to be incurred in respect of any such distribution), amounts held in escrow, amounts held in reserve pursuant to any letter of credit or otherwise as collateral or amounts that are deposited with a third party (other than a bank or similar financial institution); provided, that, notwithstanding anything contained in this definition to the contrary, fifty percent (50%) of the lease deposit made by the Company in connection with the Nancy Ridge Lease shall expressly constitute Closing Cash; and provided further, that, the amount of cash or cash equivalents excluded pursuant to clause (y) on account of any Taxes or incremental costs, expenses, penalties, or other amounts, or (z) any cash or cash equivalents used to repay Indebtedness or Closing Transaction Expenses prior to the Closing.

"Closing Cash Consideration" shall mean: (a) \$25,500,000, minus (b) the amount, if any, by which the Target Working Capital exceeds the Estimated Net Working Capital Amount or <u>plus</u> the amount, if any, by which the Estimated Net Working Capital Amount exceeds the Target Working Capital, <u>plus</u> (c) the amount of Estimated Closing Cash, minus (d) the amount of Estimated Closing Indebtedness, if any, minus (e) the Estimated Closing Transaction Expenses, and minus (f) the Escrow Amount.

"Closing Indebtedness" means all outstanding Indebtedness of the Company, as of immediately prior to the Closing.

"Closing Transaction Expenses" means, without duplication, in each case, to the extent outstanding or unpaid as of immediately prior to the Closing (whether or not invoiced), all fees, costs and expenses incurred by or on behalf of the Company in connection with the preparation, negotiation, execution and consummation of the Transactions (including with respect to any auction or sales process conducted in respect thereof or any offering or marketing materials prepared in connection therewith or the solicitation, discussion or negotiation of strategic alternatives hereto or thereto), including (i) all such fees, costs, and expenses of legal counsel, investment bankers, brokers, accountants and other representatives and consultants; and (ii) all change-of-control bonuses or similar amounts payable by the Company to any director, officer, employee, consultant or independent contractor in connection with the entry into or consummation of the transactions contemplated by

this Agreement that become payable as a result of the Closing and remain outstanding or unpaid as of immediately prior to the Closing and the employer portion of any employment or payroll Taxes with respect to any such payments; <u>provided</u>, that, Closing Transaction Expenses shall be calculated without duplication of any amounts to the extent included in the calculation of Closing Indebtedness or Net Working Capital Amount and shall expressly exclude any and all severance obligations of the Company and the employer portion of any employment or payroll Taxes with respect thereto.

"Code" means the Internal Revenue Code of 1986, as amended.

"Company Disclosure Schedule" means the disclosure schedule constituting exceptions to and applicable disclosures associated with the Company's representations and warranties set forth in Article V hereof, prepared and delivered by the Company concurrently with the execution of this Agreement, as the same may be amended and supplemented from time to time, as required and/or permitted herein.

"Company Fundamental Representations" means each of the following representations and warranties made by the Sellers or the Company, as applicable: Section 4.1 (Power and Authorization), Section 4.2(a) (Binding Effect and Noncontravention; Approvals), Section 4.3 (Units), Section 4.5 (Broker Fees), Section 5.1 (Organization; Qualification; Corporate Power and Authorization), Section 5.2 (Binding Effect and Noncontravention; Approvals) (solely with respect to clause (a) and clause (b)(i)), Section 5.3 (Capitalization), Section 5.4 (Subsidiaries), and Section 5.19 (Broker Fees).

"Company Intellectual Property Rights" means all of the Intellectual Property Rights exclusively owned by the Company.

"Company Material Adverse Change" means any change, event, occurrence, development, circumstance, effect, condition or state of facts that, individually or in the aggregate, is or would reasonably be expected to be, individually or in the aggregate, materially adverse to (a) the business, assets, properties, condition (financial or otherwise), liabilities or results of operations of the Company, or (b) the ability of the Company to perform its obligations under this Agreement and the Transaction Documents to which it is a party or on the ability of the Company to consummate the Transactions; provided, however, that, solely with respect to clause (a), none of the following shall be deemed in themselves, either alone or in combination, to constitute, and none of the following shall be taken into account in determining whether there has been, or will be, a Company Material Adverse Change: any adverse change, event, occurrence, development, circumstance, effect, condition or state of facts attributable to (i) operating, business, regulatory or other conditions in the industry in which the Company operates, (ii) general economic conditions, including changes in the credit, debt or financial or capital markets, in each case, in the United States or anywhere else in the world, (iii) changes in GAAP or other accounting requirements or principles or any changes in applicable Laws or the interpretation thereof, in each case, after the date hereof, (iv) the failure of the Company to meet or achieve the results set forth in any projection or forecast (provided that (x) the change, event, occurrence, development, circumstance, effect, condition or state of facts underlying such failure and/or (y) any variance in the actual financial position, condition, performance or results of operations of the Company for any then-current period as compared to one or more prior periods, in each case, may be considered in determining whether there has been a Company Material Adverse Change), (vi) global, national or regional political conditions, including hostilities, acts of war, sabotage or terrorism or military actions or any escalation, worsening or diminution of any such hostilities, acts of war, sabotage or terrorism or military actions existing or underway as of the date hereof, (vii) hurricanes, earthquakes, floods or other natural disasters, and (viii) any natural or man-made disaster, acts of God, epidemic, pandemic (including COVID-19), disease outbreak or other health crises or public health event; provided, further, that with respect to subclauses (i), (ii), (iii), (vi), (vii) and (viii), any such change, event, occurrence, development, circumstance, effect, condition or state of facts may be considered in determining whether a Company Material Adverse Change has occurred to the extent such change, event, occurrence, development,

circumstance, effect, condition or state of facts has had a disproportionate and adverse effect on the Company as compared to other Persons that conduct business in the industry in which the Company conducts business.

"Confidentiality Agreement" means the Confidentiality Agreement, by and between the Company and the Purchaser dated as of March 24, 2021.

"Consideration Spreadsheet" means a spreadsheet which shall be prepared in accordance with the Company's Organizational Documents and which shall include the following information, as of the Closing: (a) the names of all Sellers, (b) the Units held by each Seller, (c) the portion of the Closing Cash Consideration payable to each Seller, (d) the portion of the Note allocable to the applicable Seller, (e) the number of Purchaser Shares issuable to each Seller, and (f) each Seller's applicable Pro Rata Portion.

"Contracts" means all written or oral contracts, agreements, subcontracts, indentures, notes, loans, instruments, leases, mortgages, franchises, licenses, purchase orders, sale orders, proposals, bids, understandings or commitments, which are legally binding.

"Covid-19 Quarantine Period" means, with respect to each regular work location, the period during which any Governmental Entity restricted nonessential work on-site at such location.

"Employee Plans" means each "employee benefit plan" (as defined in Section 3(3) of ERISA), whether or not subject to ERISA, and each employment, bonus, incentive compensation, stock option, stock purchase or other equity-based (or equity-derivative) compensation, benefit, profit-sharing, retirement, disability, insurance, vacation, deferred compensation, severance, termination, retention, change of control compensation and other similar material fringe, welfare or other employee benefit plans, programs, agreements, contracts, policies or binding arrangements (whether or not in writing) (i) that is in each case maintained or contributed to by the Company or any of its ERISA Affiliates for the benefit of any current or former employee, director, or other service provider of the Company or (ii) with respect to which the Company or any ERISA Affiliate has any Liability, contingent or otherwise.

"Environmental Laws" means all Laws concerning pollution or protection of human health or the environment and Laws which prohibit, limit, regulate or control Hazardous Material, including but not limited to the Clean Air Act, the Clean Water Act, the Resource Conservation and Recovery Act, the Comprehensive Environmental Response, Compensation, and Liability Act of 1980, the Safe Drinking Water Act, the Toxic Substances Control Act, as amended by the Frank R. Lautenberg Chemical Safety for the 21St Century Act, the Emergency Planning and Community Right-to-Know Act of 1986, the Occupational Safety and Health Act, and analogous state laws, including the California Safe Drinking Water and Toxic Enforcement Act of 1986 (known as Proposition 65).

"Environmental Permit" means any Permit required by Environmental Laws for the operation of the business of the Company as currently conducted.

"ERISA" means the Employee Retirement Income Security Act of 1974, as amended.

"ERISA Affiliate" means any trade or business, whether or not incorporated, that, together with the Company, is treated as a "single employer" within the meaning of Sections 414(b), (c), (m), or (o) of the Code.

"Escrow Agent" means Citibank, N.A.

"Escrow Agreement" means the Escrow Agreement, dated as of the Closing Date, by and among the Escrow Agent, the Purchaser and the Sellers' Representative, in substantially the form attached hereto as Exhibit A.

"Escrow Amount" means, collectively, the Adjustment Escrow Amount and the Indemnity Escrow Amount.

"Evaluation Material" means any information, documents or materials regarding the Company or its respective business furnished or made available to the Purchaser and its Representatives in the online data room, any management presentations attended by the Purchaser or in any other reasonable form for dissemination of such information, documents or materials in expectation of, or in connection with, the Transactions.

"Ex-Im Laws" means all applicable Laws relating to export, reexport, transfer, and import controls, including the Export Administration Regulations, and the customs and import Laws administered by U.S. Customs and Border Protection.

"Family Group" means for any individual, such individual's current or former spouse, their respective parents, descendants of such parents (whether natural or adopted and, for the avoidance of doubt, including any of such individual's children) and the spouses of such descendants, and any trust, limited partnership, corporation or limited liability company controlled by such individual and established solely for the benefit of such individual or such individual's current or former spouse, their respective parents, descendants of such parents (whether natural or adopted) or the spouses of such descendants.

"FDA" means the United States Food and Drug Administration, or any successor organization.

"GAAP" means United States generally accepted accounting principles as in effect from time to time.

"Government Bid" means any bid, proposal, offer or quote for supplies, services or construction, whether solicited or unsolicited, made by the Company which is intended by the Company to result in a Government Contract and is outstanding as of the date hereof. A Government Bid (a) includes but is not limited to any bid, proposal, offer or quote made by the Company that has been received or accepted by the offeree or other recipient but has not resulted in a Government Contract prior to the Closing Date, but (b) does not include any bid, proposal, offer or quote made by the Company that has resulted in a Government Contract prior to the Closing Date.

"Governmental Entity" means any court, tribunal, arbitrator or any government or political subdivision thereof, whether federal, state, county, local or foreign, or any agency, authority, or instrumentality of such governmental or political subdivision, or any entity exercising executive, legislative, judicial, regulatory or administrative functions of government.

"Government Official" shall mean any officer or employee of any Governmental Entity or any Person acting in an official capacity for or on behalf of any such Governmental Entity or for or on behalf of any public international organization or any political party or official thereof or any candidate for political office.

"Government Vendor Subcontract" means a Contract between the Company and another Person to supply supplies or services to the Company to be used in performing a Government Contract.

"<u>Hazardous Material</u>" means any chemical, pollutant, contaminant, material, substance or waste that has been designated by any Governmental Entity to be radioactive, toxic, hazardous, or otherwise a danger to

health, reproduction or the environment because of its dangerous or deleterious properties or characteristics, or is otherwise regulated by any Environmental Law.

"Healthcare Laws" means the Federal Food, Drug, Cosmetic Act, or FDCA, and any other applicable Law in the United States or outside of the United States where the Company operates as a contract manufacturer or contract research organization, including Laws that regulate the design, development, testing, studying, contract pharmaceutical manufacturing, processing, storing, importing or exporting, licensing, labeling or packaging of pharmaceutical products, or that regulate fraud and abuse, kickbacks, recordkeeping, referrals, the hiring of employees or acquisition of services or supplies from those who have been excluded from government health care programs, quality, safety, privacy, security, licensure, accreditation or any other aspect of providing contract pharmaceutical manufacturing and research services.

"Indebtedness" means the aggregate principal amount of, and accrued interest with respect to, and any other payment obligations (including prepayment penalties, premiums, breakage costs and other fees and expenses associated with the repayment thereof), without duplication, of the Company with respect to: (a) indebtedness for borrowed money; (b) obligations evidenced by bonds, debentures, notes, letters of credit, or similar instruments, declared but unpaid or accrued but unpaid dividends or distributions payable to members; (c) obligations to pay amounts under any lease which obligation is required to be included on the balance sheet of the Company as of such date as determined in accordance with GAAP, including, for the avoidance of doubt, Accounting Standards Codification 842 which became effective for the Company on January 1, 2021 (expressly excluding, for purposes of clarity, any amounts due under the Nancy Ridge Lease in excess of the amount set forth on Schedule I); (d) obligations for the deferred purchase price of property or services (including "earn-outs" and "seller notes" related to business acquisitions) (expressly excluding, for purposes of clarity, any amounts due under the Nancy Ridge Lease in excess of the amount set forth on Schedule I); (e) all obligations under any interest rate swap agreements or interest rate hedge agreements; (f) the items and corresponding amounts as expressly set forth on Schedule I; and (g) guarantees made directly or indirectly by the Company of obligations of the type described in clauses (a) through (f) above.

"Indemnified Person" means any Purchaser Indemnified Person or Seller Indemnified Person, as applicable.

"Indemnifying Person" means any Person(s) against whom a claim for indemnification is being asserted under any provision of Article VII herein.

"Indemnity Escrow Amount" means Two Hundred and Fifty Thousand Dollars (\$250,000).

"Indemnity Period" means the period from the Closing Date to the date that is twelve (12) months from the Closing Date.

"Intellectual Property Rights" means all (a) patents and patent applications, as well as any reissues, continuations, divisions, extensions or reexaminations thereof, (b) trademarks, service marks, trade dress, trade names, corporate names and registrations and applications for registration thereof, together with all of the goodwill associated therewith, (c) registered copyrights and applications for registration thereof, (d) computer software, and (e) any or all of the following and all worldwide rights in, arising out of, or associated therewith: domain names, uniform resource locators and other names and locators associated with the Internet.

"International Employee Plan" means each Employee Plan that has been adopted or maintained by the Company or any of its ERISA Affiliates, or with respect to which the Company will or may have any Liability, for the benefit of any current or former employee, director, or other service provider of the Company whose primary work location is based outside the United States.

"IRS" means the United States Internal Revenue Service.

"Knowledge" means (a) as it applies to the representations, warranties and covenants made by the Company in this Agreement, the actual knowledge of Gerald Yakatan, Robert Giannini, Louis Scotti, Jean Wang, Adolfo Ramirez, Victoria Smith, Igor Nikoulin, and Brooke Yakatan (or the knowledge such Person would reasonably be expected to have after making reasonable inquiry, including of their direct reports), in each case, as of the date hereof; and (b) as it applies to representations, warranties and covenants made by a particular Seller in this Agreement, the actual knowledge of such Seller (or the knowledge such Seller would reasonably be expected to have after making reasonable inquiry with respect to the applicable matter).

"Law" means any law, statute, rule, regulation, ordinance, regulation, code, directive, and other pronouncement having the effect of law of the United States of America, any foreign country or any domestic or foreign state, county, city or other political subdivision or of any Governmental Entity.

"Liability" or "Liabilities" means any and all debts, liabilities and obligations of any nature whatsoever, whether accrued or fixed, absolute or contingent, mature or unmatured or determined or indeterminable.

"<u>Licensed Intellectual Property Rights</u>" means all Intellectual Property Rights, other than Company Intellectual Property Rights, that are licensed by Company from third parties and that are used by Company used in connection with the Company's business.

"Lien" means any mortgage, lien, pledge, charge, security interest, claim, contractual restriction, easement, right-of-way, option, conditional sale or other title retention agreement or encumbrance of any kind.

"Loss" means any direct or indirect Liability, indebtedness, claim, loss, Taxes, damage, Lien, deficiency, obligation, judgment, penalty, responsibility, remediation costs or other costs or expenses (including reasonable attorneys' fees and expenses paid or incurred in connection with any of the foregoing); provided, that Loss shall expressly exclude any punitive damages except to the extent included in any Third-Party Claim.

"Members" has the meaning set forth in the Operating Agreement.

"Nancy Ridge Lease" means that certain Lease, by and between BMR-6828 Nancy Ridge LP and the Company, dated as of June 26, 2014, as amended.

"NASDAQ" means the NASDAQ Capital Market.

"Net Working Capital Amount" means (a) the aggregate dollar amount of all assets characterized as current assets, excluding (x) loans receivable from any employees of the Company or any Seller or any of their respective Affiliates or Related Parties under GAAP and (y) any contract assets or other assets related to revenue recognition, minus (b) the aggregate dollar amount of all liabilities characterized as current liabilities of the Company under GAAP (excluding any contract liabilities or liabilities for customer prepayments), in the case of each of clause (a) and clause (b), as of 12:01 a.m. Pacific time on the Closing Date; provided, that "current assets" and "current liabilities" shall include only those specific balance sheet line items included in the illustrative calculation of the Net Working Capital Amount set forth on Exhibit B; provided, further, that, notwithstanding anything herein to the contrary, in no event will the determination of "Net Working Capital Amount" include any (i) Closing Cash, (ii) Closing Indebtedness, solely to the extent Indebtedness is already included within the calculation of Closing Cash Consideration, (iii) Closing Transaction Expenses, solely to the extent Transaction Expenses are already included within the calculation of the Closing Cash Consideration, or (iv) deferred Tax assets or liabilities.

"Note" means the unsecured, subordinated promissory note of the Purchaser issuable to one or more of the Sellers as set forth in the Consideration Spreadsheet in the aggregate principal amount of Six Million and One Hundred Sixteen Thousand Six Hundred Seventy-Two Dollars and Seventy-Two Cents (\$6,116,672.72), in substantially the form attached hereto as Exhibit D.

"Off-The-Shelf Software" means any shrinkwrap, clickwrap or other commercially available software licenses granted to the Company for third party software used by the Company.

"Operating Agreement" means the Third Amended and Restated Operating Agreement of the Company by and among the Members party thereto, dated as of February 22, 2019.

"Order" means any writ, judgment, decree, injunction or similar order of any Governmental Entity, in each case whether preliminary or final.

"Organizational Documents" means, with respect to any Person, as applicable, the certificate of incorporation, by-laws, partnership agreement, limited liability company agreement, operating agreement or other similar document, all as amended, supplemented, restated and replaced from time to time.

"<u>Partnership Audit Provisions</u>" means the Bipartisan Budget Act of 2015 and Sections 6221-6231 of the Code and the Treasury Regulations promulgated thereunder (and comparable provisions of state and local Tax Law).

"<u>Permits</u>" means all permits, licenses, grants, easements, clearances, variances, orders, certificates, exemptions, registrations, authorizations, certificates of occupancy and other permits, consents and approvals required by any Governmental Entity to lawfully operate the Company's business as currently conducted.

"Permitted Liens" means (a) Liens for Taxes, which either are (i) not yet due and payable or (ii) being contested in good faith and by any appropriate Action, and adequate reserves (as determined in accordance with GAAP) have been established on the Company's books with respect thereto, (b) Liens to secure landlords, sublandlords, licensors, sublicensors or licensees under real estate leases, licenses or other rental or lease agreements, (c) deposits or pledges made in connection with, or to secure payment of, utilities or similar services, workers' compensation, unemployment insurance, pension or other social security, governmental insurance and governmental benefits mandated under applicable Laws, or to secure the performance of tenders, statutory obligations, surety and appeal bonds, bids, leases, government contracts, performance and return of money bonds and similar obligations, (d) mechanics', materialmen's or contractors' Liens or any similar statutory Lien for amounts not yet due and payable and incurred in the ordinary course of business, (e) zoning, entitlement, building and other similar restrictions which are not violated by the current conduct of the Company's business, (f) purchase money Liens in any property acquired by the Company in the ordinary course of business, and (g) Liens or encumbrances imposed on the underlying fee interest in real property subject to Company Leases.

"Person" means any individual, partnership, corporation, association, limited liability company, joint stock company, a trust, joint venture, firm, association, unincorporated organization, Governmental Entity or other entity.

"PPP Loan" means a Paycheck Protection Program Loan pursuant to Section 1102 of the CARES Act.

"Pro Rata Portion" means, with respect to each Seller, the applicable portion of the Aggregate Consideration then received by such Seller relative to the Aggregate Consideration then received by all Sellers, as set forth on the Consideration Spreadsheet.

- "Purchaser Common Stock" means the Purchaser's common stock, par value \$0.01 per share.
- "<u>Purchaser Fundamental Representations</u>" means each of the following representations and warranties made by the Purchaser: <u>Section 3.1</u> (Organization; Corporate Power and Authorization), Section 3.2 (Binding Effect and Noncontravention; Approvals) (solely with respect to clause (a) and clause (b)(i)), and <u>Section 3.3</u> (Broker Fees).
- "<u>Purchaser Material Adverse Change</u>" means any change, event, occurrence, circumstance, condition, effect, development or state of facts that is or is reasonably likely to be, individually or in the aggregate, a material impairment in the ability of the Purchaser to perform its obligations under this Agreement and the Transaction Documents to which it is a party or on the ability of the Purchaser to consummate the Transactions.
- "<u>Purchaser Shares</u>" means 9,302,718 shares of Purchaser Common Stock. Notwithstanding the foregoing, if there are any stock splits, stock combination or similar events with respect to the Purchaser Common Stock on or after the Closing Date and through and including the Purchaser Shares Issuance Date, the number of "Purchaser Shares" shall be equitably adjusted.
 - "R&W Policy" means the representation and warranties insurance policy set forth on Exhibit E.
- "Related Party" means (i) each Seller; (ii) any Affiliate of any Seller (other than the Company); or (iii) each member of the Family Group of the foregoing in clauses (i) and (ii).
- "Related Party Transaction" means any Contract, arrangement or transaction between the Company, on the one hand, and any Related Party, on the other hand, other than (x) salary or other compensation or benefits under Employee Plans paid or payable in the ordinary course of business to employees in consideration for bona fide services performed by such employees and (y) this Agreement, the Transaction Documents and the Transactions).
- "Release" means spilling, leaking, pumping, emitting, emptying, discharging, injecting, escaping, leaching, dumping, discarding, burying, abandoning, migration, or disposing into the environment.
- "Representatives" means, with respect to a particular Person, any manager, director, officer, employee, agent, consultant, advisor, or other representative of such Person, including legal counsel, accountants, and financial advisors.
- "Restrictive Covenant Agreement" means each Restrictive Covenant Agreement, dated as of the date hereof, to be entered into by and between the Purchaser, each Seller and the other parties thereto.
- "Sanctioned Country" means, at any time, a country, region or territory which is itself the subject or target of any applicable comprehensive Sanction Laws.
- "Sanctioned Person" means, at any time, (a) any Person listed in any applicable Sanctions-related list of designated Persons maintained by U.S. Office of Foreign Assets Control or the U.S. Department of State, (b) any Person operating, organized or resident in a Sanctioned Country or (c) any Person owned fifty percent (50%) or more, individually or in the aggregate, directly or indirectly, by any such Person or Persons described in the foregoing clause (a).
- "Sanctions Laws" any applicable Laws regarding economic or financial sanctions or trade embargoes imposed, administered or enforced from time to time by any Governmental Entity.

"Securities Act" means the Securities Act of 1933, as amended.

"Seller Material Adverse Change" means, with respect to any Seller, any change, event, occurrence, circumstance, condition, effect, development or state of facts that is or is reasonably likely to be, individually or in the aggregate, a material impairment in the ability of such Seller to perform such Seller's obligations under this Agreement and the Transaction Documents to which such Seller is a party or on the ability of such Seller to consummate the Transactions.

"Straddle Period" means any taxable period beginning prior to and ending after the Closing Date. For purposes of apportioning liability for Taxes of the Company in connection with any Straddle Period, (i) in the case of Taxes based upon or related to income or receipts, the amount of any such Taxes allocable to the portion of the taxable period ending on the Closing Date shall be determined based on an interim closing of the books as of the close of business on the Closing Date; and (ii) in the case of Taxes other than Taxes described in clause (i), the amount of such Taxes allocable to the portion of the taxable period ending on the Closing Date shall be the product of (x) the amount of such Taxes for the entire period and (y) a fraction the numerator of which is the number of calendar days in the period ending with the Closing Date and the denominator of which is the number of calendar days in the entire period.

"Subsidiary" means, with respect to any Person, any corporation or other entity of which a majority of (i) the voting power of the voting equity securities or (ii) the outstanding equity interests is owned, directly or indirectly, by such Person.

"Target Working Capital" means Two Million Dollars (\$1,900,000.00).

"Tax" or "Taxes" means any foreign, federal, state or local income, earnings, profits, gross receipts, franchise, capital stock, net worth, sales, use, value added, occupancy, general property, real property, personal property, intangible property, transfer, fuel, excise, payroll, withholding, employment, unemployment compensation, workers' compensation, social security, retirement, pension plan, estimated, severance, environmental, stamp, occupation, premium, windfall profits, customs, duties or other taxes of any nature or kind whatsoever, together with any interest, additions or penalties with respect thereto and any interest in respect of such additions or penalties.

"<u>Tax Returns</u>" means all returns, information returns, declarations, reports, claims for refund, statements or other documents filed or required to be filed with any Governmental Entity with respect to the determination, assessment, collection or payment of any Tax or in connection with the administration, implementation or enforcement of, or compliance with, any Tax, including any schedule or attachment thereto, and including any amendment thereof.

"<u>Transaction Documents</u>" means this Agreement and all other agreements, instruments, certificates and other documents to be entered into or delivered by any Party pursuant to this Agreement, including, but not limited to the Note, the Escrow Agreement and each Restrictive Covenant Agreement.

" $\underline{\text{Transactions}}$ " means the transactions contemplated by the Transaction Documents.

ARTICLE II

PURCHASE AND SALE OF UNIT; CLOSING; CONSIDERATION

2.1 <u>Purchase and Sale of Units</u>. At the Closing, subject to the terms and conditions of this Agreement, the Sellers shall sell, transfer and deliver the Units to the Purchaser, free and clear of all Liens

(other than any applicable transfer restrictions arising under applicable securities Laws), and the Purchaser shall purchase and accept the Units from the Sellers.

- 2.2 <u>Closing</u>. The closing of the Transactions (the "<u>Closing</u>") will take place at the offices of Wilson Sonsini Goodrich & Rosati, Professional Corporation, 12235 El Camino Real, San Diego, California 92130 (or at such other location as the Parties may agree or via the electronic exchange of execution versions of the Transaction Documents and the signature pages thereto via facsimile or via email by .pdf) on the date hereof (the "<u>Closing Date</u>").
 - 2.3 <u>Closing Consideration</u>. At the Closing, the following transactions shall occur:
- (a) <u>Closing Cash Consideration</u>. The Purchaser shall pay to the Sellers an aggregate amount in cash equal to the Closing Cash Consideration, which amount shall be allocated among the Sellers in accordance with the Consideration Spreadsheet. The Purchaser shall pay the Closing Cash Consideration by wire transfer of immediately available funds to such accounts as the Sellers may direct by written notice delivered to the Purchaser not less than two (2) Business Days prior to the Closing Date.
- (b) <u>Notes</u>. The Purchaser shall deliver to the applicable Seller(s) the Note(s) in accordance with the Consideration Spreadsheet, duly executed by the Purchaser.

2.4 Purchaser Shares.

- (a) On the date that is six (6) months from the Closing Date (the "<u>Purchaser Shares Issuance Date</u>"), the Purchaser shall issue to the Sellers (or their transferees or designees, as permitted pursuant to Section 2.4(b)) the Purchaser Shares in accordance with the Consideration Spreadsheet. The Parties hereto acknowledge and agree that no Seller shall have any voting or dividend rights in the Purchaser Shares prior to the Purchaser Shares Issuance Date.
- (b) The right of any Seller to receive any Purchaser Shares shall not be assignable or transferable by any Seller; provided, that (i) any Seller shall have the right to assign or transfer the right to receive any Purchaser Shares to the equityholder(s) of such Seller and (ii) the equityholder(s) of such Seller shall have the right to assign or transfer the right to receive any Purchaser Shares as follows, if such equityholder is an individual or a trust, and, in the case of clauses (i) and (ii), in each case, subject to compliance with applicable Law, including the Securities Act and the Exchange Act: (A) on death by will or intestacy, or (B) to or for the benefit of any spouse, children, parents, uncles, aunts, siblings or grandchildren of such equityholder (collectively, "Approved Relatives") or to a trust established solely for the benefit of such equityholder or his, her or its Approved Relatives or to one or more family limited partnerships or family limited liability companies where all of the equity interests are owned by such equityholder or his, her or its Approved Relatives (all such Persons under clauses (i) and (ii) shall be referred to as "Permitted Purchaser Share Transferees").

2.5 Escrow Amount.

- (a) <u>Escrow</u>. At the Closing, the Purchaser shall deposit the Escrow Amount into an interest bearing escrow account (the "<u>Escrow Account</u>") established pursuant to the Escrow Agreement to secure the payment obligations of the Sellers under <u>Section 2.6</u> and the indemnification obligations of the Sellers under <u>Section 7.1</u>.
- (b) <u>Escrow Account</u>. The interest, earnings and income that accrue upon the Escrow Amount during the period of time during which it is held in the Escrow Account shall be deemed to become part of the Escrow Amount. The release of funds from the Escrow Account shall be governed by the Escrow Agreement

and the terms of this Agreement; provided, that in the event of a conflict between the terms of the Escrow Agreement and the terms of this Agreement, the terms of this Agreement shall control.

- (c) Release of Escrow Amount. As soon as reasonably practicable (but in any event within two (2) Business Days) following the expiration of the Indemnity Period, the Purchaser and the Sellers' Representative shall cause the Escrow Agent to release to (i) the Purchaser an amount equal to thirty percent (30%) of any income earned on the investment of the Escrow Amount and (ii) to the Sellers (in accordance with their respective Pro Rata Portion) all of the remaining Indemnity Escrow Amount (if any) in excess of an amount (if any) calculated pursuant to the following sentence to satisfy all unresolved indemnity claims for Losses specified in any Claim Notice properly delivered to the Sellers' Representative before the expiration of the Indemnity Period (if any). If any such indemnity claims are unresolved as of the expiration of the Indemnity Period, then a portion of the Indemnity Escrow Amount that equals the total amount of Losses then being reasonably claimed in good faith by the Purchaser Indemnified Persons in all such unresolved indemnity claims shall be retained in the Escrow Account, and as soon as reasonably practicable (but in any event within two (2) Business Days) following resolution of any such indemnity claim, Purchaser and the Sellers' Representative shall cause the Escrow Agent to release to the Sellers (in accordance with their respective Pro Rata Portion), the remaining Indemnity Escrow Amount (if any) not required to satisfy any then still unresolved indemnity claims.
- (d) <u>Tax Matters</u>. The parties agree that, for applicable Tax purposes, the Purchaser shall be treated as the owner of any cash in the Escrow Account, and that all interest on or other taxable income, if any, earned from the investment of such cash pursuant to this Agreement or the Escrow Agreement shall be treated for applicable Tax purposes as earned by the Purchaser. The Parties hereto further intend for the Escrow Amount to qualify for installment sale reporting under Section 453 of the Code, and will treat the Escrow Amount consistent with the foregoing for U.S. federal, state and local income tax purposes, unless otherwise required by applicable Law.
- 2.6 <u>Closing Payment Certificate</u>. The Company has delivered to the Purchaser a certificate (the "<u>Closing Payment Certificate</u>"), signed by an authorized officer of the Company, certifying as to the Company's good faith estimate of (a) the Net Working Capital Amount (such estimate, the "<u>Estimated Net Working Capital Amount</u>"), (b) Closing Indebtedness (such estimate, the "<u>Estimated Closing Indebtedness</u>"), (c) Closing Transaction Expenses (such estimate, "<u>Estimated Closing Cash</u>") and (e) a calculation of the Closing Cash Consideration based thereon, in each case, prepared by the Company in accordance with GAAP and the definitions set forth in this Agreement.

2.7 Final Calculations; Disputes.

- (a) As promptly as practicable, but in any event within sixty (60) days after the Closing Date, the Purchaser will deliver to the Sellers' Representative a reasonably detailed statement setting forth the Purchaser's calculations of the Net Working Capital Amount, Closing Indebtedness, Closing Transaction Expenses, and Closing Cash (the "Closing Statement"). The estimates and calculations set forth in the Closing Statement shall be prepared by the Purchaser in accordance with GAAP and the definitions set forth in this Agreement.
- (b) After delivery of the Closing Statement, the Sellers' Representative and its Representatives shall be permitted reasonable access during normal business hours to review the books and records of the Company and any work papers (including any work papers prepared by the Purchaser's and the Company's accountants, subject to execution of customary access letters) used in the preparation of the Closing Statement. The Sellers' Representative and its Representatives may make reasonable inquiries of the Purchaser and its Representatives regarding questions concerning or disagreements with the Closing Statement arising in

the course of their review thereof, and the Purchaser shall cause any such Representatives to cooperate with and respond to such inquiries.

(c) If the Sellers' Representative has any objections to the Closing Statement, the Sellers' Representative shall deliver a written notice to the Purchaser (the "Dispute Notice") setting forth the basis for such objections in reasonable detail, including, to the extent practicable, (i) each item or amount so objected to by the Sellers' Representative in the Closing Statement, and (ii) the Sellers' Representative's calculation of each such objected item or amount. If a Dispute Notice is not delivered to the Purchaser within thirty (30) days after the Purchaser's delivery of the Closing Statement, the Sellers' Representative shall be deemed to have irrevocably consented to the Closing Statement and the Closing Statement shall be final, binding and non-appealable by the Parties. If a Dispute Notice is delivered to the Purchaser within thirty (30) days after the Purchaser's delivery of the Closing Statement, the Sellers' Representative and the Purchaser shall negotiate in good faith to resolve any objections set forth in the Dispute Notice. Any item set forth in the Closing Statement and not objected to in the Dispute Notice shall be final and binding on the Parties. If the Sellers' Representative and the Purchaser do not reach a final resolution within thirty (30) days after the delivery of the Dispute Notice, the Sellers' Representative or the Purchaser shall submit such dispute to an independent, nationally recognized accounting firm as mutually agreed upon by the Sellers' Representative and the Purchaser (the "Independent Accountant"). If any dispute is submitted to the Independent Accountant, the Independent Accountant shall be requested to render a written determination of the applicable dispute within thirty (30) days after referral of the matter to the Independent Accountant, which determination must be in writing and must set forth, in reasonable detail, the basis therefor and must be based solely on (i) the definitions and other applicable provisions of this Agreement, (ii) a single presentation (which presentations shall be limited to the remaining items in dispute set forth in the Closing Statement and Dispute Notice) submitted by each of the Purchaser and the Sellers' Representative to the Independent Accountant within fifteen (15) days after the engagement thereof (which the Independent Accountant shall forward to the other Party) and (iii) one written response submitted to the Independent Accountant within five (5) Business Days after receipt of each such presentation (which the Independent Accountant shall forward to the other Party), and not on independent review.

(d) The Independent Accountant shall resolve only those matters that remain in dispute after the thirty (30)-day resolution period. It is the intent of the Purchaser and the Sellers' Representative that the process set forth in this Section 2.6 and the activities of the Independent Accountant in connection herewith are not intended to be and, in fact, are not arbitration and that the Independent Accountant shall act as an expert and not an arbitrator and no formal arbitration rules shall be followed (including rules with respect to procedures and discovery). Notwithstanding anything to the contrary in this Agreement, the scope of the Independent Accountant's review of any dispute between the Purchaser and the Sellers' Representative regarding the Dispute Notice and the calculations set forth therein shall be limited solely to the resolution of the remaining disputed portion(s) of such calculations that are set forth in the Dispute Notice, and the Independent Accountant shall not have authority over any other disagreement (including but not limited to questions of law, interpretation of contract (except as it may relate to the interpretation of the Dispute Notice and the calculations set forth therein at dispute). The resolution of the dispute by the Independent Accountant shall be final and binding on, and non-appealable by, the Parties (absent a showing of fraud or manifest error). No ex parte conferences, oral examinations, testimony, depositions, discovery or other form of evidence gathering or hearings shall be conducted or allowed; provided, that at the Independent Accountant's request, or as mutually agreed by the Sellers' Representative and the Purchaser, the Sellers' Representative and the Purchaser may meet with the Independent Accountant so long as representatives of both the Sellers' Representative and the Purchaser are present. The Independent Accountant shall resolve each disputed item by choosing a value not in excess of, nor less than, the greatest or lowest value, respectively, set forth in the presentations (and, if applicable, the responses) delivered to the Independent Accountant pursuant to Section 2.6(c). The Closing Statement shall be modified, if necessary, to reflect such determination of the Independent Accountant. The fees and expenses of the Independent Accountant shall be borne by the Purchaser, on the one hand, and the Sellers, on the other hand,

in inverse proportion as they may prevail on the matters resolved by the Independent Accountant, which proportionate allocation shall be calculated on an aggregate basis based on the relative dollar values of the amounts in dispute and shall be determined by the Independent Accountant at the time the determination is rendered on the merits of the matters submitted to the Independent Accountant.

(e) The Purchaser and the Sellers' Representative agree that the procedures set forth in this <u>Section 2.7</u> for resolving disputes with respect to the Closing Statement shall be the sole and exclusive method for resolving any such disputes; <u>provided</u>, that this provision shall not prohibit either Party from instituting litigation to enforce any final determination of the Independent Accountant in any court of competent jurisdiction in accordance with <u>Section 9.2</u>. The substance of any determination of the Independent Accountant shall not be subject to review or appeal, absent a showing of fraud or manifest error. It is the intent of the Parties to have any final determination by the Independent Accountant proceed in an expeditious manner; however, any deadline or time period contained herein may be extended or modified by the written agreement of the Parties and the Parties agree that the failure of the Independent Accountant to strictly conform to any deadline or time period contained herein shall not be a basis for seeking to overturn any determination rendered by the Independent Accountant which otherwise conforms to the terms of this <u>Section 2.7</u>.

2.8 Adjustments.

- (a) If the Net Working Capital Amount, as finally determined pursuant to Section 2.6, is less than Estimated Net Working Capital Amount, then the Purchaser shall be paid the amount of such deficiency in accordance with Section 2.8(e). If the Net Working Capital Amount, as finally determined pursuant to Section 2.6, is greater than the Estimated Net Working Capital Amount, then the Sellers shall be entitled to receive such excess amount in accordance with Section 2.8(e).
- (b) If the Closing Indebtedness, as finally determined pursuant to Section 2.6, is less than Estimated Closing Indebtedness, then the Sellers shall be entitled to receive such deficiency amount in accordance with Section 2.8(e). If the Closing Indebtedness, as finally determined pursuant to Section 2.6, is greater than the Estimated Closing Indebtedness, then the Purchaser shall be paid the amount of such excess in accordance with Section 2.8(e).
- (c) If Closing Transaction Expenses, as finally determined pursuant to <u>Section 2.6</u>, are less than Estimated Closing Transaction Expenses, then the Sellers shall be entitled to receive such deficiency amount in accordance with <u>Section 2.8(e)</u>. If the Closing Transaction Expenses, as finally determined pursuant to <u>Section 2.6</u>, are greater than the Estimated Closing Transaction Expenses, then the Purchaser shall be paid the amount of such excess in accordance with Section 2.8(e).
- (d) If Closing Cash, as finally determined pursuant to <u>Section 2.6</u>, is greater than Estimated Closing Cash, then the Sellers shall be entitled to receive such excess amount in accordance with <u>Section 2.8(e)</u>. If Closing Cash, as finally determined pursuant to <u>Section 2.6</u>, is less than Estimated Closing Cash, then the Purchaser shall be paid the amount of such deficiency in accordance with <u>Section 2.8(e)</u>.
- (e) Without duplication, all amounts owed pursuant to <u>Section 2.7</u> shall be aggregated, and the net amount (if any) owed by the Purchaser to the Sellers, on the one hand (any such amount, the "<u>Closing Adjustment Surplus Amount</u>"), or by the Sellers to the Purchaser, on the other hand (any such amount, the "<u>Closing Adjustment Shortfall Amount</u>"), is referred to as the "<u>Final Adjustment Amount</u>."
- (f) In the event the Final Adjustment Amount constitutes a Closing Adjustment Surplus Amount, then (x) the Purchaser shall pay to the Sellers an amount in cash equal to the Closing Adjustment Surplus Amount, in accordance with the terms in <u>Section</u> and (y) the Purchaser and the Sellers' Representative

shall deliver joint written instructions to the Escrow Agent instructing the Escrow Agent to deliver to the Sellers (in accordance with their respective Pro Rata Portion) the amount of the Adjustment Escrow Amount.

- (g) In the event the Final Adjustment Amount constitutes a Closing Adjustment Shortfall Amount, then the Purchaser and the Sellers' Representative shall deliver joint written instructions to the Escrow Agent instructing the Escrow Agent to deliver (x) to the Purchaser from the Adjustment Escrow Amount an amount in cash equal to the Closing Adjustment Shortfall Amount; provided, that if the Closing Adjustment Shortfall Amount exceeds the amount of the Adjustment Escrow Amount, then the Sellers shall pay the amount of such excess directly to the Purchaser in accordance with their respective Pro Rata Portions; provided, that in the case of any Seller who holds (or will hold) Purchaser Shares as of the applicable payment date (if applicable), such Seller's payment obligation may be satisfied at such Seller's option by the forfeiture and delivery (if applicable) to the Purchaser of a number of Purchaser Shares (valued for such purpose at the closing price of the Purchaser Common Stock as of the trading day immediately prior to the applicable payment date) having an aggregate value equal to the amount of such Seller's applicable portion of such excess amount, and such Seller and the Purchaser shall take such actions as are reasonably necessary to transfer (or forfeit, if applicable) such number of Purchaser Shares to the Purchaser; provided, further, that if any payment obligations remain outstanding to any Seller under the Note as of such payment date, such Seller's payment obligation may also be satisfied at such Seller's option by a decrease to the outstanding principal amount of the Note owing to such Seller; and (y) to the Sellers (in accordance with their respective Pro Rata Portion) any remaining portion of the Adjustment Escrow Amount, if any, after giving effect to the payment of the amounts contemplated by the foregoing clause (x).
- (h) Payment of the Final Adjustment Amount shall be made within five (5) Business Days after the last date of final determination pursuant to Section 2.6.
- 2.9 <u>Closing Transaction Expenses</u>; <u>Closing Indebtedness</u>. At the Closing, the Purchaser shall pay, or cause to be paid, (a) all then unpaid Closing Transaction Expenses included in the Closing Payment Certificate to the applicable recipient thereof (net of any withholding Taxes, if applicable) (except that any change of control bonuses that constitute unpaid Closing Transaction Expenses shall be paid to the Company on the Closing Date and shall be distributed to the applicable recipient thereof (net of any applicable withholding Taxes) promptly following the Closing in accordance with the Company's ordinary course payroll procedures or a special payroll or other payroll for the payment of such change of control bonuses), and (b) the amount of Indebtedness indicated in the Payoff Letter(s) delivered to Purchaser at least two (2) Business Days prior to the Closing in accordance with the wire instructions set forth therein.
- 2.10 <u>Reliance on Consideration Spreadsheet</u>. The Purchaser and the Escrow Agent shall be permitted to rely, without further inquiry, on the Consideration Spreadsheet in making any payments to the Sellers under this Agreement. The Consideration Spreadsheet shall be prepared in accordance with the Organizational Documents of the Company.
- 2.11 <u>Withholding</u>. The Company, the Purchaser and the Escrow Agent shall be entitled to deduct and withhold from any consideration payable pursuant to this Agreement to any Person such amounts as are required to be deducted or withheld therefrom under any provision of U.S. federal, state, local or non-U.S. Tax Law or under any other applicable legal requirement. To the extent such amounts are so deducted or withheld, such amounts shall be properly remitted to the appropriate Tax authority by the Company, the Purchaser and the Escrow Agent, as the case may be, and treated for all purposes as having been paid to the Person to whom such amounts would otherwise have been paid.
 - 2.12 Registration of Purchaser Shares.

- (a) Promptly (but in any event no later than thirty (30) days) after the Closing Date (the "Filing Deadline"), the Purchaser shall prepare and file with the U.S. Securities and Exchange Commission (the "SEC") a Form S-3 registration statement covering the resale of all of the Purchaser Shares (the "Registration Statement"). The Purchaser shall use its reasonable best efforts to, as soon as reasonably practicable following the filing of the Registration Statement, but in no event later than 20 days following the filing of the Registration Statement (or 60 days following the filing of the Registration Statement solely in the event of a full review of the Registration Statement by the SEC), cause the Registration Statement to be declared effective by the SEC.
- (b) After the Registration Statement has been declared effective by the SEC, the Purchaser shall use its reasonable best efforts to keep the Registration Statement continuously effective until the earlier of December 31, 2022 and the date that all Purchaser Shares covered by the Registration Statement are sold or are able to be sold by the holders thereof by relying on Rule 144 under the Securities Act without any restriction, including volume limitation. Without limiting the foregoing, the Purchaser shall (i) prepare and file with the SEC such amendments to the Registration Statement and amendments or supplements to the prospectus used in connection therewith as may be reasonably necessary to comply with the provisions of the Securities Act with respect to the sale or other disposition of all Purchaser Shares included in the Registration Statement and (ii) register or qualify the Purchaser Shares included in the Registration Statement under the applicable state securities or blue sky laws.
- (c) The Purchaser will pay all expenses associated with the Registration Statement and any amendments or supplements thereto, and any actions or filings necessary to maintain the effectiveness of the Registration Statement, including filing and printing fees, the fees and expenses of the Purchaser's counsel and accounting fees and expenses, costs associated with clearing the Registrable Securities for sale under applicable state securities laws and listing fees, but excluding discounts, commissions, fees of underwriters, selling brokers, dealer managers or similar securities industry professionals with respect to the Purchaser Sales being sold.
- (d) The Purchaser covenants that it will file the reports required to be filed by it under the Securities Act and the Exchange Act (as defined below) and the rules and regulations adopted by the SEC thereunder in accordance with the requirements of the Securities Act and the Exchange Act and it will take such further action as Sellers may reasonably request to make available adequate current public information with respect to the Purchaser meeting the current public information requirements of Rule 144(c) under the Securities Act, to the extent required to enable Sellers to sell all Purchaser Shares without registration under the Securities Act within the limitation of the exemptions provided by (i) Rule 144 under the Securities Act or (ii) any similar rule or regulation hereafter adopted by the SEC. In addition, the Purchaser agrees, at its sole expense, to cause its legal counsel to provide a legal opinion to the Purchaser's transfer agent that is in form and substance sufficient for such transfer agent to remove all applicable legends in connection with, and to otherwise facilitate, the sale of Purchaser Shares under Rule 144 under the Securities Act.
- (e) The Purchaser shall indemnify and hold harmless, the Sellers Indemnified Persons, from and against any Losses to which any Seller Indemnified Person may become subject (under the Securities Act or otherwise) to the extent such Losses arise out of, directly or indirectly, any untrue statement of a material fact contained in the Registration Statement or any other document filed in accordance with this Section 2.12(e), or any omission to state therein a fact required to be stated therein or necessary to make the statements therein not misleading; provided, however, that the Purchaser will not be liable in any such case to the extent that any such Losses arising out of, directly or indirectly, any untrue statement or omission, made in reliance upon and in conformity with written information furnished to the Purchaser by or on behalf of any Seller specifically for use in the preparation of the Registration Statement. The procedures set forth in Section 7.5 shall apply to any claim for indemnification made pursuant to this Section 2.12(e), mutatis mutandis.

- (f) At the written request by any Seller to the Purchaser, the Purchaser agrees to use its reasonable best efforts to facilitate and coordinate one or more underwritten sale(s) of the Purchaser Shares by such Seller to one or more third parties (including by way of a "block-trade"), as and when requested in writing by such Seller following the Purchaser Shares Issuance Date.
- (g) The term (i) "Sellers" for purposes of this section shall mean (A) the Sellers (as defined above) and (B) Permitted Purchaser Share Transferees who are transferred the right to receive Purchaser Shares in accordance with this Section, and who shall have third-party beneficiary rights under this section; and (ii) "Purchaser Shares" for purposes of this section shall mean (A) the shares of Purchaser Common Stock issued to the Sellers under this Agreement and (B) any securities issued or issuable upon any stock split, dividend or other distribution, recapitalization or similar event with respect to the foregoing in clause (ii)(A).

ARTICLE III

REPRESENTATIONS AND WARRANTIES OF THE PURCHASER

The Purchaser hereby represents and warrants to each Seller and to the Company that the following representations and warranties are true and correct as of the date hereof.

3.1 Organization; Corporate Power and Authorization. The Purchaser is a Pennsylvania corporation duly organized, validly existing and in good standing under the Laws of the state of its incorporation. The Purchaser has the requisite corporate power and authority necessary to enter into, deliver and carry out its obligations pursuant to each of the Transaction Documents to which it is a party. The Purchaser is duly qualified to transact business and is in good standing in each jurisdiction in which the failure to so qualify would cause a Purchaser Material Adverse Change. The Purchaser's execution, delivery and performance of each Transaction Document to which it is a party has been duly authorized by the Purchaser and no other corporate proceeding on the part of the Purchaser is necessary to authorize the Transaction Documents and the Transactions, and the Purchaser has duly executed and delivered this Agreement and will have, as of the Closing Date, duly executed and delivered each other Transaction Document to which it is a party.

3.2 Binding Effect; Noncontravention; No Approvals.

- (a) Each Transaction Document to which the Purchaser is a party constitutes, or when executed will constitute, a valid and binding obligation of the Purchaser enforceable against the Purchaser in accordance with its terms, subject to the effect of (i) applicable bankruptcy, insolvency, reorganization, moratorium or other similar laws now or hereafter in effect relating to rights of creditors generally and (ii) rules of law and equity governing specific performance, injunctive relief and other equitable remedies.
- (b) The execution, delivery and performance by the Purchaser of the Transaction Documents to which the Purchaser is a party and the consummation of the Transactions do not and shall not (with or without notice or lapse of time or both): (i) conflict with or result in a breach of the terms, conditions or provisions of any Organizational Document of the Purchaser; (ii) result in the imposition of any Lien upon any of the properties or assets of the Purchaser; (iii) result in a material breach or material violation by the Purchaser of any of the terms, conditions or provisions of any Law or Order to which the Purchaser or any of its properties or assets is subject; or (iv) require any Approval to be obtained by the Purchaser which has not been obtained by the Purchaser as of the date hereof.

- 3.3 <u>Broker Fees</u>. The Purchaser has not used any broker or finder, or entered into any Contract with any Person which would result in the obligation of the Company or any Seller to pay any fees or commissions to any broker, finder, or agent, with respect to the Transactions.
- 3.4 <u>Investment</u>. The Purchaser is acquiring the Units for its own account, for investment only, and not with a view to any resale or public distribution thereof. The Purchaser shall not offer to sell or otherwise dispose of the Units in violation of any Law applicable to any such offer, sale or other disposition. The Purchaser acknowledges that (a) the Units have not been registered under the Securities Act, or any state securities Laws; (b) there is no public market for the Units and there can be no assurance that a public market will develop; and (c) the Purchaser must bear the economic risk of its investment in the Units for an indefinite period of time. The Purchaser is an "accredited investor" within the meaning of Rule 501 of the Securities Act as presently in effect, and has knowledge and experience in financial and business matters such that it is capable of evaluating the merits and risks of acquiring and holding the Units.
- 3.5 No Litigation. There is no Action pending or, to the Purchaser's knowledge, threatened against the Purchaser or its properties, assets or businesses, or Order to which the Purchaser is subject which would result in a Purchaser Material Adverse Change.
- 3.6 Purchaser SEC Filings. Purchaser has filed all forms, reports, statements and documents (including all exhibits, schedules and annexes thereto) required to be filed by it with the SEC since the date that is twelve (12) months prior to the Closing Date, including any amendments or supplements thereto (collectively, the "Purchaser SEC Reports"). As of their respective dates (and, if amended or superseded by a filing prior to the date of this Agreement, then on the date of such filing) none of the Purchaser SEC Reports contained at the time of filing, and none of the Purchaser SEC Reports filed after the date of this Agreement and prior to the Closing Date (and, if amended or superseded by a filing prior to the Closing Date, then on the date of such filing) will contain, any untrue statement of a material fact or omitted (or will have omitted) to state a material fact required to be stated therein or necessary in order to make the statements therein, in the light of the circumstances under which they were made, not misleading, and all of the Purchaser SEC Reports complied in all material respects with the requirements of the Securities Act and the Securities Exchange Act of 1934, as amended, and the applicable rules and regulations promulgated thereunder (the "Exchange Act"), as the case may be. Since the date of the last Purchaser SEC Report, there has not been the occurrence or non-occurrence of any event, the occurrence or non-occurrence of which resulted in, or would reasonably be likely to result in, a material adverse effect with respect to Purchaser. The financial statements (including any related notes) contained in Purchaser SEC Reports (collectively, the "Purchaser Financial Statements") (i) complied as to form in all material respects with the published rules and regulations of the SEC applicable thereto and (ii) were prepared in accordance with GAAP, consistently applied, and present fairly in all material respects the consolidated financial position and results of operations of Purchaser and its Subsidiaries (taken as a whole) as of the times and for the periods referred to therein, subject in the case of the unaudited financial statements to the absence of footnote disclosures and other presentation items and changes resulting from normal year-end adjustments. To Purchaser's knowledge, none of the Purchaser SEC Reports is the subject of ongoing SEC review and there are no inquiries or investigations by the SEC or any internal investigations pending or threatened, in each case regarding any accounting practices of Purchaser or any of its Subsidiaries.
- 3.7 <u>Purchaser Shares</u>. At the Closing, (a) Purchaser will have an adequate amount of authorized shares of Purchaser Common Stock to effect the issuance to the Sellers of the Purchaser Shares in accordance with this Agreement and (b) all of the Purchaser Common Shares (i) will be duly authorized, validly issued, fully paid and nonassessable, (ii) will be issued in compliance with applicable securities Laws or exemptions therefrom, (iii) will not be subject to, or issued in violation of, any purchase option, call option, right of first refusal, preemptive right or any similar right, and (iv) will be free of all Liens, other than restrictions imposed under applicable securities Laws.

- 3.8 Exchange Act Compliance. The Purchaser Common Stock is registered pursuant to Section 12(b) of the Exchange Act and listed on NASDAQ. The issuance of the Purchaser Shares to the Sellers under this Agreement does not require stockholder approval of Purchaser, including, without limitation, pursuant to the rules and regulations of NASDAQ.
- 3.9 Form S-3 Eligibility. The Purchaser is eligible to register the Purchaser Shares to be issued to the Sellers under this Agreement for resale using Form S-3 promulgated under the Securities Act.
- 3.10 <u>Number of Purchaser Shares</u>. The number of Purchaser Shares represents approximately 19.999% of the issued and outstanding shares of capital stock of the Purchaser as of immediately prior to the Closing.

ARTICLE IV

REPRESENTATIONS AND WARRANTIES OF THE SELLERS

Each Seller, as to such Seller, hereby represents and warrants to the Purchaser that the following representations and warranties are true and correct as of the date hereof.

4.1 <u>Power and Authorization</u>. If such Seller is not an individual, such Seller is validly existing and in good standing under the Laws of the state of its formation, as applicable, and has the requisite entity power and authority necessary to enter into, deliver and carry out its obligations pursuant to each of the Transaction Documents to which it is a party. If such Seller is not an individual, such Seller's execution, delivery and performance of each Transaction Document to which it is a party has been duly authorized by such Seller and no other entity proceeding on the part of such Seller is necessary to authorize the Transaction Documents to which such Seller is a party. Such Seller has duly executed and delivered this Agreement and will have, as of the Closing Date, duly executed and delivered each other Transaction Document to which it is a party.

4.2 Binding Effect and Noncontravention; Approvals.

- (a) Each Transaction Document to which such Seller is a party constitutes, or when executed will constitute, a valid and binding obligation of such Seller enforceable against such Seller in accordance with its terms, subject to the effect of (i) applicable bankruptcy, insolvency, reorganization, moratorium or other similar laws now or hereafter in effect relating to rights of creditors generally and (ii) rules of law and equity governing specific performance, injunctive relief and other equitable remedies.
- (b) The execution, delivery, and performance by such Seller of the Transaction Documents to which such Seller is a party and the consummation of the Transactions do not and shall not (with or without notice or lapse of time or both): (i) conflict with or result in a breach of the terms, conditions, or provisions of any Organizational Document of such Seller (if applicable); (ii) result in a material breach or material violation of any of the terms, conditions or provisions of any Law or Order to which such Seller or any of its properties or assets is subject; (iii) result in a material breach or material violation of any of the terms, conditions or provisions of any Contract to which such Seller is a party or by which such Seller's assets or properties are bound; or (iv) require any Approval to be obtained by such Seller which has not been obtained by such Seller as of the date hereof.
- 4.3 <u>Units</u>. Such Seller holds of record, owns beneficially and has good and valid title to the Units set forth opposite such Seller's name on <u>Section 5.3(a)</u> of the Company Disclosure Schedule, as applicable, free and clear of any and all Liens, other than any applicable transfer restrictions arising under applicable securities Laws or under the Operating Agreement. Such Seller is not a party to any voting trust, proxy or other agreement

or understanding with respect to the voting of its Units. Such Seller is not a party to any option, warrant, purchase right or other contract that would require such Seller to sell, transfer or otherwise dispose of its Units (other than to the Purchaser pursuant to the terms of this Agreement). Except for the Units set forth opposite such Seller's name on Section 5.3(a) of the Company Disclosure Schedule, such Seller does not own any securities of the Company or any right to acquire any securities of the Company.

- 4.4 No Litigation. There is no Action pending or, to such Seller's Knowledge, threatened in writing, against such Seller (a) challenging or seeking to restrain, delay or prohibit any of the Transactions, (b) preventing such Seller from performing in all material respects such Seller's obligations under the Transaction Documents to which such Seller is a party, or (c) related to the Company.
- 4.5 <u>Broker Fees</u>. Such Seller has not used any broker or finder, or entered into any Contract with any Person which would result in the obligation of the Company to pay any fees or commissions to any broker, finder, or agent, with respect to the Transactions.

4.6 Purchaser Shares.

- (a) Such Seller (i) is an "accredited investor" (within the meaning of Rule 501(a)(1), Rule 501(a)(2), Rule 501(a)(3), or Rule 501(a) (7) under the Securities Act); (ii) will receive the Purchaser Shares only for such Seller's own account and not for the account of others and such Seller has full investment discretion with respect to each such account, and the full power and authority to make the acknowledgements, representations and agreements herein on behalf of each owner of each such account, provided, however, that this does not limit the transfer of the right to receive any of the Purchaser Shares pursuant to Section 2.4(b); and (iii) is not receiving the Purchaser Shares with a view to, or for offer or sale in connection with, any distribution thereof in violation of the Securities Act.
- (b) Such Seller understands that the Purchaser Shares are being offered in a transaction not involving any public offering within the meaning of the Securities Act and that the Purchaser Shares have not been registered under the Securities Act or any state securities Law in reliance on the availability of an exemption from such registration. Such Seller understands that the Purchaser Shares may not be offered, resold, transferred, pledged or otherwise disposed of by such Seller absent an effective registration statement under the Securities Act, except (i) to the Purchaser or a subsidiary thereof, or (ii) pursuant to an applicable exemption from the registration requirements of the Securities Act, and, in each of cases (i) and (ii), in accordance with any applicable securities Laws of the states and other jurisdictions of the United States, and that any certificates or book entries representing the Purchaser Shares shall contain a legend to such effect. Such Seller acknowledges that the Purchaser Shares will not be eligible for resale pursuant to Rule 144A promulgated under the Securities Act. Such Seller understands and agrees that the Purchaser Shares will be subject to transfer restrictions and, as a result of these transfer restrictions, such Seller may not be able to readily resell the Purchaser Shares and may be required to bear the financial risk of an investment in the Purchaser Shares for an indefinite period of time. Such Seller understands that such Seller has been advised to consult legal counsel prior to making any offer, resale, pledge or transfer of any of the Purchaser Shares.
- (c) Such Seller acknowledges that such Seller is aware that there are substantial risks incident to the purchase and ownership of the Purchaser Shares. Such Seller is a sophisticated accredited investor and has such knowledge and experience in financial and business matters, and in investing in private placement securities, as to be capable of evaluating the merits and risks of an investment in the Securities. At the time of making such Seller's investment decision, such Seller has had access to such financial and other information concerning the Purchaser and its subsidiaries as such Seller deemed necessary or desirable in making a decision to receive the Purchaser Shares, including an opportunity to ask questions and receive answers from officers of the Purchaser and to obtain additional information necessary to verify the accuracy of any information furnished

to such Seller or to which such Seller had access. Such Seller has independently made such Seller own analysis and decision to acquire the Securities and determined based on such Seller own independent review, and such professional advice from such Seller own advisors (including as to tax, legal and accounting matters) as such Seller may deem appropriate, that such Seller's receipt of the Purchaser Shares (i) is consistent with such Seller's financial needs, objectives and condition, (ii) complies with all investment policies, guidelines and other restrictions that are applicable to such Seller, (iii) does not and will not violate any law, rule, regulation, agreement or other obligation to which such Seller is bound (assuming the accuracy of the Company's representations and warranties contained herein), and (iv) is a fit, proper and suitable investment for such Seller, notwithstanding the risks associated with a purchase of the Purchaser Shares and ownership of the Purchaser Shares. Without limitation of the foregoing, such Seller acknowledges that the Purchaser and its advisors are not providing tax advice of any kind with respect to the transactions contemplated hereby, including the tax consequences relating to receipt of the Purchaser Shares contemplated hereby, and such Seller acknowledges that such Seller has been advised to consult with such Seller's own tax advisor regarding the tax consequences of receiving Purchaser Shares pursuant to this Agreement.

(d) Such Seller is not (i) a person named on the List of Specially Designated Nationals and Blocked Persons administered by the U.S. Treasury Department's Office of Foreign Assets Control ("OFAC") or in any Executive Order issued by the President of the United States and administered by OFAC ("OFAC List"), or a person prohibited by any OFAC sanctions program or (ii) a Designated National as defined in the Cuban Assets Control Regulations, 31 C.F.R. Part 515. Such Seller agrees to provide law enforcement agencies, if requested thereby, such records as required by applicable law, provided that such Seller is permitted to do so under applicable law. Neither such Seller nor any of such Seller's Affiliates is subject to any of the "Bad Actor" disqualifications described in Rule 506(d)(1)(i) to (viii) under the Securities Act.

ARTICLE V

REPRESENTATIONS AND WARRANTIES OF THE COMPANY

Except as expressly set forth on the Company Disclosure Schedule, the Company hereby represents and warrants to the Purchaser that the following representations and warranties are true and correct as of the date hereof.

5.1 Organization; Qualification; Limited Liability Company Power and Authorization. The Company is a limited liability company duly formed, validly existing and in good standing under the Laws of the State of California. The Company is duly authorized to (a) conduct business in the manner in which its business is currently being conducted; (b) to own and use its assets in the manner in which its assets are currently owned and used; and (c) to execute and deliver the Transaction Documents to which it is a party and perform its obligations hereunder and thereunder. The Company is duly qualified to transact business and is in good standing in each jurisdiction in which the failure to so qualify or be in good standing would cause a Company Material Adverse Change. The Company has the requisite limited liability company power and authority necessary to enter into, deliver and carry out its obligations pursuant to each of the Transaction Documents to which it is a party. The execution, delivery and performance of each Transaction Document by the Company to which it is a party has been duly authorized by the Company and no other limited liability company proceeding on the part of the Company is necessary to authorize the Transaction Documents and the Transactions, and the Company has duly executed and delivered this Agreement and the Company will have, as of the Closing Date, duly executed and delivered each other Transaction Document to which it is a party. The Company has not conducted business under any d/b/a or fictitious name, other than its company name.

5.2 Binding Effect and Noncontravention; Approvals.

- (a) Each Transaction Document to which the Company is a party constitutes, or when executed will constitute, a valid and binding obligation of the Company enforceable against the Company in accordance with its terms, subject to the effect of (i) applicable bankruptcy, insolvency, reorganization, moratorium or other similar laws now or hereafter in effect relating to rights of creditors generally and (ii) rules of law and equity governing specific performance, injunctive relief and other equitable remedies.
- (b) Except as otherwise set forth in Section 5.2 of the Company Disclosure Schedule, the execution, delivery, and performance by the Company of the Transaction Documents to which the Company is a party and the consummation of the Transactions do not and will not (with or without notice or lapse of time or both): (i) conflict with or result in a breach of the terms, conditions, or provisions of the Organizational Documents of the Company; (ii) result in the imposition of any material Lien (other than a Permitted Lien) upon any of the properties or assets of the Company; (iii) result in a material breach or violation by the Company of any of the terms, conditions or provisions of any Law or Order to which the Company is subject or by which the Company's assets or properties are bound; (iv) conflict with, or result in any violation of or default under, or give rise to a right of termination, cancellation, acceleration or modification of any obligation to or loss of a benefit under, or to increased, additional, accelerated or guaranteed material rights or entitlements of any Person under, any Material Contract, in each case except where such occurrence set forth in this clause (iv) would not result in a Company Material Adverse Change; or (v) require any material Approval to be obtained by the Company which has not been obtained by the Company as of the date hereof.

5.3 Capitalization.

- (a) The Sellers own the Units set forth opposite each such Seller's name on Section 5.3(a) of the Company Disclosure Schedule and all such Units set forth on Section 5.3(a) of the Company Disclosure Schedule represent 100% of the issued and outstanding equity securities of the Company. All of the outstanding Units have been duly authorized, are validly issued, fully paid, and non-assessable, have been offered, issued, sold and delivered in compliance with all applicable federal and state securities Laws and are held of record and beneficially by the Sellers and are not subject to or have not been issued in violation of any preemptive or subscription rights or any option, call option, right of first refusal, or any similar right under any provision of applicable Law, any Organizational Document of the Company or any Contract to which the Company is a party or otherwise bound, in each case that has not been waived. There are no outstanding or authorized options, warrants, puts, calls, purchase rights, subscription rights, conversion rights, exchange rights, restricted stock awards, or other contracts or commitments that would require the Company to issue, sell or otherwise cause to become outstanding any of its equity. There are no outstanding or authorized equity appreciation, phantom equity, profit participation, earnout rights or similar rights with respect to the Company. There are no, and at the Closing there shall not be any, voting agreements, proxies, registration rights, rights of first refusal, preemptive rights, co-sale rights or other restrictions applicable to any outstanding securities of the Company. There are no outstanding obligations, contingent or otherwise, of the Company to repurchase, redeem, exchange or otherwise acquire the Units or any other equity interests of the Company.
- (b) There is no Liability for dividends accrued and unpaid by the Company. All prior repurchases of equity consummated by the Company, if any, were effected in compliance with all Laws and applicable Contracts, and there are no contingent or other Liabilities remaining in connection therewith.
 - 5.4 <u>Subsidiaries</u>. The Company has no Subsidiaries and does not own any equity interests in any other Person.

5.5 Financial Statements.

- (a) Section 5.5(a) of the Company Disclosure Schedule sets forth the Company's (i) unaudited consolidated balance sheet of the Company as of June 30, 2021 (the "Latest Balance Sheet") and the related consolidated statements of income and cash flows for the six-month period then ended (collectively, the "Unaudited Financial Statements"), and (ii) audited balance sheet and statements of income, members' equity and cash flows of the Company for each of the fiscal years ended December 31, 2020, December 31, 2019 and December 31, 2018 (collectively, the "Audited Financial Statements"), and together with the Unaudited Financial Statements, the "Financial Statements"). The Financial Statements are complete and correct in all material respects, have each been prepared from the books and records of the Company and in accordance with GAAP, consistently applied, and present fairly in all material respects the consolidated financial condition and results of operations of the Company as of the times and for the periods referred to therein, subject in the case of the unaudited financial statements to (x) the absence of footnote disclosures and other presentation items and (y) changes resulting from normal year-end adjustments which are not material in amount or significance in any individual case or in the aggregate.
- (b) Except as set forth in the Financial Statements, the Company has no material Liabilities or obligations, contingent or otherwise, other than: (i) current Liabilities incurred in the ordinary course of business, consistent with past practice, subsequent to the date of the Latest Balance Sheet (none of which is a liability resulting from a violation of Law, breach of contract, breach of warranty, tort, infringement or misappropriation); (ii) obligations under any Contract to which the Company is a party that has been made available to the Purchaser prior to the date hereof and that are expressly set forth in and are identifiable by reference to the text of such Contract; and (iii) other commitments that are disclosed on Section 5.5(b) of the Company Disclosure Schedule.
- (c) The Company maintains a standard system of accounting established and administered in accordance with GAAP and sufficient to provide reasonable assurances that (i) all transactions are executed in accordance with management's general or specific authorization, (ii) all transactions are recorded as necessary to permit the preparation of the Financial Statements in conformity with GAAP and to maintain proper accountability for items, (iii) access to its property and assets is permitted only in accordance with management's general or specific authorization and (iv) the recorded accountability for items is compared with the actual levels at reasonable intervals and appropriate action is taken with respect to any differences. The Company is not party to any "off balance sheet arrangements" (as defined in the Exchange Act).
- 5.6 <u>Absence of Certain Developments</u>. Since the date of the Latest Balance Sheet through the date hereof, (i) the Company has conducted its business in the ordinary course of business consistent with past practice, (ii) no Company Material Adverse Change has occurred, and (iii) the Company has not:
- (a) declared, set aside or paid any dividends on or in respect of, any of its equity, or otherwise make any payments to its owners in their capacity as such, (ii) split, combined or reclassified any of its capital equity, (iii) purchased, redeemed or otherwise acquired any of its equity or any other securities thereof or any rights, warrants or options to acquire any such equity or other securities or (iv) adopted a plan of or effect any complete or partial liquidation or adopt resolutions providing for or authorizing such liquidation or adopt a plan of or effect any dissolution, merger, consolidation, restructuring, recapitalization or reorganization of the Company;
- (b) authorized for issuance, issued, delivered, sold, pledged, disposed of or granted (i) any Company equity, (ii) any other equity or voting securities, or (iii) any securities convertible into or exchangeable for, or any options, warrants or rights to acquire, any such equity, equity or voting securities or convertible or exchangeable securities;

- (c) amended any Organizational Document of the Company;
- (d) acquired or agreed to acquire (i) by merging or consolidating with, or by purchasing all or a substantial portion of the assets of, any business of any Person or division thereof, (ii) any assets other than those acquired in the ordinary course of business, or (iii) equity interests in any Person;
- (e) except as required by the terms of any Employee Plan or any applicable Law, (i) granted to any employee, director or other service provider of the Company any increase in compensation or benefits, except for any grants made in the ordinary course of business, (ii) granted to any employee, director or other service provider of the Company any increase in severance or termination pay, (iii) entered into or amended any employment, consulting, severance or termination agreement with any Person, except for any employment or consulting agreements entered into in the ordinary course of business, (iv) established, adopted, entered into, terminated or amended in any material respect any Employee Plan (or arrangement that, had it been in existence on the date of this Agreement, would be an Employee Plan), (v) taken any action to accelerate any rights or benefits, or made any material determinations not in the ordinary course of business, under any collective bargaining agreement or Employee Plan, (vi) loaned or advanced money or other property to any employee of the Company, other than employee advances for travel, business or entertainment expenses made in the ordinary course of business, or (vii) failed to make contributions to an Employee Plan in accordance with past practice;
- (f) made any change in accounting methods, principles or practices affecting the reported consolidated assets, Liabilities or results of operations of the Company, except insofar as may have been required by a change in GAAP, or changed or modified any cash management, credit collection or payment policies, practices or procedures (including any acceleration in the collection or receivables or delay in the payment of payables);
- (g) sold, leased, licensed, mortgaged, encumbered or otherwise disposed of or permit to become subject to any Lien, other than a Permitted Lien, any properties or assets of the Company, tangible or intangible, other than in the ordinary course of business;
 - (h) entered into any transaction with an Affiliate or other Related Party;
- (i) (i) incurred any Indebtedness or guaranteed any Indebtedness of another Person or (ii) made any loans, advances or capital contributions to, or investments in, any other Person (other than employee advances for travel, business or entertainment expenses made in the ordinary course of business in accordance with the Company's customary expense reimbursement policies) or (iii) canceled or compromised any debt or material claim of the Company in any material respect;
- (j) made or changed any Tax election, adopted or changed any Tax accounting method, consented to the extension or waiver of the limitations period applicable to any Tax claim or assessment, entered into any closing agreement with respect to any Taxes, or settled or compromised any Tax Liability;
 - (k) initiated, settled or compromised any Action;
- (l) terminated or extended, waived, modified, rescinded or made (or agreed to) any material amendments to, or waived or accelerated any rights or benefits under, any Material Contract;
 - (m) made or entered into any commitment for material capital expenditures of the Company; or

- (n) authorized the Company to do any of the foregoing, or entered into any Contract to do any of the foregoing.
- 5.7 Assets. Except as set forth on Section 5.7 of the Company Disclosure Schedule, the Company has good and valid title to, or a valid leasehold interest in, all of its assets, free and clear of any and all Liens, other than Permitted Liens. After giving effect to the divestiture of the Patents to IriSys on or prior to the date hereof, all such assets collectively represent all assets necessary for the conduct of the business of the Company as currently conducted and in the manner conducted during the periods covered by the Financial Statements. Except for leased items, no Person other than the Company owns any interest in any such assets. All such assets are (i) in good operating condition and repair for the conduct of the business of the Company as of immediately after the Closing, (ii) free of defects, normal wear and tear excepted, and (iii) sufficient for the uses to which they are being put, except as would not, individually or in the aggregate, be material to the business of the Company, taken as a whole.

5.8 Tax Matters.

- (a) The Company has filed all income and other material Tax Returns that it is required to file. All such Tax Returns are true, complete and correct in all material respects. All Taxes due and owing by the Company (whether or not shown on any Tax Return) have been timely paid. The Company is not currently the beneficiary of any extension of time within which to file any Tax Return that has continuing effect, other than any such extension that was automatically granted.
- (b) None of the Company's Tax Returns are currently the subject of any audit or proceeding by any Tax authority. The Company has not been notified in writing of any request for an audit of, or a proceeding related to, the Company's Tax Returns.
- (c) The Company is not a party to any Tax allocation, indemnity or sharing agreement (other than any commercial agreement entered into in the ordinary course of business the primary purpose of which is not Tax related). The Company has never been a member of an Affiliated Group filing a consolidated federal Tax Return. The Company has no liability for Taxes of any Person (other than the Company) under Treasury Regulations Section 1.1502-6 (or any corresponding provision of state, local or foreign Law), as transferee or successor, by contract or otherwise (other than pursuant to a commercial agreement entered into in the ordinary course of business the primary purpose of which is not Tax related).
- (d) The Company has not agreed to make, and the Company is not required to make, any adjustment under Section 481(a) of the Code (or any similar provision of applicable state, local or foreign Law) by reason of a change in accounting method or otherwise occurring prior to or on the Closing Date. The Company will not be required to include any material item of income in, or exclude any material item of deduction from, taxable income for any taxable period (or portion thereof) ending after the Closing Date as a result of any: (i) "closing agreement" as described in Section 7121 of the Code (or any corresponding provision of state, local or foreign income Tax Law) entered into prior to the Closing; (ii) installment sale or open transaction disposition made prior to the Closing; (iii) prepaid amount received outside of the ordinary course of business prior to the Closing; or (iv) any election under Section 108(i) of the Code.
- (e) The Company has timely paid or withheld with respect to its employees, independent contractors, creditors, members and other third parties (and timely paid over any withheld amounts to the appropriate Tax authority) all Taxes required to be withheld or paid and the Company has complied in all material respects with all information reporting and backup withholding provisions of applicable Law.

- (f) There is no Tax deficiency outstanding, assessed or proposed in writing against the Company, nor has the Company executed any waiver of any statute of limitations on or extension of the period for the assessment or collection of any Tax that remains in effect. No adjustment relating to any Tax Return filed by the Company has been proposed in writing by any Tax authority to the Company. No written claim has been made by an authority in a jurisdiction where the Company does not file Tax Returns that it is or may be subject to taxation by that jurisdiction.
 - (g) There are no Liens on the assets of the Company relating to or attributable to Taxes other than Permitted Liens.
- (h) For purposes of the Code, the Company is and has been classified as a partnership as described in Treasury Regulations Section 301.7701-3(b)(1)(i) during all periods prior to the Closing.
- (i) The unpaid Taxes of the Company did not, as of the date of the Latest Balance Sheet, exceed the reserve for Tax liability set forth on the face of the Financial Statements.
- (j) No Seller is a "foreign person" as that term is used in Treasury Regulations Section 1.1445-2. The Company is not, nor has it been, a United States real property holding corporation (as defined in Section 897(c)(2) of the Code) during the applicable period specified in Section 897(c)(1)(a) of the Code.
- (k) The Company is not, and has not been, a party to, or promoter of, a "reportable transaction" within the meaning of Section 6707A(c)(1) of the Code and Treasury Regulations Section 1.6011-4(b).
- (l) The Company (i) has not made any election under the Partnership Audit Provisions to have them apply to the Company with respect to any period (or portion thereof) before January 1, 2018, and (ii) has not made any election to opt out of the Partnership Audit Provisions with respect to any period (or portion thereof) after December 31, 2017.
- (m) Except as set forth on Section 5.8(n) of the Company Disclosure Schedule, the Company has not (i) deferred payment of any payroll Taxes pursuant to Section 2302 of the CARES Act, (ii) claimed the employee retention credit pursuant to the Section 2301 of the CARES Act or other credit pursuant to any Covid-19 Law, or (iii) amended any income Tax return for a taxable year prior to 2021 in order to carry back a net operating loss to such year.
- (n) Neither the Company nor any related persons within the meaning of Section 197 of the Code has held or used, at any time on or prior to August 10, 1993, any Section 197 intangible described in subparagraph (A) or (B) of Section 197(d)(1) of the Code.

5.9 Environmental Matters.

(a) The Company has during the past three (3) years been, and is currently, in compliance, in all material respects, with all applicable Environmental Laws. The Company has not received any written notice of a potential violation of applicable Environmental Laws or any potential Liability pursuant to Environmental Law. There is no Action pending or, to the Knowledge of the Company, threatened against the Company pursuant to Environmental Laws. Except in compliance with Environmental Laws or in a manner that would not be reasonably likely to result in a material Liability to the Company, no Hazardous Material has been Released at, on, under or are present on any real property (including the soil, groundwater, surface water on or under the property) currently or formerly owned, operated, occupied, or leased by the Company (including the soil, groundwater, surface water on or under the property). The Company has made available to the

Purchaser all Environmental Permits, environmental site assessment reports, evaluations, or investigations and environmental audits in their possession or control relating to the operation of the business or any real property currently or formerly owned, leased, occupied, or operated by the Company at any time, and any other material documents related to the Company's compliance with Environmental Law or any potential Liability pursuant to Environmental Law. The Company has not agreed to indemnify or hold harmless or assume any actual or potential Liability under any Environmental Laws of any other Person. Neither this Agreement nor the consummation of this Transaction will result in any obligations for site investigation or cleanup, or notification to or consent of any Governmental Entity or any other Person, pursuant to any of the so-called "transaction-triggered" or "responsible property transfer" Environmental Laws.

(b) Section 5.9(b) of the Company Disclosure Schedule contains a true, correct and complete list of all Environmental Permits pertaining to the Company and its business. The Company currently has all Environmental Permits that are required for the operation of its business as presently operated, all of which are in full force and effect, except as would not reasonably be expected to adversely affect the Company in any material respect. The Company (i) is not in violation, in any material respect, of any terms or conditions of any Environmental Permit, or (ii) has not received any written notice of a pending violation of any Environmental Permit. No Action is pending or, to the Knowledge of the Company, threatened, to revoke, prevent the renewal of, rescind, modify, refuse to renew or limit any such Environmental Permit, nor has the Company received any written notice from any Governmental Entity with respect to same. The Company has timely filed or caused to be timely filed any and all applications for the renewal of any Environmental Permit, except as would not reasonably be expected to adversely affect the Company in any material respect.

5.10 Intellectual Property.

- (a) Section 5.10(a) of the Company Disclosure Schedule sets forth a true and complete list of all (i) patented and registered Company Intellectual Property Rights, (ii) pending patent applications and applications for registration of other Company Intellectual Property Rights filed by or on behalf of the Company, (iii) trade or corporate names used by the Company, (iv) material unregistered trademarks and service marks owned or used by the Company and (v) all domain names owned or used by the Company. None of the patents and patent applications disclosed or required to be disclosed on Section 5.10(a)(i) of the Company Disclosure Schedule (the "Patents") is necessary or useful for the performance in full of any of the business or operations of the Company as currently conducted.
- (b) To the Knowledge of Company, after due inquiry, (i) all necessary registration, maintenance and renewal fees for each item of registered Company Intellectual Property Rights have been paid and all necessary documents, recordations and certificates in connection with such Company Intellectual Property Rights have been filed with the relevant Governmental Entity for the purposes of maintaining or perfecting such Company Intellectual Property Rights; (ii) all registrations for registered Company Intellectual Property Rights are in good standing and subsisting; and (iii) except as set forth in Section 5.10(b) of the Company Disclosure Schedule, there are no actions that must be taken or payments that must be made by the Company within one hundred and eighty (180) days following the date hereof that, if not taken, will adversely affect any registered Company Intellectual Property Rights.
- (c) Except with respect to any Off-The-Shelf Software or as set forth on Section 5.10(c) of the Company Disclosure Schedule, the Company owns all right, title and interest to, or has the right to use, free and clear of all Liens (except Permitted Liens and restrictions contained in any applicable license), all Company Intellectual Property Rights and Licensed Intellectual Property Rights used by the Company in the conduct of the Company's business.

- (d) All of the agreements pursuant to which the Company has been granted rights by third parties with respect to Licensed Intellectual Property Rights used by the Company in the conduct of the Company's business (each, an "IP License") are in force and effect and are binding and enforceable against the Company. Neither the Company nor, to the Knowledge of the Company, any other party thereto is in material breach or material default of any IP License. To the Knowledge of the Company, no event has occurred that with notice or lapse of time would constitute a material breach or material default under any IP License by the Company. The Company has not received any written notice from any third party asserting a claim, or threatening to make a claim, which would materially and adversely affect the rights of the Company under any IP License.
- (e) The Company has not received any written notice from any third party challenging or threatening to challenge the right, title or interest of the Company in, to or under the Company Intellectual Property Rights or asserting any opposition, interference, invalidity, termination, abandonment, or other infirmity of any Company Intellectual Property Rights.
- (f) Except as set forth in Section 5.10(f) of the Company Disclosure Schedule, the execution and delivery of this Agreement and the consummation of the Transactions will not result in (i) the Company granting to any third party any rights or licenses to any Intellectual Property Rights, (ii) any IP License being subject to any non-compete or other restriction on its use or operation, to which it is not subject prior to the Closing, or (iii) the imposition of any Lien on Company Intellectual Property Rights. Except as set forth in Section 5.10(f) of the Company Disclosure Schedule, following the execution and delivery of this Agreement and the consummation of the Transactions, the Purchaser will be permitted to exercise all of the Company's rights under all IP Licenses to the same extent the Company would have been able to had the Transactions not occurred and without being required to pay any additional amounts or consideration other than fees, royalties or payments which the Company would otherwise be required to pay had the Transactions not occurred.
- (g) The Company has taken reasonable steps in accordance with industry standards to protect the Company's trade secrets, including by having officers, employees and consultants of the Company execute agreements providing for the protection of proprietary information of the Company. All officers, employees and consultants of the Company who are or have been involved in the creation or development of Company Intellectual Property Rights have executed and delivered to the Company an agreement providing for the assignment by such Persons to the Company of any Company Intellectual Property Rights made in the course of such Persons' employment or engagement by the Company. To the Knowledge of the Company, no officer, employee or consultant of the Company is in violation of any term of any such proprietary information and assignment agreement between such Person and the Company.

5.11 Real Estate.

- (a) The Company does not own any real property and has never owned any real property, nor is the Company party to any agreement to purchase or sell any real property. Section 5.11(a) of the Company Disclosure Schedule sets forth all real property that the Company leases, subleases, uses or occupies (the "Company Facilities") and a listing of the lease or sublease documents pertaining thereto. The Company has provided (or made available to) the Purchaser with true, correct and complete copies of all leases, lease guaranties, subleases, agreements for the leasing, use or occupancy of, or otherwise granting a right in or relating to the Company Facilities, including all amendments, terminations and modifications thereof (the "Company Leases").
- (b) Except as set forth on Section 5.11(b) of the Company Disclosure Schedule, with respect to each Company Lease, (i) the Company has a valid leasehold interest, free and clear of any and all Liens, other than Permitted Liens, (ii) there is not, under any Company Leases, any existing material breach or material

default (or event which with notice or lapse of time, or both, would constitute a material breach or material default) of the Company or, to the Knowledge of the Company, any other party thereto, (iii) the Company's possession and quiet enjoyment of the Leased Real Property under such Lease has not been disturbed, and to the Knowledge of the Company, there are no disputes with respect to such Lease and (iv) the Company has not subleased, licenses or otherwise granted any Person the right to use or occupy any real property subject to such Company Leases or any portion thereof; and (v) the Company has not collaterally assigned or granted any other security interest in such Company Lease or any interest therein.

5.12 <u>Litigation</u>. Except as set forth on <u>Section 5.12</u> of the Company Disclosure Schedule, there is, and for the past three (3) years has been, no Action pending or, to the Knowledge of the Company, threatened in writing, against (a) the Company, (b) the Company's officers, managers, employees or members, in their capacity as such, or (c) the assets, properties or business of the Company. The Company is not subject to any Order. There is no Action pending or threatened by or on behalf of the Company against any other Person. There is no Action pending or, to the Knowledge of the Company, threatened in writing against the Company which challenges the validity of this Agreement or the Transactions or otherwise seeks to delay, limit or enjoin the Transactions.

5.13 Employee and Labor Relations.

(a) Section 5.13(a) of the Disclosure Schedule sets forth a complete and accurate list of all current employees of the Company with the following information: (i) employee name and identification number, (ii) job title, (iii) hire date, (iv) hourly rate for non-exempt employees or base salary for exempt employees, (v) bonus eligibility, (vi) commission eligibility, (vii) Fair Labor Standards Act classification as exempt or non-exempt, (viii) part-time/full-time status, (ix) active/inactive status (and if inactive, start date of leave and expected return to work date), (x) visa status (including visa type and expiration date, if applicable), and (xi) assigned work location (city, state). Section 5.13(a) of the Disclosure Schedule sets forth a complete and accurate list of all single natural person independent contractors who currently provide services to the Company with the following information: name, brief description of services, term of engagement, compensation arrangement (hourly rate, fixed fee, etc.) and annual fees paid, and location of services (city, state). Except as provided in Section 5.13(a) of the Company Disclosure Schedule, all employees are employed at-will and are not entitled to severance pay or other compensation (except accrued wages and benefits) upon termination of employment. To the Knowledge of the Company, the Company does not employ or retain any employee or independent contractor who is a party to, or is otherwise bound by, any agreement that adversely affects or restricts the performance of the employee's or independent contractor's duties for the Company, including any confidentiality, non-solicitation or non-competition agreement between any such employee or independent contractor and a third party

(b) The Company has complied, and is currently in compliance, in all material respects with, all applicable Laws relating to employment of employees and the engagement of independent contractors/consultants, including, all such applicable Laws relating to labor relations, equal opportunity, fair employment practices, prohibited discrimination and harassment, wage and hour, immigration, employee safety and health, workers' compensation, leaves of absence, and paid sick leave. There has been no charge of discrimination filed against the Company with the Equal Employment Opportunity Commission or similar Governmental Entity during the three (3) years prior to the date of this Agreement. In the three (3) years prior to the date of this Agreement, there have been no Actions against the Company nor are any such Actions pending, or to the Knowledge of the Company, threatened to be brought or filed, by or with any Governmental Entity in connection with the application for employment, employment or termination of employment of any current or former employee of the Company, the retention or use of any independent contractor or leased employee, or any other labor or employment practices or laws. In accordance with applicable Laws, each person performing services for the Company is and has been for the past three (3) years (i) properly classified as an

employee or independent contractor, (ii) if an employee, properly classified as an exempt or non-exempt employee under the Fair Labor Standards Act or other applicable wage and hour Law, and (iii) paid all wages (including any required minimum wage, overtime pay, meal and rest breaks and waiting time penalties), benefits, and other compensation for all services performed by such person.

- (c) There has not been pending or existing during the three (3) year period preceding the date of this Agreement any strike, picketing, slowdown, work stoppage, lockout or other labor dispute involving the Company. There is no unfair labor practice charge or complaint against the Company pending before the National Labor Relations Board or similar Governmental Entity, and to the Knowledge of the Company, no such charge or complaint has been made against the Company during the three (3) years prior to the date of this Agreement. No application or petition for an election of or certification of a collective bargaining agent relating to the Company is pending as of the date of this Agreement, and there has been no union organizing activity in the past three (3) years. The Company is not and has never been a party to any collective bargaining agreement.
- (d) The Company maintains I-9 Forms for all employees, which have been completed as required by applicable Law and retained in accordance with applicable Law. All United States-based Company employees are legally authorized to work in the United States.
- (e) Except as set forth in Section 5.13(e) of the Company Disclosure Schedule, no layoff, reduction in force, furlough, reduction in hours or compensation, or temporary layoff affecting employees of the Company has occurred within the past 12 months, as a result of COVID-19 or any Laws or Orders by any Governmental Entity in connection with or in response to COVID-19.
- (f) In the 12 months prior to the date of this Agreement, the Company has not engaged in any plant closing or employee layoffs or reduction in employee hours that would violate or give rise to an obligation to provide any notice required under the federal Worker Adjustment Retraining and Notification Act or any similar state, local or foreign Laws (collectively, "WARN"), and it has no plans to undertake any actions that would require the providing of any such notice.

5.14 Employee Plans.

(a) Section 5.14(a) of the Company Disclosure Schedule sets forth each material Employee Plan (other than (i) offer letters or other similar employment Contracts with employees that are terminable at-will by the Company or any ERISA Affiliate without severance or change of control pay or benefits, and do not contain any non-recurring or special bonuses or payments, in which case only the form of such offer letters or other service Contracts will be listed, or (ii) consulting agreements or other similar service Contracts with consultants that are terminable without penalty on less than thirty (30) days' notice, in which case only forms of such consulting agreements or other service Contracts will be listed, unless any such Contract provides severance, change of control pay or benefits, or a non-recurring or special bonus or payment). With respect to each material Employee Plan, to the extent applicable, the Company has made available to the Purchaser (i) the plan document, summary plan description and any summary of material modifications or amendments; (ii) the most recent annual report on Form 5500 required to have been filed with the IRS and non-discrimination tests for the last three plan years; (iii) the most recent determination letter, if any, from the IRS for any Employee Plan that is intended to qualify pursuant to Section 401(a) of the Code; (iv) any related trust agreements, insurance contract or other document related to funding or payment of benefits under such Employee Plan; and (v) any notices to or from the IRS or any office or representative of the United States Department of Labor or any governmental agency relating to any material compliance issues in respect of any such Employee Plan.

- (b) The Company and each of its ERISA Affiliates in all material respects, have performed all obligations required to be performed by them under, and are in compliance with, the requirements prescribed by any and all applicable statutory or regulatory Laws, are not in material default or violation of, and the Company has no Knowledge of any such default or violation by any other party to, any Employee Plan.
- (c) For each Employee Plan that is intended to be qualified under Section 401(a) of the Code, the Company has obtained a favorable determination and/or opinion letter and there has been no event, condition or circumstances that has adversely affected or is reasonably likely to adversely affect such qualified status. No "prohibited transaction," within the meaning of Section 4975 of the Code or Sections 406 and 407 of ERISA, and not otherwise exempt under Section 408 of ERISA, has occurred with respect to any Employee Plan. There are no Actions pending or to the Knowledge of the Company threatened or reasonably anticipated (other than routine claims for benefits) against any Employee Plan or against the assets of any Employee Plan. None of the Company nor any of its ERISA Affiliates is subject to any penalty or Tax with respect to any Employee Plan under Section 502(i) of ERISA or Sections 4975 through 4980 of the Code.
- (d) Neither the Company nor any of their ERISA Affiliates maintains, contributes to, participates in, or sponsors (or has in the past six (6) years maintained, contributed to, participated in, or sponsored), or has any Liability, contingent or otherwise under any (i) defined benefit pension plan or plan subject to Section 302 of Title I of ERISA, Section 412 of the Code, or Title IV of ERISA, (ii) multiemployer plan (as defined in Section 3(37) of ERISA), (iii) multiple employer plan or to any plan described in Section 413 of the Code, (iv) "funded welfare plan" within the meaning of Section 419 of the Code, or (v) multiple employer welfare arrangement, as defined under Section 3(40)(A) of ERISA (without regard to Section 514(b)(6)(B) of ERISA).
- (e) Neither the Company nor any of their ERISA Affiliates has any Liability in respect of post-retirement health, medical or life insurance benefits for retired, former or current employees or other service providers of the Company or any of its ERISA Affiliates, except as required to avoid excise tax under Section 4980B of the Code or except for the continuation of coverage through the end of the calendar month in which termination from employment occurs.
- (f) Neither the execution and delivery of this Agreement nor the consummation of the Transactions (either alone or together with any other event) or any termination of employment or service in connection therewith will (i) entitle any employee, director, or other service provider of the Company (whether current, former or retired) or their beneficiaries to severance pay under an Employee Plan (ii) accelerate the time of payment or vesting or trigger any payment or funding of compensation or benefits under any Employee Plan, or (iii) increase the amount payable or trigger any other financial obligation pursuant to an Employee Plan. No payment which is or may be made to any employee, director or other service provider of the Company or any ERISA Affiliate (whether current, former or retired), either alone or together with any other payment, event or occurrence, will or could fail to be deductible for federal income tax purposes by virtue of Section 280G of the Code or subject to an excise tax under Section 4999 of the Code. The Company has no obligation to gross up, indemnify or otherwise reimburse any current or former employee, director or other service provider of the Company or any ERISA Affiliate for any Tax incurred by such individual under Section 4999 of the Code.
- (g) Each Employee Plan that is a "nonqualified deferred compensation plan" (within the meaning of Section 409A of the Code) complies and has at all times complied in both form and operation, with the requirements of Section 409A of the Code. The Company has no obligation to gross up, indemnify or otherwise reimburse any current or former employee, director, or other service provider of the Company for any Tax incurred by such individual under Section 409A of the Code.

- (h) Neither the Company nor any of its ERISA Affiliates maintains, participates in, contributes to, or sponsors (or has in the past six years maintained, contributed to, participated in, or sponsored) any International Employee Plan.
- (i) Neither the Company nor any ERISA Affiliate has failed to comply with Sections 601 to 608 of ERISA and Section 4980B of the Code, and the Company and each ERISA Affiliate has, for any relevant period, offered the requisite number of "full-time employees" group health coverage that is "affordable" and of "minimum value" (as such terms are defined by the employer shared responsibility provisions of the Patient Protection and Affordable Care Act).
- 5.15 Affiliate Transactions. Except as set forth on Section 5.15 of the Company Disclosure Schedule, no Related Party is a party to any Related Party Transaction with the Company or has any material interest in any material assets or property used by the Company. To the Knowledge of the Company, no Related Party has any direct or indirect ownership, participation, royalty or other interest in, or is an officer, manager, director, employee of or consultant or contractor for any firm, partnership, entity or corporation that competes with, or does business with, or has any contractual arrangement with, the Company; provided, however, that ownership of no more than three percent (3%) of the outstanding voting stock of a publicly traded company shall not be deemed to be an "interest in" any entity described in this Section 5.15.
- 5.16 Insurance. Section 5.16 of the Company Disclosure Schedule sets forth a list of each insurance policy currently maintained by the Company with respect to its properties, assets and business (the "Insurance Policies"), along with the policy type, insurer, policy number, effective date and expiration date of each such Insurance Policy. The Company has delivered to the Purchaser a true, correct and complete copy of each Insurance Policy. The Insurance Policies have been issued by licensed insurers and are in full force and effect. All premiums due and payable under the Insurance Policies have been paid and there are no material defaults under any Insurance Policy by the Company. There is no material claim by the Company pending under any Insurance Policies as to which coverage has been denied or disputed by the insurer. Such Insurance Policies are in full force and effect, all premiums with respect thereto are currently paid, and the Company is in compliance in all material respects with the terms thereof. The Insurance Policies under which the Company is the primary named insured provide coverages as required by the applicable Governmental Entity, Law and any Contract to which the Company is party or by which any of the Company's assets or properties is bound. Immediately following the Closing, each Insurance Policy shall be available to the Company on substantially the same terms and conditions as immediately prior to the Closing.

5.17 Material Contracts.

- (a) <u>Section 5.17(a)</u> of the Company Disclosure Schedule sets forth as of the date of this Agreement each of the following Contracts to which the Company is a party or by which it is bound or by which the Company's assets are subject (other than Employee Plans set forth on <u>Schedule 5.14(a)</u> and Company Leases set forth on <u>Schedule 5.11(a)</u>) (each Contract set forth or required to be set forth on <u>Section 5.17(a)</u> of the Company Disclosure Schedule, a "<u>Material Contract</u>" and collectively, the "<u>Material Contracts</u>"):
- (i) Contracts involving payment obligations of, or to, the Company in excess of \$50,000 per annum or that have terms in excess of one (1) year that cannot be terminated by the Company on less than ninety (90) days' notice and without penalty;
 - (ii) Contracts relating to Indebtedness of the Company;
- (iii) (A) any Contract limiting the freedom of the Company to engage or participate, or compete with any other Person, in any material line of business, market or geographic area, (B)

any Contract under which the Company grants most favored nation pricing, volume requirements or commitments, exclusive sales, exclusive distribution, exclusive marketing or other exclusive rights of refusal, exclusive rights of first negotiation or similar exclusive rights and/or terms to any Person, or (C) any Contract otherwise limiting the right of the Company to sell, distribute or manufacture any products or services or to purchase or otherwise obtain any software, components, parts, subassemblies or services;

- (iv) any Contract under which the Company leases any personal property which involve payment obligations of, individually or in the aggregate, at least \$50,000 per annum;
- (v) any power of attorney or agency Contract with any Person pursuant to which such Person is granted the authority to act for or on behalf of the Company, or the Company is granted the authority to act for or on behalf of any Person;
- (vi) any Contract relating to any disposition of any material assets or business of the Company not in the ordinary course of business (whether by merger, sale of equity, sale of assets or otherwise);
- (vii) any Contract relating to the acquisition of the business or equity of another Person (whether by merger, sale of equity, sale of assets or otherwise);
- (viii) any Contract with any investment banker, broker, advisor or similar party retained by the Company in connection with this Agreement and the Transactions;
- (ix) any collective bargaining agreement or other Contract with any labor organization, labor union or employees representative body;
- (x) any Contract with any officer, employee, or consultant, independent contractor, or director that (A) provides total annual payments in excess of \$50,000, whether pursuant to any Employee Plan or otherwise or (B) cannot be terminated with less than ninety (90) days' notice or further liability to the Company without the payment of severance;
 - (xi) any Contract with a Major Customer or a Major Supplier;
 - (xii) any Contract with a Governmental Entity (each, a "Government Contract") or Government Bid;
- (xiii) any settlement, conciliation or similar Contract (i) with any Governmental Entity, (ii) entered into in the past three (3) years pursuant to which the Company has or will have material outstanding obligations after the date hereof or (iii) that governs or restricts the use of any Intellectual Property;
 - (xiv) any Contract for capital expenditures or the acquisition or construction of fixed assets in excess of \$50,000;
 - (xv) any Related Party Transaction; and
 - (xvi) any partnership or joint venture agreement.

With respect to each of the Material Contracts: (i) such Material Contract is legal, valid, binding, enforceable, and in full force and effect with respect to the Company, as applicable, except as such

enforceability may be limited by (A) applicable insolvency, bankruptcy, reorganization, moratorium, or other similar Laws affecting creditors' rights generally and (B) applicable equitable principles (whether considered in a proceeding at Law or in equity), (ii) the Company is not in material breach or default under any Material Contract, and (iii) to the Knowledge of the Company, no other party to any Material Contract is in material breach or default thereof. The Company party will not, and the Company does not intend to, terminate or materially modify any Material Contract and, to the Knowledge of the Company, no other party to such Material Contract has provided written notice that it will, or intends to, terminate any Material Contract or make any material modification to any Material Contract. The Company has provided or made available to the Purchaser true, correct, and complete copies of all Material Contracts.

(b) Immediately following the Closing, each Material Contract shall be available to the Company on substantially the same terms and conditions as immediately prior to the Closing.

5.18 FDA and Regulatory Matters.

- (a) The Company is, and for the last three (3) years has been, in compliance, in all material respects, with all Healthcare Laws applicable to the Company. The Company's business as currently conducted, including contract manufacturing and research services, is being, and for the last three (3) years has been, conducted in compliance, in all material respects, with all applicable Healthcare Laws. The Company has not, during the last three (3) years, received any written communication or written notification of any pending or threatened Action from any Governmental Entity or any qui-tam relator, including, without limitation, the FDA, the Centers for Medicare & Medicaid Services, and the U.S. Department of Health and Human Services Office of Inspector General, or any comparable state, federal or foreign Governmental Entity, alleging non-compliance by, or liability of, the Company under any Healthcare Law.
- (b) For the past three (3) years, all product development and manufacturing services of the Company, including all design, manufacturing, processing, assembly, packaging, labeling, holding, storage, and, distribution activities, (i) have been performed in compliance in all material respects with all applicable Healthcare Laws, except as disclosed on Section 5.18(b) of the Company Disclosure Schedule, in conformance with the applicable specifications for the manufacture, storage, and handling of the applicable product in effect at the time of delivery thereof; (ii) have complied in all material respects with cGMPs and, at the time of delivery thereof, were not adulterated or misbranded within the meaning of the FDCA and (iii) neither the FDA nor any Governmental Entity is currently alleging non-compliance with cGMPs and no pending clinical trial is subject to termination or suspension relating to any product developed or manufactured by or on behalf of the Company or with respect to which the Company provides development or manufacturing services, except, in each case, as has not been and would not reasonably be expected to be individually or in the aggregate, material to the business of the Company, taken as a whole.
- (c) To the Knowledge of the Company, (i) in the past three (3) years, no product developed, tested, processed, manufactured, packaged, labeled, or sterilized by the Company has been seized, withdrawn, recalled, detained, subject to a public health notification or safety alert or any clinical hold, or has been subject to a suspension of manufacturing, distribution, or commercialization activity as a result of any action taken or failure to take action by the Company and (ii) each third party that is a supplier or contractor for the Company is in material compliance with all applicable Healthcare Laws and holds all applicable Permits except, in each case, as has not been and would not reasonably be expected to be individually or in the aggregate, material to the business of the Company, taken as a whole.
- (d) The Company holds such Permits of Governmental Entities from the United States or foreign governments or government agencies necessary and required for the conduct of its business as currently conducted, including, where applicable and without limitation, all authorizations under the Federal Food, Drug

and Cosmetic Act of 1938, as amended (the "FDCA"), the Public Health Service Act of 1944 and the regulations of the FDA promulgated under any of the foregoing or any Law or authorization of any other Governmental Entity, and any other Governmental Entity that regulates the quality, identity, strength, purity, safety, efficacy, developing or manufacturing of pharmaceutical products and all other Permits necessary and required to permit the contract manufacturing and research services provided by the Company in jurisdictions where it currently conducts such activities with respect to each of the Products (collectively, the "Company Licenses"). The Company has fulfilled and performed, in all material respects, all of its required obligations with respect to each Company License and is in compliance, in all material respects, with all terms and conditions of each Company License, and, to the Knowledge of the Company, no event has occurred which allows, or after notice or lapse of time would allow, revocation or termination thereof.

- (e) For the last three (3) years, no manufacturing site (whether owned or operated by the Company) has been subject to a Governmental Entity (including FDA) shutdown or import or export prohibition.
- (f) For the last three (3) years, there have been no material adverse regulatory actions taken (or, to the Knowledge of the Company, threatened) by the FDA or any other Governmental Entity with respect to any facilities where pharmaceutical products are tested, produced, processed, packaged or stored, including, without limitation, warning letters, untitled letters, FDA Form 483 Notices of Inspectional Observations, or similar material written notices or written communications of enforcement action issued by the FDA or comparable written communications from any other Governmental Entity, except, in each case, as has not been and would not reasonably be expected to be individually or in the aggregate, material to the business of the Company, taken as a whole.
- (g) The Company is not the subject of any pending or, to the Knowledge of the Company, threatened investigation regarding any of the Company's contract manufacturing or research services by the FDA pursuant to its "Fraud, Untrue Statements of Material Facts, Bribery, and Illegal Gratuities" Final Policy set forth in 56 Fed. Reg. 46,191 (September 10, 1991) and any amendments thereto ("FDA Fraud Policy"), or similar and applicable policy enforced by any other Governmental Entity. The Company has not, nor has, to the Knowledge of the Company, any officer, employee, or agent of the Company, during the past three (3) years, made a materially untrue statement to the FDA or any other Governmental Entity, failed to disclose a material fact required to be disclosed to the FDA or any other Governmental Entity, or committed an act, made a statement or failed to make a statement of material fact that, at the time such disclosure was made, would reasonably be expected to provide a basis for the FDA or any other Governmental Entity to invoke the FDA Fraud Policy or any similar policy. The Company has not, and, to the Knowledge of the Company, none of its officers, employees, or agents, been or is currently subject to any debarment, suspension or exclusion under any government healthcare program or under the FDCA. The Company has not, nor, to the Knowledge of the Company, has any officer, employee, or agent of the Company been convicted of any crime or engaged in any conduct for which such person could be excluded from participating in the federal health care programs under Section 1128 of the Social Security Act of 1935, as amended, or any similar Law. No Actions that would reasonably be expected to result in a material debarment or exclusion are pending or, to the Knowledge of the Company, threatened against the Company or, to the Knowledge of the Company, any of its directors, officers, employees or agents.
- 5.19 <u>Accounts Receivable</u>. All of the accounts receivable reflected on the Latest Balance Sheet, whether billed or unbilled, of the Company arose in the ordinary course of business consistent with past practice from sales actually made or services actually performed and are carried at values determined in accordance with GAAP. Unless paid prior to the Closing, as of the Closing Date, all accounts receivable will be current and collectible net of the respective reserves shown on the Latest Balance Sheet (which reserves are calculated consistent with past practice and (b) will not represent a change in the composition of such accounts receivable

in terms of aging). Subject to such reserves or, with respect to accounts receivable arising after the Latest Balance Sheet Date, reserves shown on the accounting records of the Company for the applicable period, each account receivable is collectible in full or sellable in connection with the factoring of accounts receivable consistent with past practices, without any set-off. There is no contest, written claim or right of set-off, under any Contract with any obligor of any accounts receivable related to the amount or validity of such accounts receivable, and no bankruptcy, insolvency or similar Actions have been commenced by or against any such obligor.

5.20 Customers and Suppliers.

- (a) Section 5.20(a) of the Company Disclosure Schedule contains a list setting forth the twenty (20) largest customers of the Company, by dollar amount, over the twelve (12) months ended June 30, 2021 (collectively, the "Major Customers") (and the amount of sales with respect to each such Major Customer during such twelve (12) month period), and the twenty (20) largest suppliers of the Company, by dollar amount, over the twelve (12) months ended June 30, 2021 (collectively, the "Major Suppliers") (and the amount of purchases from each such Major Supplier during such period).
- (b) Within the last three (3) years, the Company has not received written notice that any Major Customer has taken action to, or will take action to (i) terminate or modify in a manner materially adverse to the Company its relationship with the Company, (ii) materially reduce the amount of products or services that it is willing to purchase from the Company, or (iii) materially reduce the price of products or services that it has previously purchased from the Company.
- (c) Within the last three (3) years, the Company has not received written notice that any Major Supplier has taken action to, or will take action to (i) terminate or modify in a manner materially adverse to the Company its relationship with the Company, (ii) materially reduce the amount of goods that it is willing to supply to the Company, or (iii) materially increase the price of goods that it has previously supplied to the Company.
- (d) The Company is not involved in, and, for the past three (3) years, has not been involved in, any Action or any material controversy or dispute with any Major Customer or Major Supplier and, to the Knowledge of the Company, there exists no facts or circumstances that would result in any such material Action or controversy or dispute being initiated.
- 5.21 <u>Broker Fees</u>. Except as set forth on <u>Section 5.21</u> of the Company Disclosure Schedule, the Company has not used any broker or finder, or entered into any Contract with any Person which would result in the obligation of the Company or the Purchaser to pay any fees or commissions to any broker, finder, or agent, with respect to the Transactions.
- 5.22 <u>Bank Accounts. Section 5.22</u> of the Company Disclosure Schedule sets forth (a) a true and complete list of the names and locations of all banks, trust companies, securities brokers and other financial institutions at which the Company has a deposit, lock box or cash collection, management, securities or other account or a safe deposit box or maintains a banking, custodial, trading or other similar relationship, and (b) a true and complete list of each such account, box and relationship, indicating in each case the account number and the names of the respective officers, employees, agents or other similar representatives of the Company having signatory power with respect thereto.
- 5.23 <u>Corrupt Practices Act; Ex-Im Laws</u>. In the last three (3) years, no action has been taken nor any payment or other benefit has been made or conferred by the Company or any Seller or, to the Company's Knowledge, by any Representative of the Company or other Person acting on behalf of the Company or any

Seller in violation of applicable Law (including any applicable Anti-Corruption Laws). In the last three (3) years, none of the Company or any Seller or any of their respective Related Parties, or, to the Company's Knowledge, any of their respective Representatives (a) has made, offered or authorized, directly or indirectly, any contribution, bribe, rebate, payoff, influence payment, kickback, other payment or gift of anything of value, regardless of what form, whether in money, property or services (each, a "Payment"), directly or indirectly, to a Government Official, in order to or that would be reasonable likely to (i) improperly influence any act or decision of such Government Official in his/her or its official capacity; (ii) improperly induce such Government Official to do or omit to do any act in violation of the lawful duty of such Government Official; (iii) improperly induce such Government Official to use his/her or its influence to affect an official act or decision; (iv) secure any improper business advantage; or (v) otherwise be in violation of any applicable Laws (including any applicable Anti-Corruption Laws) or (b) has used any Company funds for unlawful contributions, gifts, entertainment or other unlawful expenses related to any political activity. In the last three (3) years, no Payment has been made or conferred by the Company nor, to the Company's Knowledge, by any Person on behalf of the Company or any of its Related Parties in connection with the Company's business in violation of applicable Laws (including applicable Anti-Corruption Laws). No employee or consultant of the Company is a Government Official or, to the Company's Knowledge, a member of the Family Group of a Government Official. None of the Company, or, to the Company's Knowledge, any Representative or other Person acting for or on their behalf, is currently, or has been in the last five (5) years, (i) a Sanctioned Person; (ii) organized, resident or located in a Sanctioned Country; (iii) engaging in any dealings or transactions with any Sanctioned Person or in any Sanctioned Country, to the extent such activities violate applicable Sanctions Laws or Ex-Im Laws; or (iv) otherwise in violation of applicable Sanctions Laws, Ex-Im Laws, or U.S. anti-boycott Laws (collectively, the "Trade Control Laws").

5.24 Government Contracts and Regulatory Matters.

- (a) The Company has been compliant in all material respects with all pre-award requirements as described in each solicitation, synopsis, tender, or other form of government statement of requirements, and all rules and regulations applicable thereto at the time of award of each Government Contract. The Company has not received written notice that any Government Contract is the subject of bid or award protest proceedings. The Company has made available to the Purchaser a complete and correct copy of each Government Contract and modification thereof that is a Material Contract, including each purchase order, task order or delivery order issued under such Government Contract or under any blanket purchase agreement, federal or state supply schedule contract or similar purchasing arrangement.
- (b) Except as set forth on Section 5.24(b) of the Company Disclosure Schedule, there exists no Government Contract or Government Bid (A) in connection with which the Company explicitly represented to the applicable Governmental Entity that the Company qualified as a "Small Business Concern," a "Small Disadvantaged Business," an "8(a) concern," a "Service-Disabled Veteran-Owned Small Business Concern," a "Veteran-Owned Small Business Concern, a Historically Underutilized Business Zone Small Business Concern, a Woman-Owned Small Business Concern," a ""protégé" under a mentor-protégé agreement or program," or any other preferential small business status established under the U.S. Small Business Administration's set-aside program rules (collectively, "Set-Aside Program Status"); or, (B) in the case of a Government Contract, that the Company would not have been eligible or invited to submit a bid or receive a contract award but for its Set-Aside Program Status.
- (c) Except as set forth on Section 5.24(c)(i) of the Company Disclosure Schedule, to the Company's Knowledge, the Company has never submitted a Government Bid or been awarded a Government Contract which the Company was ineligible to be awarded due to its business classification or any other reason at the time such Government Bid was submitted in connection with a procurement reserved or set-aside for companies having Set-Aside Program Status or any other specified business classification or eligibility

criterion. Except as set forth on Section 5.24(c)(ii) of the Company Disclosure Schedule, to the Company's Knowledge, no Government Contract is required to be terminated by a Governmental Entity and no such Governmental Entity is prohibited from exercising any option period under such Government Contract as a result of the consummation of the transactions contemplated by this Agreement.

- (d) With respect to each Government Contract and Government Bid:
- (i) The Company has complied, in all material respects, with all terms and conditions of each Government Contract, including all clauses, provisions and requirements incorporated expressly by reference and including any requirements relating to the charging of prices or costs, most favored customer pricing, minimum qualifications of personnel, warranties, industrial funding fees and price reductions. To the Company's Knowledge, no event has occurred in connection with a Government Contract or Government Bid which, with the passage of time or the giving of notice or both, would reasonably be expected to result in a condition of default or material breach of a Government Contract or material non-compliance with requirements applicable thereto or to any Government Bid.
- (ii) All representations and certifications made, acknowledged or set forth in or pertaining to each Government Contract or Government Bid were current, accurate and complete, in all material respects, as of their effective date, and all such representations and certifications have continued to be current, accurate and complete, in all material respects, to the extent required by the terms of a Government Contract. During the past six (6) years, the Company has not submitted a Government Bid or been awarded a Government Contract based upon material misrepresentations or inaccuracies in representations and certifications executed in connection therewith or contained in the Company's annual registration on the Government's System for Award Management database.
- (iii) All invoices for payment, reimbursement or adjustment, including requests for progress payments and provisional payments, submitted during the past six (6) years by or on behalf of the Company in connection with a Government Contract were current, accurate and complete, in all material respects, as of their applicable submission dates.
- (iv) Except as provided in Section 5.24(d)(iv) of the Company Disclosure Schedule, the Company has not filed any material claims or other non-routine requests for payment, but all such claims and other non-routine requests for payment listed therein, which includes requests for additional compensation stemming from disputes, changes and controversies, which were submitted during the past six (6) years by or on behalf of the Company in connection with a Government Contract were current, accurate and complete, in all material respects, as of their applicable submission dates. Except as set forth on Schedule 5.24(d)(iv), (A) within the past three (3) years, no Government Contract has been terminated for convenience or default and (B) the Company has not received any written cure notice or show cause notice regarding performance of a Government Contract or any written notice of, Claim for, or assertion of, a condition of default, breach of contract, or material violation of Applicable Law, in connection with a Government Contract or Government Bid.
- (v) To the Company's Knowledge, all "certified cost and pricing data" and all "other than certified cost and pricing data" submitted during the past six (6) years by or on behalf of the Company in connection with a Government Contract or Government Bid were current, accurate and complete as of the certification date. For avoidance of doubt, certified cost and pricing data and other than certified cost and pricing data, respectively, have the meaning ascribed at FAR Subpart 15.4.
- (vi) The Company has, to the extent required by Applicable Laws and the terms of its Government Contracts, maintained systems of required internal controls, including quality control

systems, cost accounting systems, estimating systems, purchasing systems, proposal systems, billing systems and material management systems, that are in compliance, in all material respects, with all requirements of the Government Contracts. The Company has not received written or, to the Company's Knowledge, oral notice that any of its subcontractors, teaming partners, consultants, agents or representatives has violated any Applicable Law in connection with any Government Contract or any Government Bid for which the Company would reasonably be expected to incur any material liability. The Company has not received any adverse or negative past performance evaluations or ratings pertaining to any Government Contract within the past three (3) years.

(vii) Except as set forth on Section 5.24(d)(vii) of the Company Disclosure Schedule, in the last six (6) years, there has not been any material withholding, recoupment or setoff of or against any payments due Company by a Governmental Entity or prime contractor or higher-tier subcontractor nor, to the Company's Knowledge, has there been any attempt to materially withhold, recoup or setoff, any payments due under any Government Contract on any basis, including the basis that a cost incurred or invoice rendered by the Company was questioned or disallowed by a Governmental Entity, prime contractor or higher-tier subcontractor or any of their audit representatives, nor to the Company's Knowledge would any such withholding or setoff reasonably be expected to occur.

(viii) Except as set forth on Section 5.24(d)(viii) of the Company Disclosure Schedule, no Government Contract has, to date, or, to the Company's Knowledge, is currently projected to have, fully burdened costs incurred in excess of the Government Contract fixed price, or, in the case of flexibly-priced or cost-reimbursement Government Contracts, would reasonably be expected to have fully burdened costs incurred in excess of the ceiling price or funded amount of the Government Contract.

(e) Except as set forth on the Section 5.24(e) of the Company Disclosure Schedule, at all times over the past six (6) years, with respect to any Government Contract or Government Bid: (A) There is no pending material Claim or, to the Company's Knowledge, any reasonable basis to give rise to any material Claim against the Company for fraud or under any applicable state laws, rules, regulations, contract clauses or bidding terms and conditions, or any applicable federal laws, including the United States civil or criminal False Claims Acts, the United States Procurement Integrity Act, or other Applicable Law; (B) the Company has not been served with any document requests, subpoenas, search warrants or civil investigative demands addressed to or requesting information involving the Company, or any of its officers, employees, Affiliates, agents or representatives in connection with or related to any Government Contract or Government Bid; (C) the Company has not received written or, to the Company's Knowledge, oral notice that it, or any of its predecessors, managers, officers, directors, employees, Affiliates, agents or representatives, has been under administrative, civil or criminal investigation, indictment or criminal information, or audit by a Governmental Entity (other than routine audits by any government audit agency in the ordinary course of business) with respect to any Government Contract, Government Bid or Applicable Law, including any audit relating to a suspected, alleged or possible violation of applicable state laws or the United States civil or criminal False Claims Acts or the United States Procurement Integrity Act, provision of defective or non-compliant products or services, mischarging of prices or costs, misstatements of fact, or other acts, omissions or irregularities relating to any Government Contract or Government Bid; and (D) neither the Company nor, to the Company's Knowledge, any other Person, has conducted any internal investigation, audit, review or inquiry (whether or not any outside legal counsel, auditor, accountant or investigator was engaged) with respect to any suspected, alleged or possible material violation of any Government Contract, Government Bid or Applicable Law.

(f) The Company has not made, and to the Company's Knowledge is unaware of any circumstances under which it is or has been obligated or required to make a mandatory administrative, civil or criminal disclosure to a Governmental Entity. Neither the Company nor, to the Company's Knowledge, any predecessor of the Company has, in the past six (6) years, made a voluntary disclosure to any Governmental

Entity with respect to any alleged suspected, alleged or possible breach, violation, irregularity, mischarging, misstatement or other improper act or omission arising under or relating to any Government Contract or Government Bid. The practices and procedures used by the Company in estimating costs and pricing proposals and accumulating, recording, segregating, reporting and invoicing costs in connection with a Government Contract or Government Bid are, in all material respects, in compliance with Applicable Laws, including FAR Part 31 and all applicable Cost Accounting Standards and related regulations, to the extent such requirements are applicable. No audit by a Governmental Entity (including the Government Audit Agency) in the past six (6) years has questioned such costs or identified any other failure to comply, in all material respects, with contractual requirements or Applicable Law. Section 5.24(f) of the Company Disclosure Schedule lists each draft or final written audit report received by the Company during the past three (3) years issued by any Governmental Entity with respect to any Government Contract, Government Bid or any direct or indirect cost or other accounting practice of the Company. The Company has delivered or otherwise made available to the Purchaser correct and complete copies of each such report. The Company has not been determined by a Governmental Entity to be non-responsible, not presently responsible, or otherwise ineligible for award of a Government Contract within the past three (3) years.

- (g) With respect to Government Contracts and Government Bids: (i) the Company does not have any outstanding requests for a material equitable adjustment or materials Claims asserted by or against a Governmental Entity or any prime contractor, subcontractor, vendor or other third party arising under or relating to a Government Contract or Government Bid; (ii) the Company has not received written notice of any outstanding disputes between the Company, and any Governmental Entity under the Contract Disputes Act or any other Applicable Law governing disputes arising under such Government Contracts; (iii) to the Company's Knowledge, there are no outstanding disputes between any prime contractor for which the Company serves as a subcontractor under the Contract Disputes Act or any other Applicable Law governing disputes arising under such Government Contracts; (iv) the Company has not received written or, to the Company's Knowledge, oral notice of any disputes between the Company, on the one hand, and any prime contractor, subcontractor or vendor, on the other hand, arising under or relating to any such Government Contract or Government Bid; and (v) there are no financing arrangements or assignments of proceeds with respect to any Government Contract other than as provided in the Company's commercial bank or financing documents set forth on Section 5.24 of the Company Disclosure Schedule.
- (h) <u>Section 5.24(h)</u> of the Company Disclosure Schedule identifies all personal property, equipment and fixtures loaned, bailed or otherwise furnished to the Company, and in the Company's possession as of the date hereof, by or on behalf of a Governmental Entity (the "<u>Government Furnished Items</u>"), the current locations thereof and the Government Contract pursuant to which such Government Furnished Items were issued. The Company has complied, in all material respects, with all of its obligations relating to the Government Furnished Items and upon the return thereof to any Governmental Entity in the condition thereof on the date hereof, the Company would reasonably be expected to have no liability with respect thereto.
- (i) Except as stated in Section 5.24(i) of the Company Disclosure Schedule, the Company does not currently hold a facility clearance granted by the United States and no such clearance is required of it or any of its subcontractors, vendors, or consultants, and no personal security clearances are required of any officers, directors or employees of such entities for performance of any of obligations under a Government Contract or to submit a Government Bid.
- 5.25 <u>Disclaimer of the Company</u>. Except as otherwise specifically provided in <u>Article IV</u> or <u>Article V</u> (as modified by the Company Disclosure Schedule) or except as set forth in the other Transaction Documents or any certificate or other instrument delivered to the Purchaser hereunder or thereunder, (a) neither any Seller, the Sellers' Representative, the Company nor any of its managers, officers, employees, members,

agents, Affiliates or Representatives thereof, nor any other Person, has made or shall be deemed to have made any representation or warranty to the Purchaser or any of its Affiliates, express or implied, at Law or in equity, with respect to the Sellers, the Sellers' Representative, the Company, the Company's business or the assets, Liabilities, results of operations or financial condition of the Company, including any representations and warranties as to the accuracy or completeness of any Evaluation Material or any other information provided to the Purchaser or any of their respective Affiliates or Representatives pursuant to the Confidentiality Agreement or as to the future sales, revenue, profitability or success of the Company's business, or any representations or warranties arising from statute or otherwise in Law, from a course of dealing or a usage of trade and (b) all such other representations and warranties are expressly disclaimed by the Sellers and the Company. Nothing herein shall limit any rights or remedies of the Purchaser in respect of any claim for fraud.

ARTICLE VI

COVENANTS AND OTHER AGREEMENTS

6.1 <u>Public Announcements</u>. Following the Closing, the Purchaser and the Sellers' Representative shall jointly agree on the content and substance of all public announcements concerning the Transactions. Notwithstanding the foregoing, the Parties may make disclosures required by any applicable Law or Order or the rules of any applicable securities exchange, in which case the Party required to make such disclosure shall allow the other Party reasonable time to comment on such release or announcement in advance of such disclosure. The Parties acknowledge that either or both Parties may be obligated to make filings (including, but not limited to, the filing of a copy of this Agreement or the Transaction Documents) with the SEC or other Governmental Entity. Each Party shall be entitled to make such required filings, provided that it requests confidential treatment of sensitive terms of this Agreement to the extent such confidential treatment is reasonably available to such Party.

6.2 Employee Matters.

- (a) Following the Closing, the Purchaser shall, and shall cause the Company to, comply with any notice or filing requirements under WARN occurring after the Closing Date. The Company shall be responsible for complying with any notice or filing requirements under WARN occurring on or before the Closing Date. The Company shall provide the Purchaser at Closing with a list of all employment terminations for the Company for the ninety (90) days prior to the Closing Date, and for each employment termination, the Company shall provide the employee's name, date of termination, and location of employment.
- (b) The Purchaser agrees that during the period commencing on the Closing Date and ending on the first anniversary thereof, each employee of the Company shall be paid an annual base salary or base wages at a rate no less favorable than the rate in effect for such employee immediately prior to the Closing.
- (c) The Purchaser agrees that, during the period commencing on the Closing Date and ending on the first anniversary thereof, employees of the Company who are receiving benefits under the Employee Plans immediately prior to the Closing ("Affected Employees") will thereafter continue to be provided with benefits under employee benefit plans that are no less favorable in the aggregate than either (i) those provided by the Company to such employees as of immediately prior to the Closing, or (ii) those provided by the Purchaser to similarly situated employees from time to time during such period, provided, however, that for the purposes of this Section 6.7(c), no equity-based, retention, change in control, non-qualified deferred compensation or other non-recurring or special compensation or benefits provided prior to the Closing shall be taken into account.

- (d) Following the Closing, the Purchaser shall, or shall cause its Affiliates, to grant each Affected Employee credit for any service with the Company (or predecessor employers, as applicable) (i) for purposes of eligibility to participate and vesting under any benefit or compensation plan, program, agreement or arrangement that may be established or that is maintained by the Purchaser or its Affiliates on or after the Closing Date for similarly situated employees (but not under any equity incentive plan) and (ii) solely with respect to vacation accrual and severance benefit determinations, benefit accrual; provided, however, that such service shall not be recognized or credited to the extent that such recognition would result in a duplication of benefits provided to the Affected Employee.
- (e) From and after the Closing, the Purchaser shall use commercially reasonable efforts to (i) cause to be waived all pre-existing condition exclusions and actively-at-work requirements and similar limitations, eligibility waiting periods and evidence of insurability requirements under the welfare benefit plans, policies or practices of the Purchaser or any of its Affiliates in which Affected Employees participate or are eligible to participate, and (ii) for the plan year in which the Closing Date occurs, cause any deductible, co-pay, coinsurance and covered out-of-pocket expenses paid by any Affected Employee (or covered dependent thereof) during the portion of the plan year in which such Affected Employee participated in an Employee Plan to be taken into account for purposes of satisfying the corresponding deductible, co-pay, coinsurance and maximum out-of-pocket provisions after the Closing Date under any applicable welfare benefit plan, policy or practices of the Purchaser or any of its Affiliates in the applicable plan year.
- (f) Nothing contained in this Section 6.2 shall (i) be treated as the establishment, amendment or modification of any Employee Plan or constitute a limitation on rights to amend, modify, merge or terminate any Employee Plan, (ii) give to any Person any third party beneficiary or other rights under this Agreement or otherwise, or (iii) obligate the Company, the Purchaser, or any of their Affiliates to maintain any particular Employee Plan or retain the employment or services of any current or former employee, director or other service provider of the Company for any specific period of time.
- (g) The Company shall take (or cause to be taken) all action necessary or appropriate to terminate, effective no later than the day immediately preceding the Closing Date, any Employee Plan that contains a cash or deferred arrangement intended to qualify under Section 401(a) of the Code (the "401(k) Plans"), unless the Purchaser, in its sole and absolute discretion, agrees to sponsor and maintain any such 401(k) Plan by providing the Company with written notice of such election (an "Election Notice") at least three (3) days before the Closing Date. Unless the Purchaser timely provides an Election Notice to the Company, the Company shall deliver to Purchaser, prior to the Closing Date, evidence that the Company's board of directors has validly adopted resolutions to terminate the 401(k) Plans (the form and substance of which resolutions shall be subject to review and approval of the Purchaser), effective no later than the date immediately preceding the Closing Date.
- 6.3 <u>Record Retention</u>. The Parties agree that for a period of seven (7) years after the Closing Date, or for a longer period if required by applicable Law, without the prior written consent of the Sellers' Representative (which consent shall not be unreasonably withheld, delayed or conditioned), neither the Purchaser nor any of its Affiliates (including the Company) shall dispose of or destroy any of the books and records of the Company which may be reasonably relevant to any legal, regulatory or Tax audit, investigation, inquiry or requirement of any of the Sellers without first offering such records to the Sellers' Representative; <u>provided</u>, that the Company shall not be required to offer any such books and records to the Sellers to the extent such action would constitute a waiver of the attorney-client privilege or to supply the Sellers with any information which, in the reasonable judgment of the Purchaser, the Purchaser is under a legal obligation not to supply; <u>provided</u>, that in each such case, the Purchaser shall use reasonable best efforts to provide an alternative means of disclosing such information.

6.4 Indemnification of Directors and Officers; Insurance.

- (a) The Purchaser agrees that all rights to indemnification, advancement of expenses and exculpation now existing in favor of each Person who, as of the Closing Date, is a current or former manager, officer, or member of the Board of Managers as described in Section 7 of the Operating Agreement (collectively, the "Covered Persons") pursuant to the applicable Organizational Documents of the Company in effect on the date hereof and any individual indemnity agreements in existence on the date hereof and set forth on Section 6.4(a) of the Company Disclosure Schedule (the "Indemnification Agreements") shall survive the Closing and shall continue in full force and effect in accordance with their terms for a period of six (6) years from the Closing Date. Following the Closing, neither the Purchaser nor the Company shall amend, repeal or otherwise modify such arrangements in any manner that would adversely affect the rights of the Covered Persons thereunder for a period of six (6) years from the Closing Date.
- (b) For a period of six (6) years from the Closing Date, the Purchaser shall cause the Company to honor, to the fullest extent permitted by applicable Law, all of the obligations of the Company to indemnify (including any obligations to advance funds for expenses) the Covered Persons to the extent that such obligations of the Company exist on the Closing Date pursuant to the applicable Organizational Documents of the Company in effect on the date hereof or the Indemnification Agreements; <u>provided</u> that such indemnification rights shall not apply to any Covered Person with respect to any Liability for which such Covered Person is obligated to indemnify the Purchaser under <u>Article VII</u> of this Agreement.
- (c) In the event that the Purchaser, the Company or any of their respective successors or assigns after the applicable Closing Date (i) consolidates with or merges into any other Person and shall not be the continuing or surviving corporation or entity of such consolidation or merger or (ii) transfers or conveys all or a substantial portion of its properties and assets to any Person, then, and in each such case, the Purchaser shall use commercially reasonable efforts to cause the successors and assigns of the Purchaser, the Company or of their respective successors or assigns to assume the obligations of the Purchaser and/or the Company or their respective successors or assigns as contemplated by this Section 6.4.
- (d) The provisions of this Section 6.4 shall survive the consummation of the Closing and expressly are intended to benefit each of the Covered Persons. Notwithstanding anything to the contrary, it is agreed that the rights of a Covered Person under this Section 6.4 shall be in addition to, and not a limitation of, any other rights such Covered Person may have under the Organizational Documents of the Company in effect on the date hereof or the Indemnification Agreements, and nothing in this Section 6.4 shall have the effect of, or be construed as having the effect of, reducing the benefits to the Covered Persons under such arrangements.

6.5 Tax Matters.

(a) <u>Pre-Closing Income Tax Returns</u>. At the sole expense of the Sellers, the Sellers' Representative shall prepare, or cause to be prepared, all income, franchise or other similar Tax Returns with respect to the Company for which the items of income, deductions, credits, gains or losses are passed through to the Sellers for taxable periods ending on or prior to the Closing Date (a "<u>Pre-Closing Tax Period</u>"), including those that are filed after the Closing Date, in accordance with the provisions of this Agreement and in a manner that is consistent with past practice except for changes in applicable Law. The Company, or Purchaser, as applicable, shall cooperate with the Sellers' Representative to file, or cause to be filed, such Tax Returns on a timely basis, as applicable. The Purchaser shall file or cause to be filed all other Tax Returns of the Company that are required to be filed after the Closing Date (including those for a Straddle Period), and any such Tax Returns that relate to a Pre-Closing Tax Period shall be prepared and filed consistently with past practices, unless otherwise required by applicable Law. The Purchaser shall provide any such Tax Return that reflects a Tax for which the Sellers may be responsible pursuant to this Agreement to the Sellers' Representative for its

review and comment fifteen (15) days prior to the due date for filing and shall reflect, or cause to be reflected, any of the Sellers' Representative's reasonable comments on such Tax Return.

(b) Cooperation on Tax Matters.

(i) The Purchaser, the Company and the Sellers shall cooperate fully, as and to the extent reasonably requested by the other Party, in connection with the filing of Tax Returns pursuant to this Section 6.5 and any audit or Action with respect to Taxes. Such cooperation shall include the retention and (upon the other Party's request) the provision of records and information which are reasonably relevant to any such audit or Action and making employees available on a mutually convenient basis to provide additional information and explanation of any material provided hereunder. The Company and the Sellers agree (A) to retain all books and records with respect to Tax matters pertinent to the Company relating to any taxable period beginning before the Closing Date until the expiration of the statute of limitations (and, to the extent notified by the Purchaser or the Sellers, any extensions thereof) of the respective taxable periods, and to abide by all record retention agreements entered into with any taxing authority, and (B) to give the other Party reasonable written notice prior to transferring, destroying or discarding any such books and records and, if the other Party so requests, the Company or the Sellers, as the case may be, shall allow the other Party to take possession of such books and records.

(ii) The Purchaser and the Sellers further agree, upon request, to use their commercially reasonable efforts to obtain any certificate or other document from any Governmental Entity or any other Person as may be necessary to mitigate, reduce or eliminate any Tax that could be imposed with respect to the Transactions.

- (c) Amended Tax Returns. After the Closing, the Purchaser and its Affiliates shall not, and shall not cause or permit the Company to (i) amend or cause to be amended any income, franchise or other similar Tax Return of the Company (or file any new income, franchise or other similar Tax Return of the Company in a jurisdiction with which the Company has not previously filed Tax Returns) for a Pre-Closing Tax Period, (ii) make any election with respect to income, franchise or other similar Taxes that has a retroactive effect with respect to any Pre-Closing Tax Period, or (iii) initiate any voluntary disclosure agreement or similar process with any Governmental Entity with respect to any Pre-Closing Tax Period, in each case, without the prior written consent of the Sellers' Representative (which consent shall not be unreasonably withheld, delayed or conditioned).
- (d) <u>Transaction Tax Deductions</u>. To the maximum extent permitted by applicable Law, Transaction Tax Deductions shall be reflected on the Tax Returns of the Sellers and the Company, as applicable, for the Pre-Closing Tax Period. For this purpose, "Transaction Tax Deductions" means any item of loss, deduction, or credit of the Sellers or the Company resulting from or attributable to fees, costs, and expenses of Sellers or the Company related to or arising out of the Transactions, and which (in each case) are (i) paid by Sellers or the Company prior to the Closing Date, (ii) reflected as a current liability in the calculation of Net Working Capital Amount, or (iii) included in the calculation of Closing Transaction Expenses.

(e) Audits.

(i) The Purchaser shall provide the Sellers' Representative with notice of any written inquiries, audits, examinations or proposed adjustments by the IRS or any other Tax authority, which relate to any Pre-Closing Tax Periods of the Company ("<u>Tax Proceedings</u>") within twenty (20) days of the receipt of such notice. The Purchaser shall have the sole right to represent the interests of the Company in any Tax Proceedings, <u>provided</u>, the Sellers' Representative shall have the right to (A) control any Tax Proceedings related to an income, franchise or other similar Tax Return for a Pre-Closing Tax Period and (B) participate in

any other Tax Proceedings and to employ counsel of its choice at its own expense. The Purchaser shall not settle any such Tax Proceedings that could reasonably be expected to materially increase the Taxes for which the Sellers would be liable under applicable Law or for which the Sellers are required to indemnify the Purchaser Indemnified Persons pursuant to this Agreement without the consent of the Seller's Representative, which consent shall not be unreasonably withheld, delayed or conditioned.

- (ii) The Purchaser and the Company shall execute and deliver to the Sellers' Representative such powers of attorney and other documents as may be necessary or appropriate to give effect to the foregoing participation rights.
- (iii) To the extent the provisions of this <u>Section 6.5(e)</u> conflict with the provisions of <u>Section 7.5(b)</u>, the provisions of this <u>Section 6.5(e)</u> shall control.
- (f) <u>Certain Taxes</u>. All transfer, documentary, sales, use, stamp, registration and other such Taxes and fees, including any related penalties and interest, incurred in connection with the Transaction Documents ("<u>Transfer Taxes</u>") shall be borne fifty percent (50%) by the Sellers, on the one hand, and fifty percent (50%) by the Purchaser, on the other hand. The party required by Law to file any necessary Tax Returns with respect to such Transfer Taxes shall file all such Tax Returns and other documentation with respect to all such Transfer Taxes and pay all such Taxes, and the non-filing Parties shall timely reimburse the filing Parties upon receipt of written notice of payment of any such Transfer Taxes. The Parties shall cooperate to minimize any Transfer Taxes to the extent permitted by applicable Law.
- (g) Tax Refunds. The Sellers shall have the right to any Tax refunds actually received by the Company arising from any payments for Taxes made by the Company relating to any Pre-Closing Tax Period or in respect of refundable credits for amounts paid prior to the Closing (to the extent such refunds are not reflected as an asset in the calculation of the Net Working Capital Amount, as finally determined). The Purchaser shall cooperate with the Sellers' Representative to file any applicable Tax Returns to claims such Tax refunds or credits and shall pay such amounts to the Sellers no later than ten (10) days after the receipt or availability of such refunds, less reasonable costs incurred in obtaining such refund. Notwithstanding anything in this Agreement to the contrary, in the event that a cash refund of Taxes that relate to any Pre-Closing Tax Period is subsequently determined by any Governmental Entity to be less than the amount paid by the Purchaser to the Sellers, the Sellers shall promptly return any such disallowed amounts (plus any interest in respect of such disallowed amount owed to a Governmental Entity) to the Company.
- (h) <u>Tax Treatment</u>. The Parties agree to treat for all U.S. federal income Tax purposes (i) the purchase and sale of the Units pursuant to this Agreement in accordance with Revenue Ruling 99-6 (Situation 2) as a sale of the interests of a partnership by the Sellers and a purchase of the Company's assets by the Purchaser, (ii) the Note as not "readily tradable", within the meaning of Section 453(f)(5) of the Code, (iii) the Purchaser Shares as not issued and outstanding until the Purchaser Shares Issuance Date, and (iv) the purchase and sale of Units pursuant to this Agreement as an "installment sale", within the meaning of Section 453(b) of the Code. The Purchaser, the Company, and the Sellers shall (i) treat and report the transactions contemplated by this Agreement in all respects consistently with this <u>Section 6.11(h)</u> for all U.S. federal, state, local and foreign Tax purposes and (ii) not take any actions or positions inconsistent with the foregoing, unless required by a Tax Authority.
- (i) <u>Purchase Price Allocation</u>. The Purchaser, at its expense, shall engage an independent third party valuation firm for the purposes of appraising and valuing the assets and allocating the Aggregate Consideration among the assets of the Company in accordance with Section 1060 of the Code and the regulations promulgated thereunder (the "<u>Allocation</u>") and in a manner consistent with <u>Exhibit C</u>. The Purchaser shall provide the Seller Representative with the Allocation within one hundred twenty (120) days of

the Closing Date. The Seller Representative shall provide its comments on the Allocation within thirty (30) days of the receipt of the Allocation, and the Purchaser shall consider in good faith the Seller Representative's reasonable comments and deliver a final Allocation to the Seller Representative within thirty (30) days after receipt of any such comments from the Seller Representative. The final Allocation shall be conclusive and binding upon the Sellers, Purchaser and the Company for all purposes, and the Parties agree that all returns and reports (including IRS Form 8594) and all financial statements shall be prepared in a manner consistent with (and the Parties shall not otherwise take a position on a Tax Return that is inconsistent with) the Allocation unless otherwise required by Law.

(j) Form W-9. At the Closing, each Seller shall deliver a properly executed IRS Form W-9 to the Purchaser.

6.6 Sellers' Representative.

(a) Each Seller, by virtue of its execution of this Agreement, hereby makes, constitutes and appoints the Sellers' Representative, with full power of substitution and re-substitution, as its sole, exclusive, true and lawful agent, representative and attorney-in-fact for such Seller with respect to any and all matters relating to, arising out of, or in connection with this Agreement, including, in such Seller's name, place, and stead to sign, execute, deliver and perform any Transaction Documents required to be executed by such Seller (or any Transaction Documents by which such Seller is otherwise bound), to make and authorize amendments to, or waivers of, this Agreement or any other Transaction Document, to enforce the obligations of the Purchaser or the Company under this Agreement or any other Transaction Document, to give and receive all notices required or permitted by the Sellers' Representative under this Agreement or any other Transaction Document, and to defend and/or settle any indemnification claims made by the Purchaser or the Company or any other Indemnified Person pursuant to the terms of this Agreement or any other Transaction Document. This power of attorney is a special power of attorney coupled with an interest and is irrevocable, and shall survive the Closing and death, disability, legal incapacity, bankruptcy, insolvency, dissolution, or cessation of existence of any Seller. This power of attorney may be exercised by the Sellers' Representative by listing the Seller executing any Transaction Document with the signature of the Sellers' Representative acting as attorney-in-fact for such Seller. The Sellers' Representative (in its capacity as such) shall have no Liability to the Purchaser, the Company or the Sellers, or any of their respective Affiliates or Representatives, with respect to any action taken or omitted to be taken in its capacity as the Sellers' Representative, except with respect to any Liability resulting from the Sellers' Representative's gross negligence or willful misconduct. Each Seller hereby forever releases and discharges the Sellers' Representative from any and all Liability which may arise in connection with the Sellers' Representative's performance in good faith and any acts or omissions which such Sellers' Representative takes on behalf of the Sellers and their shareholders in accordance with the terms of this limited power of attorney, except in the case of gross negligence or willful misconduct of the Sellers' Representative.

(b) Each Seller shall indemnify and hold harmless and reimburse the Sellers' Representative from and against such Seller's Pro Rata Portion of any and all Liabilities, losses, damages, claims, costs or expenses suffered or incurred by the Sellers' Representative arising out of or resulting from any action taken or omitted to be taken by the Sellers' Representative under this Agreement or the other Transaction Documents, other than such Liabilities, losses, damages, claims, costs or expenses arising out of or resulting from the Sellers' Representative's gross negligence or willful misconduct, and in the event there are any remaining funds in the Escrow Account to be distributed to the Sellers at the termination of the Escrow Account pursuant to Section 2.4, such funds shall first satisfy any such loss, Liability, or expense incurred by the Sellers' Representative and thereafter the Sellers' Representative shall be entitled to recover any such losses, Liabilities or expenses from the Escrow Account prior to the distribution of funds to the Sellers.

- (c) The Purchaser is hereby relieved from any Liability to any Person for any action taken by or on behalf of the Purchaser in accordance with any decision, act, consent, or instruction of the Sellers' Representative.
- (d) The Sellers' Representative may only be removed upon delivery of written notice to Purchaser signed by the Sellers owning at least sixty-seven percent (67%) of the outstanding Units immediately prior to the Closing. If the Sellers' Representative resigns, dies or becomes legally incapacitated, the Sellers owning at least a majority of the outstanding Units immediately prior to the Closing promptly shall designate in writing to the Purchaser a single Person to fill the Sellers' Representative vacancy as the Sellers' Representative hereunder. If at any time there shall not be a Sellers' Representative or the Sellers fail to designate a successor Sellers' Representative, then the Purchaser may have a court of competent jurisdiction appoint such Sellers' Representative hereunder.
- 6.7 <u>R&W Policy</u>. From and after the Closing, the Purchaser shall use commercially reasonable efforts to maintain in full force and effect the R&W Policy. The Purchaser shall pay and be responsible for (x) the premium of the R&W Policy and (y) any and all underwriting fees, brokerage fees, and legal fees for counsel engaged by the insurer of the R&W Policy, surplus lines Tax and any other costs and expenses associated with obtaining the R&W Policy and binding coverage thereunder. The Purchaser agrees not to take any action to amend, restate or terminate any provision of the R&W Policy in a manner adverse to the Sellers without the express prior written consent of the Sellers' Representative.

6.8 Reserved.

6.9 Release. Effective as of the Closing Date, each of the Sellers, on behalf of itself and its Affiliates, legal representatives, successors and assigns (collectively, the "Releasors"), hereby releases, acquits and forever discharges, to the fullest extent permitted by Law, including by contractually shortening any applicable statute of limitations, each of the Purchaser, the Company and each of their respective past, present or future officers, managers, directors, shareholders, partners, members, Affiliates, employees, counsel, agents and other representatives (each a "Releasee") of, from and against any and all actions, causes of action, claims, demands, damages, judgments, debts, dues and suits of every kind, nature and description whatsoever (collectively "Claims") which such Person ever had, now has or may have on or by reason of any matter, cause or thing whatsoever related to the Company prior to the Closing; provided, however, that the foregoing release shall not cover (and the definition of Claims shall not include) (i) any rights and interest of the Releasors expressly provided for in this Agreement and the Transaction Documents, (ii) any rights to or claims of indemnification, exculpation or advancement of expenses provisions for the benefit of directors, managers, officers, employees or other individuals contained in the Indemnification Agreements, any insurance policies or the Company's Organizational Documents, (iii) any rights to which any Releasor (and any owner thereof) is entitled by virtue of his or her status as an employee or consultant of any Releasee pursuant to any Employee Plan (including earned employee benefits that remain unpaid or unused accrued vacation as of the Closing Date) and any accrued compensation that remains unpaid as of the Closing Date and specifically set forth on Section 6.9 of the Company Disclosure Schedule, (iv) any claims for reimbursement for expenses incurred in the ordinary course of each such Releasor's employment or service to any Releasee which are reimbursable under the expense reimbursement policies of such Releasee, (v) any claims pursuant to any written commercial agreement set forth on Section 6.9 of the Company Disclosure Schedule between a Releasor, on the one hand, and a Releasee, on the other hand, in effect on the date hereof, and (vi) any claims that are not permitted to be released under applicable Law. Each Seller agrees not to, and agrees to cause its respective controlled Affiliates not to, assert any Claim with respect to the matters expressly released hereby against the Releasees. In addition, each Seller specifically waives the benefit of any and all provisions, rights and benefits conferred by any law of any state or territory of the United States, or principle of common law, which is similar, comparable or equivalent to California Civil Code section 1542, as well as the rights and benefits conferred by California Civil

Code section 1542 itself, which reads as follows: A GENERAL RELEASE DOES NOT EXTEND TO CLAIMS WHICH THE CREDITOR DOES NOT KNOW OR SUSPECT TO EXIST IN HIS FAVOR AT THE TIME OF EXECUTING THE RELEASE, WHICH IF KNOWN BY HIM MUST HAVE MATERIALLY AFFECTED HIS SETTLEMENT WITH DEBTOR. Each Seller understands and acknowledges the significance and consequence of this specific waiver. Each Seller acknowledges that such Seller has received or has had the opportunity to receive independent legal advice from its attorneys regarding this waiver, and each Seller hereby assumes full responsibility for any damages, loss or liability which it may hereunder incur by reason of such waiver.

- 6.10 Settlement Amounts. From and after the Closing, the Purchaser shall exercise commercially reasonable efforts to enter into a settlement agreement with respect to the litigation referred to on Section 6.10 of the Company Disclosure Schedule (the "Litigation") on the terms, in all material respects, as those set forth in the draft settlement agreement in respect thereof made available to the Purchaser prior to the Closing Date; provided, that in no event shall the Purchaser or the Company be required to incur out-of-pocket expenses in excess of the amount referenced as the "maximum Litigation out-of-pocket expenses" set forth on Section 6.10 of the Company Disclosure Schedule in connection with the exercise of such commercially reasonable efforts. IriSys shall have the right to any and all amounts actually received by the Company (through installment or otherwise) incident to the Litigation (the "Recovery"), whether by judgment, settlement, a combination thereof, and is an asset owned by IriSys, and the Purchaser waives and releases any and all claims it may have therein and thereto (to the extent such amounts are not reflected as an asset in the calculation of the Net Working Capital Amount, as finally determined); provided, however, that the amount of any Recovery under this Section 6.10 shall be net of any reasonable out-of-pocket collection costs of the Purchaser or any subsidiary thereof (including the Company) incurred after the Closing in obtaining such amounts (including payment of any legal expenses of outside counsel of the Purchaser or any subsidiary thereof (including the Company) related thereto that are incurred after the Closing, and to the extent that any such legal expenses of the Company are outstanding and unpaid as of the Closing, such legal expenses outstanding as of the Closing) net cash of Taxes of the Company incurred in the year of and in connection with such receipt and any payments made hereunder. The Purchaser shall cause the Company to pay any Recovery to IriSys within ten (10) B
- 6.11 Employee Equity Grants. On or promptly following the Closing Date, and subject in all cases to the approval of the board of directors of the Purchaser (which approval the Purchaser shall use its reasonable best efforts to so promptly obtain), the Purchaser shall grant to certain employees and service providers of the Company as of the Closing Date, in consultation with the Sellers' Representative, restricted stock units and/or options to purchase shares of Purchaser Common Stock set forth on Section 6.11 of the Disclosure Schedules, which restricted stock units and options to purchase shares of Purchaser Common Stock shall be issued pursuant to the Purchaser's Amended and Restated 2018 Equity Incentive Plan.
- 6.12 <u>Patent Matters</u>. Following the Closing, in the event Purchaser reasonably requests, in writing to Irisys within one year following the Closing, a non-exclusive license to use the patent rights set forth on Section 6.12 of the Disclosure Schedules (the "<u>Licensed Technology</u>") solely for the use with a client of the Company in the Company's ordinary course of business following the Closing, IriSys agrees to negotiate in good faith for a reasonable period of time with the Purchaser and the Company for such a non-exclusive license, which shall be on customary and market terms for such a non-exclusive license.

ARTICLE VII INDEMNIFICATIONS; SURVIVAL

- 7.1 Survival of Representations, Warranties and Covenants. All representations, warranties and covenants contained in this Agreement shall survive the Closing and remain in full force and effect for the duration of the Indemnity Period, other than with respect to (a) (i) the Company Fundamental Representations, which shall survive until six (6) years following the Closing Date and (ii) the representations and warranties of the Company set forth in Section 5.8 (the "Company Tax Representations") which shall survive until the expiration of the applicable statute of limitations with respect to the applicable matter; and (b) each covenant or agreement contained in this Agreement that is to be performed on or following the Closing Date which shall survive the Closing and remain in full force and effect until such covenant or agreement has been fully performed or fulfilled in accordance with its terms (the respective expiration dates for the survival of the representations and warranties and covenants and agreements shall be referred to herein as the "Expiration Date"), except that any representation, warranty, covenant or agreement that would otherwise terminate in accordance with clause (a) or (b) will continue to survive if a Claim Notice shall have been timely given to the Indemnifying Person by the Indemnified Person (which if the Sellers are the Indemnifying Persons, the Indemnified Person shall provide such Claim Notice to the Sellers' Representative) on or prior to such applicable Expiration Date, until the related claim for indemnification has been satisfied or otherwise resolved as provided in this Article VII.
- 7.2 <u>Indemnification by the Sellers</u>. Subject to the terms, conditions and limitations of this <u>Article VII</u>, following the Closing, the Sellers shall, on a several (and not joint and several) basis (based on their relative Pro Rata Portion), indemnify and defend the Purchaser and its Affiliates (including, after the Closing, the Company), and their respective successors, assigns, officers, directors, stockholders, members, employees and agents (collectively, the "<u>Purchaser Indemnified Persons</u>" and each, a "<u>Purchaser Indemnified Person</u>") against, and hold them harmless from, any Loss suffered or incurred by any such Purchaser Indemnified Person may be or become subject, arising or resulting from or based upon:
 - (a) any breach of any representation or warranty of any Seller contained in this Agreement;
- (b) any breach of any representation or warranty of the Company contained in this Agreement or in any certificate delivered to Purchaser hereunder:
 - (c) any breach of any covenant by any Seller or the Sellers' Representative contained in this Agreement;
 - (d) any breach of any covenant by the Company (prior to the Closing) contained in this Agreement;
- (e) any Closing Indebtedness or any Closing Transaction Expenses that are not specifically included in the calculation of the Aggregate Consideration;
- (f) any inaccuracy in the Consideration Spreadsheet or the distribution of any portion of the Aggregate Consideration by the Purchaser in the manner set forth in the Consideration Spreadsheet or as otherwise directed by the Sellers' Representative; and
- (g) any Taxes of the Company or the Sellers for any taxable period ending on or before the Closing Date or for the portion of the Straddle Period through the end of the Closing Date (to the extent not

included as a current liability in the calculation of the Net Working Capital Amount, as finally determined) and 50% of the Transfer Taxes.

7.3 <u>Indemnification by the Purchaser</u>. Subject to the terms, conditions and limitations of this <u>Article VII</u>, following the Closing, the Purchaser shall indemnify and defend the Sellers and their respective successors, assigns, officers, directors, stockholders, members, employees and agents (collectively, the "<u>Seller Indemnified Persons</u>" and each, a "<u>Seller Indemnified Person</u>"), against, and hold them harmless from, any Loss suffered or incurred by any such Seller Indemnified Person, or to which such Seller Indemnified Person may be or become subject, arising or resulting from or based upon: (a) any breach of any representation or warranty of the Purchaser contained in this Agreement or in any certificate delivered by the Purchaser hereunder; and (b) any breach of any covenant by the Purchaser contained in this Agreement.

7.4 <u>Indemnity Limitations</u>.

- (a) Notwithstanding anything to the contrary contained in this Agreement:
- (i) The Purchaser Indemnified Persons shall not be entitled to indemnification pursuant to Section 7.2(a) or Section 7.2(b) unless and until the aggregate of all Losses arising from indemnity claims made by Purchaser Indemnified Persons under Section 7.2(a) or Section 7.2(b) exceeds Two Hundred Fifty Thousand Dollars (\$250,000) (the "Basket Amount"), and if the aggregate amount of such indemnifiable Losses under Section 7.2(a) or Section 7.2(b) reaches the Basket Amount, the Purchaser Indemnified Persons shall be entitled to seek indemnity recourse for such indemnifiable Losses solely in excess of the Basket Amount; provided, that, the foregoing indemnification limitations shall not apply to indemnity claims made by Purchaser Indemnified Persons pursuant to Section 7.2(a) or Section 7.2(b) with respect to Company Fundamental Representations or Company Tax Representations.
- (ii) The aggregate Liability of the Sellers under Section 7.2(a) or Section 7.2(b) will not exceed the Indemnity Escrow Amount remaining at any given time in the Escrow Account and the Purchaser, on behalf of itself and the other Purchaser Indemnified Persons, agrees not to seek, and shall not be entitled to recover, any Losses in excess of such Indemnity Escrow Amount (and the R&W Policy shall be the sole source of recourse for the Purchaser Indemnified Persons) pursuant to Section 7.2(a) or Section 7.2(b); provided, that, the foregoing indemnification limitations shall not apply to indemnity claims made by the Purchaser Indemnified Persons pursuant to Section 7.2(a) or Section 7.2(b) with respect to Company Fundamental Representations or Company Tax Representations and under Section 7.2(a) or Section 7.2(b) with respect to with respect to Company Fundamental Representations or Company Tax Representations and under Section 7.2(c) through Section 7.2(g) shall not exceed, in the aggregate, the Aggregate Consideration actually received by the Sellers hereunder (the "Cap") and the Purchaser, on behalf of itself and the other Purchaser Indemnified Persons, agrees not to seek, and shall not be entitled to recover, any Losses or other payments pursuant to claims made by Purchaser Indemnified Persons in excess of the Cap.
- (iii) With respect to any indemnifiable Losses arising out of or resulting from any breach by any Seller under Section 7.2(a) or Section 7.2(c), Purchaser, on behalf of itself and the other Purchaser Indemnified Persons, will only have the right to seek recourse directly against such applicable Seller who committed such breach of Section 7.2(a) or Section 7.2(c), as applicable, subject to the other terms, conditions and limitations contained herein

(iv) Prior to seeking recovery directly from the Sellers pursuant to claims made by the Purchaser Indemnified Persons pursuant to Section 7.2(a) or Section 7.2(b) with respect to Company Fundamental Representations or Company Tax Representations, the Purchaser Indemnified Persons shall first

seek recourse against the Indemnity Escrow Amount and the R&W Policy and to the extent that the Indemnity Escrow Amount and R&W Policy is fully exhausted or recovery therefrom has been denied, the Purchaser Indemnified Persons may seek recourse directly against the Sellers on a several (and not joint and several) basis (based on their relative Pro Rata Portion), subject to the terms, conditions and limitations contained herein; provided, that in the case of any Seller who holds (or will hold) Purchaser Shares as of the applicable payment date for such indemnification obligation, any such direct payment obligation may be satisfied at such Seller's option by the forfeiture and delivery (if applicable) to the Purchaser of a number of Purchaser Shares (valued for such purpose at the closing price of the Purchaser Common Stock as of the trading day immediately prior to the applicable payment date) having an aggregate value equal to the amount of such Seller's applicable portion of such indemnification obligation, and such Seller and the Purchaser shall take such actions as are reasonably necessary to transfer (or forfeit, if applicable) such number of Purchaser Shares to the Purchaser; provided, further, that if any payment obligations remain outstanding to any Seller under the Note as of the applicable payment date, such direct payment obligation may also be satisfied at such Seller's option by a decrease to the outstanding principal amount of the Note owing to such Seller.

- (v) The Purchaser Indemnified Persons shall have no right to set off against any Closing Adjustment Surplus Amount or other amount payable to the Sellers (or their Affiliates) under this Agreement or any other Transaction Document in satisfaction of any indemnity rights that the Purchaser Indemnified Persons may have under this Agreement.
- (vi) The limitations set forth in this <u>Section 7.4(a)</u> or otherwise under this <u>Article VII</u> shall not apply, and nothing herein shall be construed to limit any Purchaser Indemnified Person's rights or remedies, in the case of any claim for fraud.
 - (b) Notwithstanding anything to the contrary contained in this Agreement:
- (i) The Seller Indemnified Persons shall not be entitled to indemnification pursuant to Section 7.3(a) unless and until the aggregate of all Losses arising from indemnity claims made by Seller Indemnified Persons hereunder exceeds the Basket Amount, and if the aggregate amount of such indemnifiable Losses reaches the Basket Amount, the Seller Indemnified Persons shall be entitled to seek indemnity recourse for any such indemnifiable Losses in excess of the Basket Amount; provided, that, the foregoing indemnification limitations shall not apply to indemnity claims made by Seller Indemnified Persons pursuant to Section 7.3(a) with respect to Purchaser Fundamental Representations.
- (ii) The aggregate Liability of the Purchaser under Section 7.3 will not exceed the Cap and the Sellers, on behalf of themselves and the other Seller Indemnified Persons, agree not to seek, and shall not be entitled to recover, any Losses or other payments in excess of the Cap; provided, that, the foregoing limitation shall not apply to any breach by the Purchaser of Section 2.4 or any other obligations of the Purchaser to pay any portion of the Aggregate Consideration or any other consideration required to be paid to the Sellers hereunder.
- (c) The amount of any Loss for which indemnification is provided under this <u>Article VII</u> shall be net of (i) any amounts actually recovered by the Indemnified Person from any third party (including insurance proceeds, including under the R&W Policy) as a result of the facts or circumstances giving rise to the Losses, to the extent such recovery would result in duplicative recoveries to such Indemnified Person for the same Losses and (ii) any Tax benefits that are actually realized by Indemnified Person as a result of the incurrence of Losses from which indemnification is sought in the taxable year in which such indemnification payment is made, the immediately subsequent taxable year; <u>provided</u>, that for the avoidance of doubt, no Purchaser Indemnified Person shall have any obligation to repay any amounts received by the Purchaser

Indemnified Parties from the Indemnity Escrow Amount and applied towards the satisfaction of the retention under the R&W Policy.

- (d) Each Indemnified Person shall mitigate Losses suffered, incurred or sustained by such Indemnified Person arising out of any matter for which such Indemnified Person has sought indemnification hereunder, to the same extent required by Delaware Law.
- (e) In no event shall any Purchaser Indemnified Person be entitled to indemnification to this <u>Article VII</u> with respect to a specific Loss to the extent such Loss is actually and specifically included in the calculation of the Closing Cash Consideration, as finally determined in accordance with <u>Section 2.7</u>, including any such Loss that is related to any reserve or other similar item included in such calculation.
- (f) Any Liability for indemnification under this <u>Article VII</u> shall be determined without duplication of recovery by reason of the set of facts giving rise to such Losses constituting a breach of more than one representation, warranty, covenant or undertaking, or one or more rights to indemnification.
- (g) For purposes of determining (i) the amount of any Losses arising out of, relating to or resulting from any failure of any representation or warranty to be true and correct, and (ii) whether or not such failure has occurred, such representations and warranties shall be considered without giving effect to any limitation or qualifications as to "materiality," "Company Material Adverse Change", "Seller Material Adverse Change", or "Purchaser Material Adverse Change", or any other derivation of the word "material."
- (h) Notwithstanding any other provision hereof, no Seller shall be liable to any Purchaser Indemnified Persons for Losses indemnifiable pursuant to Section 7.2(a) to the extent arising from or related to the fraudulent conduct of another Seller, and instead liability for the entire amount of such Losses shall rest with such Seller who engaged in such fraudulent conduct, and such indemnification obligation shall not be limited to such Seller's Pro Rata Portion.
- (i) Notwithstanding anything to the contrary contained in this Agreement, the Purchaser Indemnified Persons shall not be entitled to indemnification under this Article VII for Losses with respect to (i) any Taxes that are attributable to a taxable period (or portion thereof) beginning after the Closing Date and that arise from a breach of representations and warranties set forth in Section 5.8 (Tax Matters) (other than Sections 5.8(c), (d) or (n)), or (ii) the amount, value or condition of, or any limitations on, any Tax asset or attribute (e.g., net operating loss or Tax credit) of the Company or any of its Subsidiaries, including the ability of the Purchaser or any of its Affiliates (including the Company and its Subsidiaries after the Closing) to utilize such Tax asset or attribute following the Closing, in each case, in a taxable period (or portion thereof) beginning after the Closing Date.
- (j) The indemnities provided under this <u>Article VII</u> are intended only for the Indemnified Persons and are in no way intended to, nor shall they, constitute an agreement for the benefit of, or be enforceable by, any other Person.
 - 7.5 <u>Procedures Relating to Indemnification</u>.

(a) Claims Procedure.

(i) An Indemnified Person shall give prompt written notice (a "<u>Claim Notice</u>") to the Indemnifying Person (which if the Sellers are the Indemnifying Persons pursuant to <u>Section 7.2</u>, the Indemnified Person shall provide such Claim Notice to the Sellers' Representative for all purposes of this <u>Article VII</u>) after the Indemnified Person first becomes aware of any event or other facts that has resulted or

that would reasonably be expected to result in any Loss for which the Indemnified Person is entitled to any indemnification under this Agreement, and such notice shall contain (A) a description and, if known, the estimated amount of any Loss incurred or reasonably expected to be incurred by the Indemnified Person, (B) a reasonable explanation of the basis for the Claim Notice to the extent of the facts then known by the Indemnified Person and (C) a demand for payment of such Loss; <u>provided</u> that failure to give such notification shall not affect the indemnification provided hereunder except to the extent, and only to the extent that, the Indemnifying Person shall have been actually and materially prejudiced as a result of such failure.

(ii) The Sellers' Representative (if the Indemnifying Persons are the Sellers pursuant to Section 7.2) or the Purchaser (if the Indemnifying Persons are the Purchaser pursuant to Section 7.3) may in good faith, at any time on or before the tenth Business Day following its receipt of a Claim Notice (the "Objection Period"), object to the claim made in such Claim Notice by delivering written notice to the Indemnified Person and the Escrow Agent (a "Claim Objection"). The Claim Objection shall set forth in reasonable detail the good faith reasons for the objection to such claim for indemnification, and the amount of any claimed Loss which is disputed. If the Sellers' Representative (if the Indemnifying Persons are the Sellers pursuant to Section 7.2) or the Purchaser (if the Indemnifying Persons are the Purchaser pursuant to Section 7.3), as applicable, does not timely deliver a Claim Objection, or delivers a Claim Objection that does not object to all of the Losses set forth in the Claim Notice, the Indemnifying Persons shall be deemed to have accepted and agreed with all or such portion of the claim and shall be conclusively deemed to have consented to the recovery by the Indemnified Person of all or such portion of the Losses specified in the Claim Notice. If the Sellers' Representative (if the Indemnifying Persons are the Sellers pursuant to Section 7.2) or Purchaser (if the Indemnifying Persons are the Purchaser pursuant to Section 7.3) timely delivers a Claim Objection, the Purchaser and the Sellers' Representative, as applicable, shall attempt in good faith to agree upon the rights of the respective parties with respect to the disputed items of Losses and if the parties are not able to fully resolve all such differences within sixty (60) days from the applicable party's receipt of a Claim Objection, the Purchaser or the Sellers' Representative, as applicable, shall have the right to pursue legal recourse pursuant to Section 7.5(b) below.

(b) Third-Party Claims

(i) In order for an Indemnified Person to be entitled to any indemnification provided for under this Agreement in respect of, arising out of or involving a claim or demand made by any Person against the Indemnified Person (a "Third-Party Claim"), such Indemnified Person must provide the Indemnifying Person with a Claim Notice regarding the Third-Party Claim promptly and in any event within ten (10) Business Days after receipt by such Indemnified Person of written notice of the Third-Party Claim; provided that failure to give such notification shall not affect the indemnification provided hereunder except to the extent, and only to the extent that, the Indemnifying Person shall have been actually and materially prejudiced as a result of such failure. Thereafter, the Indemnified Person shall deliver to the Indemnifying Person, as promptly as practicable after the Indemnified Person's receipt thereof and to the extent permitted by applicable Law, copies of all notices and documents (including court papers) received by the Indemnified Person relating to the Third-Party Claim.

(ii) Subject to Section 7.5(b)(iii), if a Third-Party Claim is made against an Indemnified Person, the Indemnifying Person will be entitled to assume the defense thereof with counsel selected by the Indemnifying Person and reasonably acceptable to the Indemnified Person. If the Indemnifying Person so elects to assume the defense of a Third-Party Claim, the Indemnifying Person will not be liable to the Indemnified Person for legal fees and expenses subsequently incurred by the Indemnified Person in connection with the defense thereof except as otherwise set forth herein. If the Indemnifying Person assumes such defense, the Indemnified Person shall have the right, at its own expense, to participate in the defense thereof and, at its own expense, to employ counsel reasonably acceptable to the Indemnifying Person, separate from the counsel

employed by the Indemnifying Person, it being understood that the Indemnifying Person shall control such defense. If the Indemnifying Person chooses to defend or prosecute any Third-Party Claim, all the Parties hereto shall reasonably cooperate in the defense or prosecution thereof. Such cooperation shall include the retention and (upon the Indemnifying Person's request) the provision to the Indemnifying Person of records and information which are reasonably relevant to such Third-Party Claim, and making officers, directors, employees and agents of the Indemnified Person reasonably available on a mutually convenient basis to provide information, testimony at depositions, hearings or trials, and such other assistance as may be reasonably requested by the Indemnifying Person and in a manner which does not unreasonably interfere with the business and operations of the Indemnified Person. Whether or not the Indemnifying Person shall have assumed the defense of a Third-Party Claim, the Indemnified Person shall not admit any Liability with respect to, or settle, compromise or discharge, such Third-Party Claim without the Indemnifying Person's prior written consent (which consent shall not be unreasonably withheld, delayed or conditioned). The Indemnified Person's prior written consent (which consent shall not be unreasonably withheld, delayed or conditioned); provided that the Indemnified Person shall agree to any such admission of Liability, settlement, compromise or discharge of a Third-Party Claim that the Indemnifying Person may recommend if and only if (A) the Indemnified Person will have no Liability with respect to such Third-Party Claim and will be released in full in connection with such Third-Party Claim; and (B) the same would not impose any injunction or other equitable remedies on the Indemnified Person or its business.

(iii) Notwithstanding anything contained herein to the contrary, if (A) the Indemnifying Person fails to notify the Indemnified Person within ten (10) Business Days after receipt of any Claim Notice of a Third-Party Claim that the Indemnifying Person elects to defend the Indemnified Person pursuant to the terms hereunder, (B) the Indemnifying Person elects to defend the Indemnified Person pursuant to the terms hereunder but, upon petition by the Indemnified Person, a court of competent jurisdiction rules that the Indemnifying Person has failed to diligently prosecute or settle the Third-Person Claim, (C) such Third-Party Claim seeks an injunction or other equitable relief against the Indemnified Party or alleges a criminal violation, (D) the conduct of such defense by the Indemnifying Party would be inappropriate due to a conflict between the Indemnified Party and the Indemnifying Party, (E) the amount in dispute exceeds the maximum amount for which an Indemnifying Party could be liable pursuant to this Article VII in light of the limitations on indemnification herein, (F) the Indemnifying Party does not agree in writing that the Indemnifying Party is obligated to pay for any Losses arising from or related to such Third-Party Claim (subject to the limitations on indemnification set forth in Section 7.4, and (G) the Third-Party Claim involves a customer, supplier or other material business relationship of the Company, then the Indemnified Person shall have the right to defend, at the sole cost and expense of the Indemnifying Person, the Third-Person Claim by all appropriate proceedings, which proceedings shall be prosecuted by the Indemnified Person to a final conclusion or settled, subject to the limitations on settlement by the Indemnified Person set forth in this Agreement and subject to the other terms, conditions and limitations of the indemnification obligations of the Indemnifying Person(s) under this Agreement.

7.6 Further Acknowledgement by the Purchaser.

- (a) The Purchaser has conducted its own independent review and analysis of the Evaluation Materials, the Company, the Company's business and the assets, Liabilities, results of operations and financial condition of the Company. The Purchaser has undertaken such investigation as it has deemed necessary in connection with the execution, delivery and performance of this Agreement and the other Transaction Documents and the consummation of the Transactions.
- (b) Without limiting the generality of the foregoing, no Purchaser Indemnified Person shall have any claim or right to recovery pursuant to this Agreement or otherwise, with respect to (i) any information,

documents or materials furnished, delivered or made available by the Sellers, the Company or its Affiliates, officers, directors, employees, agents or advisors to the Purchaser and/or its Affiliates or Representatives in certain "data rooms," management presentations or any other form in contemplation of the Transactions or (ii) any projections, forecasts, estimates, plans or budgets of future revenue, expenses or expenditures, future results of operations (or any component thereof), future cash flows (or any component thereof) or future financial condition (or any component thereof) of the Company or the future business, operations or affairs of the Company heretofore or hereafter delivered to or made available to the Purchaser and/or its Affiliates or Representatives.

- (c) The Purchaser acknowledges that it is consummating the Transactions without any representation or warranty, express or implied, by the Company or the Sellers or any of their respective Affiliates or Representatives except as expressly set forth in Article IV or Article V herein (as modified by the Company Disclosure Schedule) or any other Transaction Document to which the Company or the Sellers is a party or in any certificate or other document or instrument to be delivered to the Purchaser pursuant to this Agreement or any such other Transaction Document to which the Company or the Sellers is a party or in any certificate or other document or instrument to be delivered to the Purchaser pursuant to this Agreement or any such other Transaction Document and the Transaction Document, it is relying on its own investigation and analysis in entering into both the Transaction Documents and the Transactions.
 - (d) Notwithstanding the foregoing, nothing herein shall limit any rights or remedies of the Purchaser in the case of a claim for fraud.
- 7.7 <u>Tax Treatment of Indemnification Payments</u>. Any indemnification payments made to the Purchaser pursuant to this Agreement shall be treated as an adjustment to the Aggregate Consideration unless otherwise required by applicable Law.
- 7.8 Exclusive Remedy. Except for (a) the Sellers' Representative's indemnities under Section 6.6 and the remedies set forth in Section 2.7 with respect to the final determination of the Closing Cash Consideration and (b) any non-monetary equitable relief to which any Party hereto may be entitled from and after the Closing, the indemnification provisions contained in this Article VII are intended to provide the sole and exclusive remedy following the Closing as to all Losses any Party may incur arising from or relating to the Agreement and the Transactions, and each Party hereby waives, to the full extent they may do so, any other rights or remedies that may arise under any applicable Law.

ARTICLE VIII

CLOSING DELIVERABLES

- 8.1 <u>Closing Deliverables of the Company and the Sellers</u>. The Company and the Sellers, as applicable, shall have delivered, or cause to be delivered, to the Purchaser the following:
- (a) a certificate of a duly authorized officer of the Company, dated as of the Closing Date, expressly certifying: (A) a copy of the resolutions duly adopted by the managers of the Company authorizing the execution, delivery and performance by the Company of each Transaction Document to which the Company is a party and the consummation of the Transactions; (B) a certified copy of the certificate of formation of the Company; and (C) a certificate, dated not more than five (5) Business Days prior to the Closing, as to the good standing of the Company from the Secretary of State of California;

- (b) the Escrow Agreement, duly executed by the Sellers' Representative and the Escrow Agent;
- (c) the written resignations of all of the managers and officers of the Company (except as otherwise agreed by the Purchaser), effective as of the Closing, in a form reasonably satisfactory to the Purchaser;
- (d) certificates evidencing the Units, if applicable, duly endorsed in blank or accompanied by unit powers duly executed in blank, or such other certificates or instruments effective under applicable Law to transfer ownership of the Units, in each case, in form and substance reasonably satisfactory to the Purchaser for valid transfer of title to the Units to the Purchaser;
- (e) a payoff letter issued by each holder of Closing Indebtedness identified by the Purchaser to be repaid concurrently with the Closing, which sets forth (A) the amount required to repay in full all Indebtedness owed to such holder at the Closing (the "Closing Repaid Indebtedness"), (B) the wire transfer instructions for the repayment of such Closing Repaid Indebtedness to such holder, and (C) if applicable, a release of all Liens granted by the Company to such holder or otherwise arising with respect to such Closing Repaid Indebtedness, effective upon repayment of such Closing Repaid Indebtedness (collectively, the "Payoff Letters");
- (f) a properly executed certificate of non-foreign status from each Seller in a form reasonably acceptable to Purchaser conforming to the requirements of Treasury Regulations Section 1.1445-2(b)(2);
 - (g) a properly completed and executed IRS Form W-9 from the Company and each Seller;
 - (h) each Restrictive Covenant Agreement, duly executed by the parties thereto; and
 - (i) the Note, duly executed by the Purchaser.
 - 8.2 <u>Closing Deliverables of the Purchaser</u>. The Purchaser shall have delivered, or caused to be delivered, to the Sellers the following:
- (a) a certificate of a duly authorized officer of the Purchaser, dated as of the Closing Date, expressly certifying a copy of the resolutions duly adopted by the board of directors of the Purchaser authorizing the execution, delivery and performance by the Purchaser of each Transaction Document to which the Purchaser is a party and the consummation of the Transactions;
 - (b) the Escrow Agreement, executed by a duly authorized officer of the Purchaser and the Escrow Agent; and
 - (c) a copy of a filed Listing of Additional Shares application for the listing of the Purchaser Shares on NASDAQ.

ARTICLE IX

MISCELLANEOUS

- 9.1 <u>Governing Law</u>. This Agreement will be governed by and construed in accordance with the internal Laws of the State of Delaware applicable to agreements made and to be performed entirely within such State, without regard to the conflicts of Law principles that would require the application of any other Law.
- 9.2 <u>Jurisdiction</u>. Each of the Parties irrevocably consents to the exclusive jurisdiction and venue of any court within Wilmington, Delaware in connection with any Action based upon or arising out of this Agreement or the Transactions and irrevocably waives, to the fullest extent permitted by Law, any objection which it may now or hereafter have to the laying of the venue of any such Action in any such court or that any such Action which is brought in any such court has been brought in an inconvenient forum. Process in any such Action may be served on any Party anywhere in the world, whether within or without the jurisdiction of any such court.
- 9.3 Waiver of Jury Trial. THE PARTIES HEREBY WAIVE ANY RIGHT TO TRIAL BY JURY IN ANY ACTION ARISING OUT OF OR RELATING TO THIS AGREEMENT OR ANY TRANSACTION CONTEMPLATED HEREBY, WHETHER NOW EXISTING OR HEREAFTER ARISING, AND WHETHER IN CONTRACT, TORT OR OTHERWISE. THE PARTIES AGREE THAT ANY OF THEM MAY FILE A COPY OF THIS SECTION WITH ANY COURT AS WRITTEN EVIDENCE OF THE KNOWING, VOLUNTARY AND BARGAINED-FOR AGREEMENT AMONG THE PARTIES IRREVOCABLY TO WAIVE TRIAL BY JURY AND THAT ANY ACTION WHATSOEVER BETWEEN THEM RELATING TO THIS AGREEMENT OR ANY TRANSACTION CONTEMPLATED HEREBY SHALL INSTEAD BE TRIED IN A COURT OF COMPETENT JURISDICTION BY A JUDGE SITTING WITHOUT A JURY.
- 9.4 <u>Expenses</u>. Whether or not the Transactions are consummated, and except as otherwise provided in this Agreement, each Party to this Agreement will bear its respective fees, costs and expenses incurred in connection with the preparation, negotiation, execution and performance of this Agreement or the Transactions (including legal, accounting and other professional fees).
- 9.5 Attorneys' Fees. If any Action for the enforcement of this Agreement is brought with respect to or because of an alleged dispute, breach, default or misrepresentation in connection with any of the provisions hereof, the successful or prevailing Party shall be entitled to recover its reasonable attorneys' fees and other costs incurred in that Action, in addition to any other relief to which it may be entitled.
- 9.6 Waiver; Remedies Cumulative. The rights and remedies of the Parties to this Agreement are cumulative and not alternative. Neither any failure nor any delay by any Party in exercising any right, power or privilege under this Agreement or any of the other Transaction Documents will operate as a waiver of such right, power or privilege, and no single or partial exercise of any such right, power or privilege will preclude any other or further exercise of such right, power or privilege or the exercise of any other right, power or privilege. To the maximum extent permitted by applicable Law, (a) no claim or right arising out of this Agreement or any of the other Transaction Documents can be discharged by one Party, in whole or in part, by a waiver or renunciation of the claim or right unless in writing signed by the other Parties; (b) no waiver that may be given by a Party will be applicable except in the specific instance for which it is given; and (c) no notice to or demand on one Party will be deemed to be a waiver of any obligation of that Party or of the right of the Party giving such notice or demand to take further action without notice or demand as provided in this Agreement or any of the other Transaction Documents.

9.7 Notices. All notices or other communications required or permitted to be given hereunder shall be in writing and shall be deemed given to a Party when (a) delivered by hand, (b) one (1) Business Day after being sent by a nationally recognized overnight courier service (costs prepaid), (c) sent by facsimile or email with confirmation of transmission by the transmitting equipment, or (d) received by the addressee, if sent by certified mail, postage prepaid and return receipt requested, in each case to the following:

if to the Purchaser (or, after the Closing, the Company), to:

Recro Pharma, Inc. 1 E. Uwchaln Ave, Suite 112 Exton, PA 19341 Attention: Ryan Lake Email: ryan.lake@recrocdmo.com

with a copy (which shall not constitute notice) to:

Troutman Pepper Hamilton Sanders LLP 3000 Two Logan Square Philadelphia, PA 19103

Attention: Rachael Bushey and Jen Porter

Email: rachael.bushey@troutman.com; jennifer.porter@troutman.com

if to the Sellers or the Sellers' Representative (or, prior to the Closing, the Company), to:

IriSys, Inc.

Attention: Gerald Yakatan 13813 Boquita Drive Del Mar, CA 92014

Email: gyakatan@sbcglobal.net

with a copy (which shall not constitute notice) to:

Wilson Sonsini Goodrich & Rosati **Professional Corporation** 12235 El Camino Real San Diego, California 92130 Attention: Martin J. Waters and Jason Skolnik

Fax: (858) 350-2399

Email: mwaters@wsgr.com; jskolnik@wsgr.com

Any Party may change its contact information for notices and other communications hereunder by written notice to the other Parties hereto in accordance with this Section 9.7.

9.8 Assignment. This Agreement shall be binding upon, and shall be enforceable by and inure solely to the benefit of the Parties hereto and their respective successors and assigns; provided, that this Agreement and the rights and obligations hereunder shall not be assignable or transferable by any Party without the prior written consent of the other Parties hereto and any assignment or transfer not in compliance with the above shall be null and void, except as set forth in Section 2.4(b) above.

- 9.9 No Third-Party Beneficiaries. Except for contemplated third-party beneficiaries as expressly provided in this Agreement relating to provisions benefiting Indemnified Persons under Article VII herein or as set forth in Section 2.12, this Agreement is for the sole benefit of the Parties hereto and their permitted successors and assigns and nothing herein expressed or implied shall give or be construed to give to any Person, other than the Parties hereto and such successors and assigns, any legal or equitable rights, remedy or claim hereunder.
 - 9.10 Amendments. No amendment to this Agreement shall be effective unless it shall be in writing and signed by the Parties hereto.
- 9.11 Company Disclosure Schedule. The Company Disclosure Schedule shall be subject to the following terms and conditions: (a) the Company Disclosure Schedule are arranged in sections corresponding to the numbered and lettered sections and subsections contained in Article V, and the disclosures in any section or subsection of the Company Disclosure Schedule shall qualify one or more other sections or subsections in Article V, to the extent it is reasonably apparent from a reading of the text of the disclosure that such disclosure is applicable to such other sections and subsections; (b) no disclosure of any matter contained in the Company Disclosure Schedule shall create an implication that such matter meets any standard of materiality (matters reflected in the Company Disclosure Schedule are not necessarily limited to matters required by this Agreement to be reflected in the Company Disclosure Schedule; such additional matters are set forth for informational purposes only and do not necessarily include other matters of a similar nature, nor shall the inclusion of any item be construed as implying that any such item is "material" for any purpose); and (c) headings and introductory language have been inserted on the sections of the Company Disclosure Schedule for convenience of reference only and shall to no extent have the effect of amending or changing the express description of the sections as set forth in this Agreement.
- 9.12 Specific Performance. Each of the Parties acknowledges and agrees that irreparable injury to the other Parties hereto may occur in the event that any provision of this Agreement was not performed in accordance with its specific terms or was otherwise breached and that such injury would not be adequately compensable in damages because of the difficulty of ascertaining the amount of damages that will be suffered in the event that this Agreement was breached. It is accordingly agreed that each of the Parties shall be entitled, in addition to any other remedy to which they are entitled at law or in equity, to specific enforcement of, and injunctive relief, without proof of actual damages, to prevent any violation of the terms hereof, and no Party hereto will take action, directly or indirectly, in opposition to a Party seeking such relief on the grounds that any other remedy or relief is available at law or in equity. Any requirements for the securing or posting of any bond with such remedy are hereby waived.
- 9.13 Construction. In construing this Agreement, including the Exhibits and Schedules hereto, the following principles shall be followed: (a) the terms "herein," "hereof," "hereby," "hereunder" and other similar terms refer to this Agreement as a whole and not only to the particular Article, Section or other subdivision in which any such terms may be employed; (b) except as otherwise set forth herein, references to Articles, Sections, Schedules and Exhibits of this Agreement, which are incorporated in and made a part of this Agreement; (c) a reference to any Person shall include such Person's predecessors; (d) all accounting terms not otherwise defined herein have the meanings assigned to them in accordance with GAAP; (e) no consideration shall be given to the headings of the Articles, Sections, Schedules, Exhibits, subdivisions, subsections or clauses, which are inserted for convenience in locating the provisions of this Agreement and not as an aid in its construction; (f) the word "or" is not exclusive and the word "includes" and "including" and their syntactical variants mean "includes, but is not limited to" and "including, without limitation," and corresponding syntactical variant expressions; (g) a defined term has its defined meaning throughout this Agreement, regardless of whether it appears before or after the place in this Agreement where it is defined, including in any Schedule or Exhibit; (h) the word "dollar" and the symbol "\$"

refer to the lawful currency of the United States of America; (i) the plural shall be deemed to include the singular and vice versa and (j) the phrase "made available" to the Purchaser shall mean "made available at least two (2) Business Days prior to the date hereof".

- 9.14 <u>Further Assurances</u>. Each Party agrees (a) to furnish upon request to each other Party such further information, (b) to execute and deliver to each other Party such other documents and (c) to do such other acts and things, all as another Party may reasonably request for the purpose of carrying out the intent of this Agreement and the other Transaction Documents.
- 9.15 Entire Agreement. This Agreement (including any Exhibit or Schedule attached hereto), the Transaction Documents and the Confidentiality Agreement contain the entire agreement and understanding among the Parties hereto with respect to the subject matter hereof and, except as explicitly set forth herein, supersede all prior and contemporaneous oral and written agreements and understandings relating to such subject matter.
- 9.16 <u>Severability</u>. If any provision of this Agreement or the application of any such provision to any Person or circumstance shall be held invalid, illegal or unenforceable in any respect by a court of competent jurisdiction, such invalidity, illegality or unenforceability shall not affect any other provision hereof.
- 9.17 <u>Mutual Drafting</u>. The Parties hereto are sophisticated and have been represented by counsel who have carefully negotiated the provisions hereof. As a consequence, the Parties do not intend that the presumptions of any Laws or other rules relating to the interpretation of contracts against the drafter of any particular clause should be applied to this Agreement and therefore waive their effects.
- 9.18 <u>Counterparts; Facsimile</u>. This Agreement may be executed in one or more counterparts, including by facsimile or PDF, all of which shall be considered one and the same agreement, and shall become effective when one or more such counterparts have been signed by each of the Parties and delivered to the other Parties.

9.19 Waiver of Conflicts Regarding Representation.

- (a) Wilson Sonsini Goodrich & Rosati, P.C. ("WSGR") has acted as counsel for the Company in connection with the sale of the Company, including the process by which the Company solicited, discussed and negotiated strategic alternatives prior to the date hereof, this Agreement and the other agreements referenced herein or therein and the Transactions (collectively, the "Sale Engagement") and, in that connection, not as counsel for any other Person, including the Purchaser or any of its Affiliates. Only the Company shall be considered a client of WSGR in the Sale Engagement. Notwithstanding anything contained herein to the contrary, if any Seller (or the Sellers' Representative) so desires, WSGR shall be permitted, without the need for any future waiver or consent, to represent any Seller or the Seller's Representative (each, a "Seller Party") after the Closing in connection with any matter related to the matters contemplated by this Agreement or any other agreements referenced herein or therein or any disagreement or dispute relating thereto and may in connection therewith represent the agents or Affiliates of any Seller Party, in any of the foregoing cases including in any dispute, litigation or other adversary proceeding against, with or involving the Purchaser, or any of its agents or Affiliates.
- (b) To the extent that communications between the Company or any Seller Party, on the one hand, and WSGR, on the other hand, relate solely to the Sale Engagement, such communication shall be deemed to be attorney-client confidences that belong solely to the Sellers' Representative, for and on behalf of the Seller Parties. Neither the Purchaser, nor any of its Affiliates, shall have access to any such communications or the files or work product of WSGR, to the extent that they relate solely to the Sale Engagement, whether or not the

Closing occurs. Without limiting the generality of the foregoing, the Purchaser acknowledges and agrees, for itself and on behalf of its Affiliates, upon and after the Closing: (i) the Sellers' Representative, for and on behalf of the Seller Parties, and WSGR shall be the sole holders of the attorney-client privilege with respect to information that relates solely to the Sale Engagement, and neither the Purchaser nor any of its Affiliates, shall be a holder thereof; (ii) to the extent that files or work product of WSGR that relate solely to the Sale Engagement constitute property of the client, only the Sellers' Representative, for and on behalf of the other Seller Parties, shall hold such property rights and have the right to waive or modify such property rights; and (iii) WSGR shall have no duty whatsoever to reveal or disclose any such attorney-client communications, files or work product to the Purchaser or any of its Affiliates, by reason of any attorney-client relationship between WSGR and the Company to the extent relating solely to the Sale Engagement; provided that, to the extent any communication is both related and unrelated to the Sale Engagement, the Sellers' Representative, for and on behalf of the other Seller Parties, shall instruct WSGR to provide copies of such communications, files or work product to the Purchaser or its Affiliates (with only that information that solely relates to the Sale Engagement redacted). Notwithstanding the foregoing, if after the Closing a dispute arises between the Purchaser or one or more of its Subsidiaries, on the one hand, and a third party other than (and unaffiliated with) any Seller Party, on the other hand, then the Purchaser or such Subsidiary (to the extent applicable) may assert the attorney-client privilege to prevent disclosure to such third party of confidential communications of the Current Representation; provided, that neither the Purchaser nor any of its Subsidiaries may waive such privilege without the prior written consent of the Sellers' Representative; provided, further, that no Seller Party will waive attorney-client privilege without the prior written consent of the Purchaser in the event that a dispute or investigation or audit arises after the Closing between the Purchaser or the Company, on the one hand, and a third party, on the other hand.

[SIGNATURES ON NEXT PAGE]

IN WITNESS WHEREOF, the undersigned Parties have executed this Unit Purchase Agreement as of the date first written above.

PURCHASER:

RECRO PHARMA, INC.

By: /s/ J. David Enloe, Jr. Name: J. David Enloe, Jr.

Title: Chief Executive Officer and President

COMPANY:

IRISYS, LLC

By: <u>/s/ Gerald Yakatan</u>
Name: <u>Gerald Yakatan</u>
Title: <u>Chief Executive Officer</u>

SELLERS:

IRISYS, INC.

By: <u>/s/ Gerald Yakatan</u>
Name: <u>Gerald Yakatan</u>
Title: <u>Chief Executive Officer</u>

CONTINENT PHARMACEUTICALS U.S., INC.

By: /s/ Ying Luo Name: Ying Luo Title: CEO GNI Group

EPS AMERICAS CORP.

By: <u>/s/ Kaz Ichinose</u>
Name: <u>Kaz Ichinose</u>
Title: <u>President and CEO</u>

SELLERS' REPRESENTATIVE:

IRISYS, INC.

By: <u>/s/ Gerald Yakatan</u>
Name: <u>Gerald Yakatan</u>
Title: <u>Chief Executive Officer</u>

SELLER NOTE

THIS NOTE (AS DEFINED BELOW) HAS NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED (THE "SECURITIES ACT"), AND MAY NOT BE OFFERED, SOLD, PLEDGED, HYPOTHECATED, ASSIGNED OR TRANSFERRED EXCEPT (I) PURSUANT TO A REGISTRATION STATEMENT UNDER THE SECURITIES ACT WHICH HAS BECOME EFFECTIVE AND IS CURRENT WITH RESPECT TO THESE SECURITIES, OR (II) PURSUANT TO A SPECIFIC EXEMPTION FROM REGISTRATION UNDER THE SECURITIES ACT, BUT ONLY UPON HOLDER HEREOF FIRST HAVING OBTAINED THE WRITTEN OPINION OF COUNSEL TO THE MAKER, OR OTHER COUNSEL, REASONABLY ACCEPTABLE TO THE MAKER, THAT THE PROPOSED DISPOSITION IS CONSISTENT WITH ALL APPLICABLE PROVISIONS OF THE SECURITIES ACT AS WELL AS ANY APPLICABLE "BLUE SKY" OR OTHER SIMILAR SECURITIES LAW.

THIS NOTE AND THE INDEBTEDNESS EVIDENCED BY THIS NOTE, AND ALL RIGHTS OF THE HOLDER HEREOF WITH RESPECT THERETO, ARE SUBORDINATE IN THE MANNER AND TO THE EXTENT SET FORTH IN THAT CERTAIN SUBORDINATION AGREEMENT, DATED AS OF EVEN DATE HEREWITH (THE "SUBORDINATION AGREEMENT"), BY AND AMONG (I) RECRO PHARMA, INC. ("BORROWER"), (II) IRISYS, INC. ("SELLER") AND (III) ATHYRIUM OPPORTUNITIES III ACQUISITION LP, AS ADMINISTRATIVE AGENT UNDER THE SENIOR CREDIT AGREEMENT (AS DEFINED BELOW) (THE "ADMINISTRATIVE AGENT"), TO THE INDEBTEDNESS (INCLUDING INTEREST) OWED BY THE BORROWER AND THE OTHER LOAN PARTIES (AS DEFINED IN THE SENIOR CREDIT AGREEMENT) TO THE ADMINISTRATIVE AGENT AND THE LENDERS (AS DEFINED IN THE SENIOR CREDIT AGREEMENT) PURSUANT TO THAT CERTAIN CREDIT AGREEMENT DATED AS OF NOVEMBER 17, 2017 (AS AMENDED, RESTATED, AMENDED AND RESTATED, SUPPLEMENTED, REFINANCED, RENEWED, REPLACED, IN WHOLE OR PART, OR OTHERWISE MODIFIED FROM TIME TO TIME, THE "SENIOR CREDIT AGREEMENT") AMONG THE BORROWER, THE OTHER LOAN PARTIES PARTY THERETO, THE ADMINISTRATIVE AGENT AND THE LENDERS FROM TIME TO TIME PARTY THERETO; AND THE HOLDER OF THIS INSTRUMENT, BY ITS ACCEPTANCE HEREOF, IRREVOCABLY AGREES TO BE BOUND BY THE PROVISIONS OF THE SUBORDINATION AGREEMENT.

THIS PROMISSORY NOTE MAY NOT BE ASSIGNED OR TRANSFERRED EXCEPT IN ACCORDANCE WITH THE EXPRESS TERMS HEREOF.

RECRO PHARMA, INC.

SUBORDINATED PROMISSORY NOTE

August 13, 2021 (the "Issue Date") \$6,116,672.72

FOR VALUE RECEIVED, Recro Pharma, Inc., a Pennsylvania corporation (the "Borrower"), hereby promises to pay to the order of IriSys, Inc., a California corporation or its successors or permitted assigns ("Seller"), in lawful money of the United States of America, the principal amount of SIX MILLION ONE HUNDRED SIXTEEN THOUSAND SIX HUNDRED SEVENTY-TWO DOLLARS AND SEVENTY-TWO CENTS (\$6,116,672.72), subject to adjustment pursuant to Section 4 below, together with interest from the Issue Date on the unpaid principal balance as set forth below, in accordance with and

subject to the provisions of this subordinated promissory note (as it may be amended from time to time, this "Note").

This Note is being issued pursuant to that certain Unit Purchase Agreement, dated as of August 13 2021, by and among the Borrower, Seller and the other parties thereto (as it may be amended from time to time, the "<u>Purchase Agreement</u>"). Certain capitalized terms used in this Note are defined in Section 6 below.

1. Interest. The outstanding principal amount of this Note shall accrue interest at a rate of 6.0% per annum (calculated on the basis of actual days elapsed in a year of 365 days as may be adjusted below, the "Interest Rate"), due and payable in cash (for the avoidance of doubt, subject to the Subordination Agreement), on each Payment Date. For the avoidance of doubt, subject to the Subordination Agreement, the Borrower shall pay all accrued but unpaid interest on this Note in cash on the Maturity Date. If any amount of principal of this Note is not paid when due (without regard to any applicable grace periods), whether at stated maturity, by acceleration or otherwise, such amount shall thereafter bear interest at 10.0% per annum (the "Default Rate") at all times to the fullest extent permitted by applicable laws. While any Event of Default exists, the Borrower shall pay interest on the principal amount of all outstanding Note Obligations hereunder at a rate per annum at all times equal to the Default Rate to the fullest extent permitted by applicable laws. Accrued and unpaid interest on past due amounts (including interest on past due interest) shall be due and payable by the Borrower upon Lender's demand. Interest on this Note shall be due and payable by the Borrower in arrears on each Payment Date and at such other times as may be specified herein. Interest hereunder shall be due and payable in accordance with the terms hereof before and after judgment, and before and after the commencement of any proceeding under any bankruptcy, insolvency or other debtor relief laws. In the event any interest is paid on this Note which is deemed to be in excess of the then legal maximum rate, then that portion of the interest payment representing an amount in excess of the then legal maximum rate shall be deemed a payment of principal and applied against the principal of this Note.

2. Payment of Principal and Interest; Maturity Date.

(a)Subject to the Subordination Agreement and the other provisions of this Note, the Borrower unconditionally promises to pay Seller all outstanding principal of this Note in three equal installments, one such installment due and payable on each Payment Date, together with interest accrued on this Note; provided, that (i) the final principal repayment installment of this Note shall be repaid on the Maturity Date and in any event shall be in an amount equal to the total principal amount of this Note outstanding on such date and (ii) the principal installment amounts shall be reduced ratably as a result of the application of any prepayments hereof made by the Borrower to Seller

(b)Subject to the Subordination Agreement, the Borrower may, at any time and from time to time, upon at least three Business Days' notice to Lender, voluntarily prepay all or any portion of the outstanding principal of this Note without penalty or premium; *provided*, any prepayment of this Note shall be in a principal amount of \$100,000 or a whole multiple of \$50,000 in excess thereof or, in each case, if less, the entire principal amount of this Note then outstanding.

(c)Any prepayment of this Note shall be accompanied by all accrued interest on the amount prepaid. Each prepayment of this Note pursuant to Section 2(b) shall be applied in the inverse order of maturity to the principal repayment installments hereof.

(d)Upon the occurrence of a Change of Control, the Borrower shall prepay all outstanding principal of, and all accrued and unpaid interest on, this Note.

3.Restrictions on Payment.

(e)<u>Unsecured Nature of Note</u>. This Note and the obligations hereunder shall at all times be unsecured. Any lien on or security interest in any property of the Borrower at any time granted to Seller shall be deemed null and void.

(f)Successors and Assigns; Third Party Beneficiary. The provisions of this Note shall be binding on, and enforceable against, Seller and the Borrower and their respective successors and assigns; provided that (i) the Borrower shall not be entitled to assign its obligations hereunder to any person or entity without the prior written consent of Seller and (ii) the Borrower shall not be entitled to assign its obligations under this Note to any person or entity without the prior written consent of the Administrative Agent.

(g)Acknowledgment; Amendments. Seller acknowledges and agrees that this Note is subject in all respects to the terms of the Subordination Agreement. Until all Senior Debt has been Paid in Full, and notwithstanding anything to the contrary contained in this Note, Seller and Borrower shall not, without the prior written consent of Administrative Agent, agree to any amendment or modification of this Note.

- 4. <u>Purchase Agreement Obligations</u>. At the option of Seller, the amount of any Purchase Agreement Obligation which is finally determined to be payable by Seller pursuant to the Purchase Agreement and unpaid from time to time shall, upon written notice of such election by Seller to the Purchaser, be automatically set off against the amount of the Note Obligations then due and payable hereunder.
- 5.Acceleration and Event of Default. Subject to the Subordination Agreement, the entire unpaid principal amount of this Note, all accrued and unpaid interest thereon and any other amounts due hereunder shall automatically accelerate and shall be due and payable by the Borrower, without presentment, demand, protest or notice of any kind, each of which is hereby expressly waived, on the date of occurrence of an Event of Default. Upon the occurrence of an Event of Default, Seller shall have available, in addition to all remedies available hereunder, those remedies available to a creditor of the Borrower or otherwise permitted to it by law, either by suit in equity or by action at law, or both.

6. Definitions. For purposes of this Note, the following capitalized terms have the following meanings.

2"Change of Control" has the meaning set forth in the Senior Credit Agreement, mutatis mutandis.

1."Event of Default" shall mean if (i) Borrower becomes insolvent or becomes unable or admits in writing its inability or fails generally to pay its debts as they become due; (ii) Borrower applies for, consents to, or acquiesces in the appointment of a trustee, receiver or other custodian for Borrower or any property thereof, or makes a general assignment for the benefit of creditors or, in the absence of such application, consent or acquiescence, a trustee, receiver or other custodian is appointed for such Borrower or for a substantial part of the property thereof and such appointment continues undischarged or unstayed for 60 calendar days; (iii) any bankruptcy, reorganization, debt arrangement, or other case or proceeding under any bankruptcy or insolvency law, or any dissolution or liquidation proceeding, is commenced in respect of Borrower, and if such case or proceeding is not commenced by Borrower, it is consented to or acquiesced in by Borrower, or remains for 60 days undismissed or unstayed; (iv) Borrower takes any action to authorize, or in furtherance of, any of the foregoing; (v) Borrower fails to pay when and as required to be paid herein, (A) any Note Obligations due on the Maturity Date or (B) any amount of principal of this Note; or (vi) Borrower fails to pay within fifteen (15) days after the same becomes due, any interest on this Note, or any other amount (except for amounts described in clause (v) of this definition) due under this Note.

2."<u>Maturity Date</u>" means the earliest to occur of (a) the third anniversary of the Issue Date and (b) the date on which payment of this Note is accelerated pursuant to <u>Section 5</u> hereof; *provided*, if such anniversary is not a Business Day, the Maturity Date shall be the immediately preceding Business Day.

"Note Obligations" means all advances to, and debts, liabilities, obligations, covenants and duties of, the Borrower arising under this Note, whether direct or indirect (including those acquired by assumption), absolute or contingent, due or to become due, now existing or hereafter arising and including interest and fees that accrue after the commencement by or against the Borrower or any affiliate thereof of any proceeding under any debtor relief laws naming such person or entity as the debtor in such proceeding, regardless of whether such interest and fees are allowed or allowable claims in such proceeding. Without limiting the foregoing, the Note Obligations include the obligation to pay principal, interest, expenses, fees, indemnities and other amounts payable by the Borrower under this Note and the Subordination Agreement.

"Paid in Full" or "Payment in Full" shall mean the payment in full in cash of all Senior Debt, and the termination of all letters of credit and all commitments to lend under the Senior Debt Documents.

"Payment Date" means each of the first and second anniversaries of the Issue Date, and the Maturity Date.

"Purchase Agreement Obligation" means (i) an indemnification obligation of Seller to any Purchaser Indemnified Party (as defined in the Purchase Agreement) arising under Section 7.2 of the Purchase Agreement and which may be recovered directly from Seller pursuant to the provisions of Article VII of the Purchase Agreement or (ii) any payment obligation of Seller pursuant to Section 2.7(g) of the Purchase Agreement.

"Senior Debt" shall mean, collectively, all Loan Documents Obligations (as defined in the Senior Credit Agreement) and any indebtedness incurred to refinance, increase, renew, supplement or replace the same.

"Senior Debt Documents" shall mean the Senior Credit Agreement and all other Loan Documents (as defined in the Senior Credit Agreement).

7. Amendment and Waiver. The provisions of this Note may be amended or waived only in a writing signed by the Borrower and Seller, and consented to in writing by the Administrative Agent.

8. Notices. All notices, requests, claims, demands and other communications under this Note will be in writing and will be deemed given if delivered personally, sent by overnight courier (providing proof of delivery) or when received via email (with confirmation of delivery) to the parties at the following addresses (or at such other address for a party as specified by like notice):

if to Borrower:

Recro Pharma, Inc. 1 E. Uwchlan Ave, Suite 112 Exton, PA 19341 Attention: Ryan Lake

Email: ryan.lake@recrocdmo.com

with a copy (which shall not constitute notice) to:

Troutman Pepper Hamilton Sanders LLP 3000 Two Logan Square Philadelphia, PA 19103 Attention: Rachael Bushey, Esq. and Jennifer Porter, Esq.

Email: rachael.bushey@troutman.com; jennifer.porter@troutman.com

If to Seller:

IriSys, Inc.

Attention: Gerald Yakatan Email: gyakatan@sbcglobal.net

with a copy (which shall not constitute notice) to:

Wilson Sonsini Goodrich & Rosati Professional Corporation 12235 El Camino Real San Diego, California 92130 Attention: Martin J. Waters and Jason Skolnik

Fax: (858) 350-2399

Email: mwaters@wsgr.com; jskolnik@wsgr.com

9. Governing Law. This Agreement shall be governed by and construed in accordance with the law of the State of Delaware, without regard to the conflicts of laws rules of such state.

10. Execution and Delivery. Facsimile, email or other electronic transmission of signatures to this Note shall be acceptable and binding.

- 11. Expenses; Waivers. If action is instituted to collect this Note, the Borrower promises to pay all costs and expenses, including, without limitation, reasonable attorneys' fees and costs, incurred in connection with such action. The Borrower hereby waives notice of default, presentment or demand for payment, protest or notice of nonpayment or dishonor and all other notices or demands relative to this instrument.
- 12. WAIVER OF JURY TRIAL. BY ACCEPTANCE OF THIS NOTE, LENDER HEREBY AGREES AND THE BORROWER HEREBY AGREES TO WAIVE THEIR RESPECTIVE RIGHTS TO A JURY TRIAL OF ANY CLAIM OR CAUSE OF ACTION BASED UPON OR ARISING OUT OF THIS NOTE OR THE SUBORDINATION AGREEMENT.

[Signature page follows.]

IN WITNESS WHEREOF, Borrower has executed and delivered this Note on the date first above written.		
	BORROWER:	
	RECRO PHARMA, INC.	
	By: Name: Its:	

SIXTH AMENDMENT TO CREDIT AGREEMENT

THIS SIXTH AMENDMENT TO CREDIT AGREEMENT (this "<u>Agreement</u>"), dated as of August 13, 2021 (the "<u>Sixth Amendment Effective Date</u>"), is entered into among RECRO PHARMA, INC., a Pennsylvania corporation (the "<u>Borrower</u>"), the Guarantors party hereto, the Lenders party hereto and ATHYRIUM OPPORTUNITIES III ACQUISITION LP, as Administrative Agent (the "<u>Administrative Agent</u>"). All capitalized terms used herein and not otherwise defined herein shall have the meanings given to such terms in the Credit Agreement (as defined below).

RECITALS

WHEREAS, the Borrower, the Guarantors party thereto, the Lenders from time to time party thereto and the Administrative Agent have entered into that certain Credit Agreement, dated as of November 17, 2017 (as amended, restated, supplemented or modified from time to time prior to the date hereof, the "<u>Credit Agreement</u>");

WHEREAS, the Borrower has requested that the Lenders amend the Credit Agreement to provide for certain modifications of the terms as set forth below; and

WHEREAS, the Lenders and the Administrative Agent are willing to amend the Credit Agreement referred to herein, in each case, subject to the terms and conditions hereof.

NOW, THEREFORE, in consideration of the premises and the mutual covenants contained herein, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto agree as follows:

1. Amendments to Credit Agreement.

a. The following definitions are added to Section 1.01 of the Credit Agreement in the appropriate alphabetical order to read as follows:

"IRISYS Acquisition" means the Acquisition by the Borrower of all of the Equity Interests of IRISYS, LLC, a California limited liability company, on the Sixth Amendment Effective Date, pursuant to that certain Unit Purchase Agreement, dated as of the Sixth Amendment Effective Date, by and among the Borrower, as the purchaser, IRISYS, LLC, as the target, and IRISYS, INC., Continent Pharmaceuticals U.S., Inc., and EPS Americas Corp., collectively, as the sellers.

"IRISYS Seller Note" means that certain Subordinated Promissory Note, dated as of the Sixth Amendment Effective Date, by and among the Borrower and IRISYS, INC., in an original principal amount of \$6,116,672.72.

"Sixth Amendment" means that certain Sixth Amendment to Credit Agreement, dated as of the Sixth Amendment Effective Date, by and among the Loan Parties, the Lenders and the Administrative Agent.

"Sixth Amendment Effective Date" means August 13, 2021.

"Sixth Amendment Fee Letter" means that certain letter agreement dated as of the Sixth Amendment Effective Date by and between the Borrower and the Administrative Agent.

b.The definition of "Agreement" in Section 1.01 of the Credit Agreement is amended and restated in its entirety to read as follows:

"Agreement" means this Credit Agreement, as amended or otherwise modified from time to time (including as amended by the First Amendment, the Second Amendment, the Third Amendment, the Fourth Amendment, the Fifth Amendment and the Sixth Amendment).

c.The definition of "Loan Documents" in Section 1.01 of the Credit Agreement is amended and restated in its entirety to read as follows:

"Loan Documents" means this Agreement, the First Amendment, the Second Amendment, the Third Amendment, the Fourth Amendment, the Fifth Amendment, the Sixth Amendment, each Note, the Fee Letter, the First Amendment Fee Letter, the Fourth Amendment Fee Letter, the Disclosure Letter, the Second Amendment Disclosure Letter, the Third Amendment Disclosure Letter, each Joinder Agreement, each Collateral Document and any other agreement, instrument or document designated by its terms as a "Loan Document", excluding, for the avoidance of doubt, the Warrants.

d.The definition of "Maturity Date" in Section 1.01 of the Credit Agreement is amended and restated in its entirety to read as follows:

"Maturity Date" means December 31, 2023; provided, that, if such date is not a Business Day, the Maturity Date shall be the first Business Day immediately preceding such date.

e.The definition of "Permitted Acquisition" in Section 1.01 of the Credit Agreement is amended and restated in its entirety to read as follows:

"Permitted Acquisition" means (a) the IRISYS Acquisition, and (b) an Investment consisting of an Acquisition by a Loan Party; provided, that, (i) the property acquired (or the property of the Person acquired) in such Acquisition is used or useful in the same or a related line of business as the Borrower and its Subsidiaries were engaged in on the Closing Date (or any reasonable extensions or expansions thereof), (ii) no Default or Event of Default shall have occurred and be continuing or would result from such Acquisition, (iii) the Administrative Agent shall have received all items in respect of the Equity Interests or property acquired in such Acquisition as and when required to be delivered by the terms of Section 7.12 and/or Section 7.14, (iv) such Acquisition shall not be a "hostile" acquisition and shall have been approved by the Board of Directors and/or the shareholders (or equivalent) of the applicable Loan Party and the target of such Acquisition, (v) the Borrower shall have delivered to the Administrative Agent pro forma financial statements for the Borrower and its Subsidiaries after giving effect to such Acquisition for the twelve month period ending as of the most recent fiscal quarter end in a form reasonably satisfactory to the Administrative Agent, (vi) the Borrower shall have demonstrated to the reasonable satisfaction of the Administrative Agent that, after giving effect to such Acquisition on a Pro Forma Basis, the Loan Parties are in compliance with the covenants set forth in Section 8.16, (vii) the representations and warranties made by the Loan Parties in each Loan Document shall be true and correct in all material respects (and in all respects if any such representation or warranty is already qualified by materiality or reference to

Material Adverse Effect) at and as if made as of the date of such Acquisition (assuming for such purposes that such Acquisition has been consummated), except to the extent any such representation and warranty expressly relates to an earlier date, in which case it shall be true and correct in all material respects (and in all respects if any such representation or warranty is already qualified by materiality or reference to Material Adverse Effect) as of such earlier date and (viii) the aggregate consideration (including any Earn Out Obligations) paid (or payable, as the case may be) in cash by the Borrower and its Subsidiaries shall not exceed an aggregate amount equal to \$40,000,000 for all such Acquisitions made in reliance on this clause (b) during the term of this Agreement.

f.Section 8.03(1) of the Credit Agreement is amended and restated in its entirety to read as follows:

(l) (i) Indebtedness constituting Earn Out Obligations or obligations in respect of working capital adjustment requirements under the agreements used to consummate a Permitted Acquisition or other Investment permitted under <u>Section 8.02</u> (other than by reference to this <u>Section 8.03</u> (or any sub-clause hereof)), and (ii) Indebtedness incurred pursuant to the IRISYS Seller Note, so long as (A) the aggregate principal amount of such Indebtedness does not exceed \$6,116,672.72, (B) such Indebtedness is unsecured and does not mature prior to the ninety-first (91st) day after the Maturity Date, (C) such Indebtedness is subordinated to the Obligations at all times on terms and conditions that are reasonably satisfactory to the Administrative Agent on the Sixth Amendment Effective Date;

2. Conditions Precedent to Effectiveness. This Agreement shall be effective upon satisfaction of the following conditions precedent:

(a)receipt by the Administrative Agent of counterparts of (i) this Agreement duly executed by the Borrower, the Guarantors, the Lenders and the Administrative Agent and (ii) the Sixth Amendment Fee Letter duly executed by the Borrower and the Administrative Agent;

(b)receipt by the Administrative Agent of a certificate of a Responsible Officer of the Borrower (i) certifying that the IRISYS Acquisition has been consummated (or shall be consummated contemporaneously with the closing of this Agreement) in accordance with the terms of that certain Unit Purchase Agreement, dated as of the Sixth Amendment Effective Date, by and among the Borrower, as the purchaser, IRISYS, LLC, as the target, and IRISYS, INC., Continent Pharmaceuticals U.S., Inc., and EPS Americas Corp., collectively, as the sellers, a true correct copy of which is attached to such certificate together with all other material agreements entered into in connection therewith (including any schedules and exhibits thereto), and (ii) attaching a true, correct and complete copy of the IRISYS Seller Note:

(c)receipt by the Administrative Agent of favorable opinions of legal counsel to the Loan Parties, addressed to the Administrative Agent and each Lender, dated as of the Sixth Amendment Effective Date, and in form and substance reasonably satisfactory to the Administrative Agent and the Lenders; and

(d)receipt by the Administrative Agent of all reasonable and documented out-of-pocket fees, charges and disbursements of counsel to the Administrative Agent and all reasonable and documented out-of-pocket due diligence expenses of the Administrative Agent and the Lenders, in each case, incurred in connection with this Agreement and the transactions contemplated hereby and for which invoices have been issued (provided, that, the issuance of such invoices shall not

thereafter preclude a final settling of accounts between the Borrower and the Administrative Agent).

- 3. <u>Reaffirmation</u>. Each Loan Party hereby acknowledges and agrees that, as of the Sixth Amendment Effective Date, after giving effect to this Agreement and the transactions contemplated hereby, the aggregate principal amount of Loans outstanding under the Credit Agreement is \$100,000,000. Each of the Loan Parties acknowledges and reaffirms (a) that it is bound by all of the terms of the Investment Documents to which it is a party and (b) that it is responsible for the observance and full performance of all Obligations, including without limitation, the repayment of the Loans. Furthermore, the Loan Parties acknowledge and confirm (i) that the Lenders have performed fully all of their obligations under the Credit Agreement and the other Investment Documents arising on or before the date hereof other than their respective obligations specifically set forth in this Agreement and (ii) that by entering into this Agreement, the Lenders do not, except as expressly set forth herein, waive or release any term or condition of the Credit Agreement or any of the other Investment Documents or any of their rights or remedies under such Investment Documents or any applicable law or any of the Obligations of the Loan Parties thereunder.
- 4. <u>Release</u>. As a material part of the consideration for the Administrative Agent and the Lenders entering into this Agreement, the Loan Parties agree as follows (the "<u>Release Provision</u>"):
 - a) The Administrative Agent, the Lenders, each of their respective Affiliates and each of the foregoing Persons' respective officers, managers, members, directors, advisors, sub-advisors, partners, agents and employees, and their respective successors and assigns (hereinafter all of the above collectively referred to as the "Lender Group"), are irrevocably and unconditionally released, discharged and acquitted from any and all actions, causes of action, claims, demands, damages and liabilities of whatever kind or nature, in law or in equity, now known or unknown, suspected or unsuspected to the extent that any of the foregoing arises from any action or failure to act under or otherwise arising in connection with the Investment Documents, in each case arising on or prior to the Sixth Amendment Effective Date, except to the extent such actions, causes of action, claims, demands, damages and liabilities result from the gross negligence or willful misconduct of any of the Lender Group as determined by a court of competent jurisdiction in a final and nonappealable judgment; provided, however, that, the Loan Parties do not release, discharge or acquit the Lender Group from their respective obligations specifically set forth in this Agreement.
 - b) Each Loan Party hereby acknowledges, represents and warrants to the Lender Group that:
 - i) it has read and understands the effect of the Release Provision. Each Loan Party has had the assistance of independent counsel of its own choice, or has had the opportunity to retain such independent counsel, in reviewing, discussing, and considering all the terms of the Release Provision; and if counsel was retained, counsel for such Loan Party has read and considered the Release Provision and advised such Loan Party with respect to the same. Before execution of this Agreement, such Loan Party has had adequate opportunity to make whatever investigation or inquiry it may deem necessary or desirable in connection with the subject matter of the Release Provision.
 - ii) no Loan Party is acting in reliance on any representation, understanding, or agreement not expressly set forth herein or in the Credit Agreement or other Investment Documents. Each Loan Party acknowledges that the Lender Group has not made any representation with respect to the Release Provision except as expressly set forth herein.

- iii) each Loan Party has executed this Agreement and the Release Provision thereof as its free and voluntary act, without any duress, coercion, or undue influence exerted by or on behalf of any person.
- iv) the Loan Parties are the sole owners of the claims released by the Release Provision, and no Loan Party has heretofore conveyed or assigned any interest in any such claims to any other Person.
- c) Each Loan Party understands that the Release Provision was a material consideration in the agreement of the Administrative Agent and the Lenders to enter into this Agreement. The Release Provision shall be in addition to any rights, privileges and immunities granted to the Administrative Agent and the Lenders under the Investment Documents.

5. Miscellaneous.

- a) The Credit Agreement and the Obligations of the Loan Parties thereunder and under the other Investment Documents, are hereby ratified and confirmed and shall remain in full force and effect according to their terms, as amended by this Agreement. This Agreement is a Loan Document.
- b) Each Guarantor (i) acknowledges and consents to all of the terms and conditions of this Agreement, (ii) affirms all of its Obligations under the Investment Documents, and (iii) agrees that this Agreement and all documents executed in connection herewith do not operate to reduce or discharge its Obligations under the Credit Agreement or the other Investment Documents.
 - c) The Loan Parties represent and warrant to the Administrative Agent and the Lenders that:
 - i) each Loan Party has taken all necessary corporate, limited liability company or other organizational action to authorize the execution, delivery and performance of this Agreement.
 - ii) this Agreement has been duly executed and delivered by each Loan Party and constitutes a legal, valid and binding obligation of each Loan Party, enforceable against each such Loan Party in accordance with its terms, subject to bankruptcy, insolvency and similar laws affecting enforceability of creditors' rights generally and to general principles of equity.
 - iii) no approval, consent, exemption, authorization or other action by, or notice to, or filing with, any Governmental Authority or any other Person is necessary or required in connection with the execution, delivery or performance by, or enforcement against, any Loan Party of this Agreement other than (A) those that have already been obtained and are in full force and effect and (B) those that may be required under any applicable notices under securities laws.
 - (iv) (A) the representations and warranties of the Borrower and each other Loan Party contained in Article VI of the Credit Agreement or any other Investment Document, or which are contained in any document furnished at any time under or in connection therewith, are true and correct in all material respects (and in all respects if any such representation and warranty is already qualified by materiality or reference to Material Adverse Effect) on and as of the date hereof, except to the extent that such representations

and warranties specifically refer to an earlier date, in which case they are true and correct in all material respects (and in all respects if any such representation and warranty is already qualified by materiality or reference to Material Adverse Effect) as of such earlier date and (B) no event has occurred and is continuing which constitutes a Default or an Event of Default.

- (d) Each of the Loan Parties hereby affirms the Liens created and granted in the Loan Documents in favor of the Administrative Agent, for the benefit of the Secured Parties, and agrees that this Agreement does not adversely affect or impair such Liens and security interests in any manner.
 - (e) This Agreement may be executed in counterparts (and by different parties hereto in different counterparts), each of which shall constitute an original, but all of which when taken together shall constitute a single contract. Delivery of an executed counterpart of a signature page of this Agreement by facsimile or other electronic imaging means (e.g. "pdf" or "tif") shall be effective as delivery of a manually executed counterpart of this Agreement.
 - (f) If any provision of this Agreement is held to be illegal, invalid or unenforceable, (i) the legality, validity and enforceability of the remaining provisions of this Agreement shall not be affected or impaired thereby and (ii) the parties shall endeavor in good faith negotiations to replace the illegal, invalid or unenforceable provisions with valid provisions the economic effect of which comes as close as possible to that of the illegal, invalid or unenforceable provisions. The invalidity of a provision in a particular jurisdiction shall not invalidate or render unenforceable such provision in any other jurisdiction.
 - (g) THIS AGREEMENT AND ANY CLAIM, CONTROVERSY, DISPUTE OR CAUSE OF ACTION (WHETHER IN CONTRACT OR TORT OR OTHERWISE) BASED UPON, ARISING OUT OF OR RELATING TO THIS AGREEMENT AND THE TRANSACTIONS CONTEMPLATED HEREBY, SHALL BE GOVERNED BY, AND CONSTRUED IN ACCORDANCE WITH, THE LAWS OF THE STATE OF NEW YORK.

[SIGNATURE PAGES FOLLOW]

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be duly executed as of the date first above written.

PHARMA, INC.,

a Pennsylvania corporation

By:/s/ Ryan Lake Name: Ryan Lake

Title: Chief Financial Officer

O GAINESVILLE LLC,

a Massachusetts limited liability company

By:/s/ Ryan Lake Name: Ryan Lake

Title: Treasurer

RECRO GAINESVILLE DEVELOPMENT LLC,

a Delaware limited liability company

By:/s/ Ryan Lake Name: Ryan Lake

Title: Secretary and Treasurer

GENT: ATHYRIUM OPPORTUNITIES III ACQUISITION LP,

a Delaware limited partnership

By: Athyrium Opportunities Associates III LP, its General Partner

By: Athyrium Opportunities Associates III GP LLC, the General Partner of Athyrium Opportunities Associates III LP

By:/s/ Rashida Adams Name: Rashida Adams Title: Authorized Signatory M OPPORTUNITIES III ACQUISITION LP,

a Delaware limited partnership

By: Athyrium Opportunities Associates III LP, its General Partner

By: Athyrium Opportunities Associates III GP LLC, the General Partner of Athyrium Opportunities Associates III LP

By:/s/ Rashida Adams
Name: Rashida Adams
Title: Authorized Signatory

JNITIES II ACQUISITION LP,

a Delaware limited partnership

By: Athyrium Opportunities Associates II LP, its General Partner

By: Athyrium GP Holdings LLC, the General Partner of Athyrium Opportunities Associates II LP

By:/s/ Rashida Adams
Name: Rashida Adams
Title: Authorized Signatory





Contacts:

Stephanie Diaz (Investors) Tim Brons (Media) Ryan D. Lake (CFO) Vida Strategic Partners Vida Strategic Partners Recro 415-675-7401 415-675-7402 770-531-8365 sdiaz@ vidasp.com tbrons@ vidasp.com ryan.lake@ recroCDMO.com

RECRO ANNOUNCES ENTRY INTO LETTER OF INTENT FOR ACQUISITION OF A FULL SERVICE CDMO

EXTON, PA, August 10, 2021— Recro Pharma, Inc. ("Recro"; NASDAQREPH), a contract development and manufacturing organization (CDMO) dedicated to solving complex formulation and manufacturing challenges for companies developing oral solid dose drug products, today announced that it has signed a non-binding letter of intent ("the LOI") for its acquisition of a full service CDMO with capabilities that range from formulation development to commercial manufacturing for various dosage forms including oral liquids, sterile injectables, tablets, topicals, liquid/powder filled capsules, ophthalmic droppers, liposomes and nano/microparticles (the "Target").

Pursuant to the LOI, Recro agreed to acquire 100% of the equity interests of the Target in exchange for consideration having an aggregate value of approximately \$49.850 million, assuming a cash free, debt free balance sheet and that the Target has, at closing, an adequate level of working capital (excluding cash) to support the operations of its business post-closing. The purchase price will be paid through: (i) \$25.5 million of cash at closing; (ii) 9,302,718 shares of common stock of Recro to be issued to Target in six months; and (iii) seller notes of \$6.1 million. The seller notes will have a three (3) year maturity date from the date of closing and bear interest at a rate of 6% annually. The seller notes will be expressly subordinated and unsecured in right of payment and priority to Recro's existing debt with Athyrium Capital Management.

"We are excited about this potential transaction and look forward to closing and sharing more details in the near term," said David Enloe, president and chief executive officer of Recro.

Completion of the transaction is subject to finalization of due diligence investigations by the relevant parties, the negotiation and execution of definitive transaction agreements and other customary closing conditions. There can be no assurance that a definitive agreement will be entered into or that the proposed transaction will be consummated. Further, readers are cautioned that those portions of the LOI that describe the proposed transaction, including the consideration to be issued therein, are non-binding.

About Recro

Recro (NASDAQ:REPH) is a contract development and manufacturing organization (CDMO) with capabilities from early feasibility to commercial manufacturing. With an expertise in solving complex manufacturing problems, Recro is a CDMO providing oral solid dosage form development, end-to-end regulatory support, clinical and commercial manufacturing, and packaging and logistics services to the global pharmaceutical market.

In addition to our experience in handling DEA controlled substances and developing and manufacturing modified release oral solid dosage forms, Recro has the expertise to deliver on our clients' pharmaceutical development and manufacturing projects, regardless of complexity level. We do all of this in our best-in-class facilities, which total 120,000 square feet, in Gainesville, Georgia.

For more information about Recro's CDMO solutions, visit recrocdmo.com.

Forward-Looking Statements

This press release includes forward-looking statements within the meaning of Section 27A of the Securities Act of 1933 and Section 21E of the Securities Exchange Act of 1934. These statements, among other things, the Company's expectations regarding the potential acquisition of the Target, the potential benefits of the acquisition of the Target, the proposed terms and conditions of a definitive agreement with the Target and the receipt of all necessary approvals and satisfaction of all closing conditions for completion of the transaction, and other statements. The words "anticipate", "believe", "could", "estimate", "upcoming", "expect", "intend", "may", "plan", "predict", "project", "will" and similar terms and phrases may be used to identify forward-looking statements in this press release. Factors that could cause the company's actual outcomes to differ materially from those expressed in or underlying these forward-looking statements include risks that the proposed transaction may not occur on the terms described in this press release or at all; if the proposed transaction completes, the results and potential of the combined business and the combination of Target's business with the Company's business may not be as anticipated; the ongoing economic and social consequences of the COVID-19 pandemic, including any adverse impact on the customer ordering patterns or inventory rebalancing or disruption in raw materials or supply chain; demand for the company's services, which depends in part on customers' research and development and the clinical plans and market success of their products; customers' changing inventory requirements and manufacturing plans; customers and prospective customers decisions to move forward with the company's manufacturing services; the average profitability, or mix, of the products the company manufactures; the company's ability to enhance existing or introduce new services in a timely manner; fluctuations in the costs, availability, and suitability of the components of the products the company manufactures, including active pharmaceutical ingredients, excipients, purchased components and raw materials, or the company's customers facing increasing or new competition. These forward-looking statements should be considered together with the risks and uncertainties that may affect our business and future results presented herein along with those risks and uncertainties discussed in our filings with the Securities and Exchange Commission at www.sec.gov. These forward-looking statements are based on information currently available to us, and we assume no obligation to update any forward-looking statements except as required by applicable law.



Exhibit 99.2

Contacts:

Stephanie Diaz (Investors) Tim Brons (Media) Ryan D. Lake (CFO) Vida Strategic Partners Vida Strategic Partners Recro 415-675-7401 415-675-7402 770-531-8365 sdiaz@ vidasp.com tbrons@ vidasp.com ryan.lake@ recroCDMO.com

RECRO ANNOUNCES ACQUISITION OF SAN DIEGO-BASED IRISYS, CREATING BI-COASTAL, FULL SERVICE CDMO

Combination of Synergistic Organizations Results in Broad CDMO Service with Capabilities Spanning from Pre-IND Development to Commercial Manufacturing and Packaging for Wide Range of Dosage Forms

Transaction Expands Recro's Global Customer Base, Offering Pipeline and Revenue Diversification Along with Significant Opportunity for Continued

Growth

EXTON, PA, August 13, 2021— Recro Pharma, Inc. ("Recro"; NASDAQREPH), a contract development and manufacturing organization (CDMO) dedicated to solving complex formulation and manufacturing challenges for companies developing oral solid dose drug products, today announced its acquisition of IRISYS, a San Diego-based CDMO that possesses capabilities that complement and expand those of RecroRecro acquired IRISYS for approximately \$50 million in a combination of cash, shares of Recro common stock and a seller promissory note. With its acquisition of IRISYS, Recro has transformed itself into a full service CDMO with operations on both the East and West Coast of the U.S. capable of offering its now expanded global client base access to services spanning from pre-Investigational New Drug (IND) development to commercial manufacturing and packaging for wide range of dosage forms.

"Today's acquisition of IRISYS is truly a transformative transaction for Recro.The joining of these two companies creates an organization that is ideally positioned to achieve Recro's previously disclosed four-pronged strategy for growth, which includes strengthening the organization's leadership and talent, expanding and diversifying our client base, fortifying our financial position, and enhancing our capabilities and competencies," said David Enloe, president and chief executive officer of Recro. "Adding the personnel, facilities, capabilities and global customer base of IRISYS advances each of these goals and positions Recro for sustainable, profitable growth moving forward. We welcome the members of the IRISYS team to the Recro family and are eager to begin implementing the integration of the companies."

IRISYS is a full service CDMO based in San Diego, CA with capabilities that range from formulation development to commercial manufacturing for various dosage forms including oral liquids, sterile injectables, tablets, topicals, liquid/powder filled capsules, ophthalmic droppers, liposomes and nano/microparticles. IRISYS serves a diverse, global client base within the biopharmaceutical industry with its nearly 40 clients located in six different countries around the world. This includes manufacturing support for four commercial and near-commercial products. Importantly, IRISYS recently added automated, aseptic fill/finish and lyophilization capabilities to its existing facility, offering a new key avenue for significant near-term growth. IRISYS expects to generate \$15.4 million in revenue for full year 2021 and has approximately \$23 million spanning its contracted backlog and vetted sales pipeline.

Strategic Rationale

Recro's acquisition of IRISYS was driven by a number of compelling factors covering the four prongs of the company's ongoing growth strategy.

Enhances Capabilities and Competencies:

- © Expands technical focus beyond oral solid dose to include sterile injectables oral liquids, tablets, topicals, liquid/powder filled capsules, ophthalmic droppers, liposomes and nano/microparticles;
- ② Adds new capabilities in the areas of aseptic fill/finish and lyophilization; and
- Establishes bi-coastal footprint from which to better serve clients within the U.S., as well as globally.

Expands and Diversifies Client Base:

- Adds more than three dozen clients spanning six countries, reducing Recro's customer dependency risk;
- ① Provides greater balance between our clinical and commercial project mix; and
- ① Diversifies portfolio from exclusively oral solid dose to a balance of multiple advanced dosage forms and oral solid dose.

Strengthens Leadership and Talent:

- Supplements Recro's seasoned leadership team with strong scientific talent credited with driving success at IRISYS; and
- ② Provides access to West Coast human capital hotbed.

Fortifies Financial Position:

- ① Adds profitable business with IRISYS full year 2021 estimated revenues of \$15.4 million, net income of \$2.1 million and EBITDA* of \$2.8 million; the corresponding anticipated improvement in Recro's EBITDA will be beneficial to Recro's debt covenants under its credit agreement. Recro intends to provide updated financial guidance when it releases third quarter 2021 earnings;
- Offers significant revenue diversification, reducing financial reliance on any particular client or service offering;
- Decomes accretive by the fourth quarter of 2021, excluding integration costs; and
- ① Enables transaction-related paydown of credit facility and an extension of the due date, leading to improved cash flow.

In addition to the advancement of the company's four-pronged growth strategy, the acquisition was also driven by key synergies between Recro and IRISYS within business development, clinical development and commercial scale-up, as well as a strong cultural alignment and fit between the companies.

*EBITDA is a non-GAAP financial measure (see Non-GAAP Financial Measures in this release).

Transaction Details

Under terms of the agreement, Recro has acquired 100% of the equity interests of IRISYS in exchange for consideration having an aggregate value of approximately \$49.850 million. The purchase price was paid

through: (i) \$25.5 million of cash at closing; (ii) 9,302,718 shares of common stock of Recro to be issued in six months; and (iii) a seller promissory note of \$6.1 million. The seller note has a three (3) year maturity date from the date of closing and bears interest at a rate of 6% annually. The seller note is expressly subordinated and unsecured in right of payment and priority to Recro's existing debt with Athyrium Capital Management.

Bailey Southwell & Co. served as the exclusive financial advisor to IRISYS. William Blair & Company, L.L.C. represented Recro on the transaction.

Non-GAAP Financial Measures

Certain financial measures of IRISYS presented in this press release, including EBITDA, were not calculated using U.S. generally accepted accounting principles ("GAAP"). We believe these financial measures are helpful in understanding IRISYS' business because it allows for greater transparency of supplemental information used by Recro's management in connection with the acquisition of IRISYS. These measures are used by investors, as well as Recro's management, in assessing IRISYS' past and potential future performance. Non-GAAP financial measures should be considered in addition to, but not as a substitute for, reported GAAP results. Further, Non-GAAP financial measures, even if similarly titled, may not be calculated in the same manner by all companies, and therefore should not be compared. Please see the section of this press release titled "IRISYS, LLC Reconciliation of GAAP to Non-GAAP Financial Measures" for a reconciliation of Non-GAAP EBITDA, to its most directly comparable GAAP measure.

About Recro

Recro (NASDAQ:REPH) is a contract development and manufacturing organization (CDMO) with capabilities from early feasibility to commercial manufacturing. With an expertise in solving complex manufacturing problems, Recro is a CDMO providing oral solid dosage form development, end-to-end regulatory support, clinical and commercial manufacturing, and packaging and logistics services to the global pharmaceutical market.

In addition to our experience in handling DEA controlled substances and developing and manufacturing modified release oral solid dosage forms, Recro has the expertise to deliver on our clients' pharmaceutical development and manufacturing projects, regardless of complexity level. We do all of this in our best-in-class facilities, which total 120,000 square feet, in Gainesville, Georgia.

For more information about Recro's CDMO solutions, visit recrocdmo.com.

Forward-Looking Statements

This press release includes forward-looking statements within the meaning of Section 27A of the Securities Act of 1933 and Section 21E of the Securities Exchange Act of 1934. These statements, among other things, the Company's expectations regarding, the potential benefits of the acquisition of IRISYS, and other statements. The words "anticipate", "believe", "could", "estimate", "upcoming", "expect", "intend", "may", "plan", "predict", "project", "will" and similar terms and phrases may be used to identify forward-looking statements in this press release. Factors that could cause the company's actual outcomes to differ materially from those expressed in or underlying these forward-looking statements include risks that the results and potential of the combined business and the combination IRISYS' business with the Company's business may not be as anticipated; the ongoing economic and social consequences of the COVID-19 pandemic, including any adverse impact on the customer ordering patterns or inventory rebalancing or disruption in raw materials or supply chain; demand for the company's services, which

depends in part on customers' research and development and the clinical plans and market success of their products; customers' changing inventory requirements and manufacturing plans; customers and prospective customers decisions to move forward with the company's manufacturing services; the average profitability, or mix, of the products the company manufactures; the company's ability to enhance existing or introduce new services in a timely manner; fluctuations in the costs, availability, and suitability of the components of the products the company manufactures, including active pharmaceutical ingredients, excipients, purchased components and raw materials, or the company's customers facing increasing or new competition. These forward-looking statements should be considered together with the risks and uncertainties that may affect our business and future results presented herein along with those risks and uncertainties discussed in our filings with the Securities and Exchange Commission at www.sec.gov. These forward-looking statements are based on information currently available to us, and we assume no obligation to update any forward-looking statements except as required by applicable law.

IRISYS, LLC

Reconciliation of GAAP to Non-GAAP Financial Measures (Unaudited)

To supplement the financial results of IRISYS, LLC ("IRISYS") determined by U.S. generally accepted accounting principles ("GAAP"), the tables below provide the following Non-GAAP information about EBITDA.

EBITDA is net income or loss as determined under GAAP excluding interest and depreciation. We believe that Non-GAAP financial measures are helpful in understanding the business of IRISYS, as it is useful to investors in allowing for greater transparency of supplemental information used by our management in connection with the acquisition of IRISYS. EBITDA is used by investors, as well as our management in assessing IRISYS' past and potential future performance. Non-GAAP financial measures should be considered in addition to, but not as a substitute for, GAAP results. Further, Non-GAAP financial measures, even if similarly titled, may not be calculated in the same manner by all companies, and therefore should not be compared.

Full year guidance

	Twelve months ended December 31,		
(amounts in millions)	20	021	
	(estimate)		
Net loss (GAAP)	\$	2.1	
Interest expense/(income)		0.1	
Depreciation	- 10	0.6	
EBITDA	\$	2.8	